Contributions to Management Science

Marc-Oliver Jauch

Uniform Acrossthe-Board Promotions



Contributions to Management Science

For further volumes: http://www.springer.com/series/1505

Marc-Oliver Jauch

Uniform Across-the-Board Promotions



Dr. Marc-Oliver Jauch Munich Germany

Dissertation awarded Summa cum Laude by the Karlsruhe Institute of Technology (KIT), Germany, on January 14, 2014. Committee members: Prof. Dr. K.-H. Waldmann, Prof. Dr. M. Klarmann, Prof. Dr. M. Ruckes and Prof. Dr. B. Neibecker

ISSN 1431-1941 ISSN 2197-716X (electronic) ISBN 978-3-319-07114-5 ISBN 978-3-319-07115-2 (eBook) DOI 10.1007/978-3-319-07115-2 Springer Cham Heidelberg New York Dordrecht London

Library of Congress Control Number: 2014941095

© Springer International Publishing Switzerland 2014

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed. Exempted from this legal reservation are brief excerpts in connection with reviews or scholarly analysis or material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work. Duplication of this publication or parts thereof is permitted only under the provisions of the Copyright Law of the Publisher's location, in its current version, and permission for use must always be obtained from Springer. Permissions for use may be obtained through RightsLink at the Copyright Clearance Center. Violations are liable to prosecution under the respective Copyright Law.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

While the advice and information in this book are believed to be true and accurate at the date of publication, neither the authors nor the editors nor the publisher can accept any legal responsibility for any errors or omissions that may be made. The publisher makes no warranty, express or implied, with respect to the material contained herein.

Printed on acid-free paper

Springer is part of Springer Science+Business Media (www.springer.com)

Acknowledgements

I would like to thank Martin, my Doktorvater, and the team from the KIT chair, for all the inspiring discussions and guidance to make this book possible.

Thanks to my wife, Christiane, without your love, help and active support this book would probably never have been published. I am very lucky to have you.

Particularly I want to thank my parents for always having given me your love and encouragement regardless of what I was up to and for always having provided me with the opportunity to follow through with what I believed would be the right way for me. Without your love and support not only this book would not have been published.

Last but not least: I ask for forgiveness to those who have been with me in the last years and who have supported this project and whose names I have failed to mention in here.

List of Abbreviations

AVE	Average variance extracted
CR	Construct reliability
DIY	Do-it-yourself (<i>DIY retailer</i> = hardware store)
EDLP	Every-day-low-price
et al.	And others (et aliae)
e.g.	For example (exempli gratia)
FMCG	Fast-moving consumer goods
HiLo	High-low-price
i.e.	That is (id est)
IR	Item reliability
SD	Standard deviation
UABP	Uniform Across the Board Promotion

Contents

1	Intr	duction	1				
	1.1	Background and Practical Relevance					
	1.2	Key Research Topics and Outline of the Thesis	3				
	Refe	ences	4				
2	Con	eptual Framework	7				
	2.1	Pricing Strategies and Definitions of Price Promotions	7				
		2.1.1 Definition of (Traditional) Price Promotions	8				
		2.1.2 Definition of Uniform Across the Board Promotions					
		(UABPs)	9				
	2.2	Theoretical Background	10				
		2.2.1 Prospect Theory and Reference Price Concept	11				
		2.2.2 Price Search and Search Costs	12				
		2.2.3 Behavioral Learning Theory	14				
		2.2.4 Attribution Theory	15				
	2.3	Literature Review Related to the Impact of UABPs	16				
		2.3.1 Short-Term Sales Impact of Price Promotions	16				
		2.3.2 Permanent/Long-Term Effects of Price Promotions	24				
		2.3.3 Summary of Theoretical Background and Literature Review					
		and Outlook of Research Contribution	28				
	Refe	ences	29				
3	Imp	ct of Uniform Across the Board Promotions on Short-Term					
	Stor	Revenues	35				
	3.1	Introduction and Context to Short-Term Effects of UABPs	35				
	3.2	Conceptual Framework	36				
		3.2.1 Definition and Characteristics of Uniform Across the Board					
		Promotions	36				
		3.2.2 Introduction of Store Level Antecedents	38				
	3.3	Method	39				
		3.3.1 Store-Level Promotional Data for Short-Term UABP					
		Impact	39				

		3.3.2	Methodology to Analyze Short-Term Promotional	
			Impact	40
		3.3.3	Hierarchical Bayes Model	41
		3.3.4	Model Variables	43
	3.4	Empir	ical Results	45
		3.4.1	Empirical Results: Short Term Performance Outcomes	45
		3.4.2	Empirical Results: Store Level Antecedents	46
		3.4.3	Comparison of Store Level Antecedents to Other Promotion	
			Types	48
	3.5	Summ	ary and Discussion	49
		3.5.1	Academic Implications from Short-Term Effects	
			UABPs	50
		3.5.2	Retailer Implications from Uniform Across the Board	
			Promotions	52
	Refe	rences		53
1	۸di	ustmon	t and Parmanant Effacts of Uniform Across the Roard	
-	Pro	motions	s	55
	4 1	Introd	uction to Adjustment and Permanent Effects of UABPs	55
	4.1	Conce	ntual Framework	56
	1.2	4 2 1	Definitions of Relevant Concents When Discussing	50
		1.2.1	Adjustment and Permanent Effects	57
		422	Hypothesis Regarding the Adjustment and Permanent	51
		1.2.2	Effects of UABPs	58
	43	Metho	d	60
		431	Store-Level Data to Analyze Adjustment and Permanent	00
			Effects	60
		4.3.2	Methodology to Analyze Adjustment and Permanent	00
			UABP Effects	60
	4.4	Empir	ical Results	64
	4.5	Summ	arv and Discussion	67
		4.5.1	Academic Implications	67
		4.5.2	Retailer Implications from Long-Term Effects of Uniform	
			Across the Board Promotions	70
	Refe	rences		71
_				= 0
5	UAL	3Ps: Ar	itecedents and Consequences on a Household Level	73
	5.1	Introd	uction and Context to UABP Antecedents and Consequences	
	5 0	on a E	Iousehold Level	73
	5.2	Conce	ptual Framework: Antecedents and Consequences	74
	5 2	5.2.1	Consequences of UABPs	/8
	5.3	Metho		80
		5.3.1	Household Level Survey and Transaction Data	80
		5.3.2	Methodology to Analyze Antecedents and	C 1
			Consequences: CFA and SEM	81

	5.4 Empirical Results			83
		5.4.1	Results Regarding the Antecedents of UABPs	83
		5.4.2	Results Regarding the Consequences of UABPs	85
5.5 Summary and Discussion				86
		5.5.1	Academic Implications	86
		5.5.2	Retailer Implications from Long-Term Effects of Uniform	
			Across the Board Promotions	87
	Refe	rences.		88
6	Sum	mary o	of Results	91
	6.1	Summ	ary and Discussion of Results	92
	6.2	Manag	gerial Implications	94
	Refe	rences.		95
Ap	pend	ix A: D	De-seasonalizing and De-trending of Sales	97
Ар	pend	ix B: S	urvey Results (Antecedents and	
Co	nseq	uences	of UABPs)	99

List of Figures

Overview of promotion types	8
Overview of consumer promotions	9
Classification of UABP among pricing strategies	10
Illustration of research framework	38
Framework of impact of Uniform Across the Board Promotions	
on store sales	58
Pre-and post-promotion dips for Uniform Across the Board	
Promotions	65
Visualization of long-term evolvement of immediate and	
adjustment effects	66
Antecedents and consequences of UABPs	75
Results of model estimation	85
	Overview of promotion types Overview of consumer promotions Classification of UABP among pricing strategies Illustration of research framework Framework of impact of Uniform Across the Board Promotions on store sales Pre-and post-promotion dips for Uniform Across the Board Promotions Visualization of long-term evolvement of immediate and adjustment effects Antecedents and consequences of UABPs Results of model estimation

List of Tables

Table 1.1	Examples of Uniform Across the Board Promotions	2
Table 2.1	Decomposition of promotional sales elasticities	18
Table 3.1	Promotion types and advertising types	39
Table 3.2	Store characteristics	40
Table 3.3	Impact of promotions on marginal sales-individual store	
	values	46
Table 3.4	Impact of promotions on marginal sales-across-store	
	values	47
Table 3.5	Impact of store characteristics on sales impact of UABPs	47
Table 3.6	Impact of location on different promotion types	49
Table 3.7	Impact of value per article on different promotion types	49
Table 3.8	Impact of age of store on different promotion types	50
Table 3.9	Illustrative calculation of required volume increase to	
	break-even during UABPs	52
Table 4.1	Adjustment effects for UABPs (pre- and post-promotion	
	dips)	65
Table 4.2	Impact of running more UABPs on immediate and adjustment	
	effects	66
Table 4.3	Estimated incremental store sales gains	69
Table 5.1	Indicator and construct reliabilities from CFA	82
Table 5.2	Correlations and measurement information	83
Table 5.3	Antecedents and consequences of Uniform Across the Board	
	Promotions	84

About the Author

Dr. Marc-Oliver Jauch is partner and co-founder of Castik Capital Partners, a European private equity firm. Prior to this he worked as an investment professional for the global private equity fund Apax Partners in their London and Munich office. Previously he was a project leader with the Boston Consulting Group based out of Munich.

Marc-Oliver has a Ph.D. (Dr. rer.pol.) in economics and managerial science (summa cum laude), holds an MBA from ESSEC Business School in Paris and a MA (Dipl.-Kfm) in Finance and Economics from the University of Mannheim (highest distinction).

In addition, Oliver has successfully invested in and co-founded several start-ups and supports the charity foundation Joblinge, on whose board he sits.

The author is married, lives in Munich and loves sports as being an active tennis and football player.

Chapter 1 Introduction

1.1 Background and Practical Relevance

Retail markets in Germany are becoming more competitive, fueled by a continuous shift from offline to online channels, globalization of the competition and demand saturation in some markets (Homburg 2012). Given this development, retailers face the challenge of attracting customers to their stores to ensure continuous sales and profit growth. With products and styles becoming more and more exchangeable and customers being willing to mix and match between different brands and stores, price management and particularly promotion management grows in importance to win "*the battle for the customer*".

However, as customers might become more accustomed to the omnipresence of promotions and might expect to buy their goods at reduced prices, retailers are coming up with new and innovative forms of price promotions. One type of price promotion, that has gained significant importance in the last years, particularly in the German market, are Uniform Across the Board Promotions (UABPs, see Sect. 2.1.2 for definition). UABPs are in a way special as they offer the customer a uniform percentage discount *on any article* she finds in the store for a limited time period.

In 2003, the DIY chain Praktiker had been among the pioneers to introduce UABPs to the German market and present their customers with a "20 % on everything—except pet food" promotion (Bomsdorf et al. 2003), which had a significant positive impact on their short-term sales and the attention drawn to the retail brand and its stores (Böcking 2012; Hielscher 2010). Given this apparent short-term success, more and more retailers have followed to adopt this type of price promotion as part of their marketing plan. For example one of the largest European fashion retailers, C&A, has offered its customers 20 % off on everything for a week to celebrate its 170 years anniversary (Sparwelt 2012). The large German shoe retailer Goertz sends out 20 % UABP-coupons once a year (Schnäppchenfuchs 2012). International high-street fashion retailers such as Esprit (Style Now 2009), and Replay offer their customer periods with 20 % off on

				Promotion	
Company	Industry	Article type	Channel	depth	Duration
Praktiker	Home Improvement	Durable	Offline	25 %	2–7days
C&A	Apparel retail	Durable	Online	24 %	2–7days
Ernest Alexander	Apparel retail	Durable	Online	30 %	1 day
NKD	Apparel retail	Durable	Online/ offline	20 %	1 day
Dorothy Perkins	Apparel retail	Durable	Online	20 %	1 day
meinPaket.de	Department stores (online)	Durable	Online	12 %	1 day
Toys R'Us	Specialty stores	Durable	Online	10 %	1 day
Reno	Apparel retail	Durable	Online/ offline	20 %	2–7days
Expert	Comp & Electro Retail	Durable	Online/ offline	10 %	2–7days
IKEA	Home improvement retail	Durable	Offline	5 %	2–7days
The Body Shop	Personal products	Consumables/ discretionary	Online	20 %	2–7days
Merkur	Food retail	Consumable	Offline	10 %	2–7days
H&M	Apparel retail	Durable	Online	10 %	2–7days
Goertz	Apparel retail	Durable	Offline	20 %	2–7days
Sony	Comp & Electro	Durable	Online	8 %	2–7days
BonPrix	Apparel retail	Durable	Offline	20 %	2–7days
Marionnaud	Personal products	Consumables/ discretionary	Offline/ online	20 %	2–7days
Ernsting's family	Apparel retail	Durable	Offline	20 %	2–7days

Table 1.1 Examples of Uniform Across the Board Promotions

everything. The electronic goods manufacturer Sony offered its retail customers 8 % discount on everything (Qipu 2012), and even grocery retailers such as Penny Markt (Paperblog 2011) and Merkur (Sparfreunde 2012) present their customers with a 10 % across the board discount for a limited period of time. Table 1.1 provides an overview of exemplary UABPs that have been run in the time period 2009–2012.

While UABPs have been increasingly used, its pioneer in Germany, Praktiker, meanwhile ran into significant problems, bringing the company to the brink of insolvency. After the original success of UABPs, Praktiker increased the UABP days significantly (to 107 in 2007), before facing significant (sales, margin and image) problems and cutting down on UABPs entirely in 2011 (Böcking 2012; Hielscher 2010). Not few attribute those problems to the excessive use of UABPs. One of the core investors of Praktiker, Thomas Fox, called the UABPs "sweet drugs", others stated that "Praktiker can now not do with nor without them" (Hielscher 2010; Läsker 2012). Against this background, it is the goal of this thesis,

to investigate the performance impact of UABPs on the short-mid and long-term sales of a retailer and explain what type of consumers are attracted to it, what store portfolio is ideal for applying a UABP promotion strategy and how consumers attitude towards the retailer is affected by it.

Moreover this work should help managers to form a substantiated view on whether UABPs can support their strategic promotional goals and if UABPs turn out not to be the right tool, how they could scale them back or adjust them in order to not risk the mid-term success of their business. The following chapter will outline the key research topics and structure of this thesis.

1.2 Key Research Topics and Outline of the Thesis

This thesis is structured in six main chapters in which the impact of the Uniform Across the Board Promotions on sales, customer behavior and customer attitude will be further discussed. The core of this thesis can be found in Chaps. 3, 4, and 5, each one addressing specific aspects of the UABPs, which have been analyzed using a hierarchical Bayes model, a (linear) mixed effects time series model with lead and lag effects as well as confirmatory factor analysis and structural equation modeling. Each chapter is based on different, extremely comprehensive proprietary datasets from a European value fashion retailer. Among these datasets are daily and weekly store-level data for ~750 retail stores over a period of around 4 years, the analysis of detailed marketing plans of the retailer as well as its typical customers.

Moreover the store-level analysis from Chaps. 3 and 4 is complemented with the analysis of the customer behavior and attitude which is based on an extensive customer survey with ~1,700 participants that have been matched with their respective transactions from UABPs over a 4 year period. The work is structured as follows:

Chapter 2 provides the conceptual framework for Uniform Across the Board Promotions and offers a definition of traditional price promotions and UABPs, while also classifying the UABP vis à vis other price promotions and pricing strategies and pointing out some of the unique characteristics of UABPs. The chapter further summarizes the theoretical background by introducing some relevant concepts and theories such as the adaption-level theory, Prospect theory, Reference Price concept, price search and information economics as well as Behavioral Learning and Attribution theory. Moreover the relevant promotion literature around the short-mid and long-term impact of promotions will be reviewed to build the theoretical foundation for the later research carried out in this thesis. Last this chapter summarizes the findings and discusses the relevance and applicability of the different research results and theoretical concepts for the Uniform Across the Board Promotions.

Chapter 3 discusses the impact Uniform Across the Board Promotions have on short-term retail sales and compares this impact to other types of price promotions.

A hierarchical Bayes model will be applied to analyze weekly store-level data of 742 stores over 4 years to further discuss which characteristics of a retail store portfolio support the short-term effectiveness of UABPs. Furthermore the analysis in Chap. 3 shows whether those store characteristics are generally favorable for retail price promotions or whether this is a particularity for UABPs. The chapter concludes with managerial considerations regarding the short-term use of UABPs.

Chapter 4 focuses on the net effects as well as the longer term effects UABPs have on sales. It therefore discusses the adjustment effects of UABPs and analyses how these develop over time. Furthermore the chapter focuses on the question whether UABPs have a negative long-term effect on baseline sales. The analysis is based on daily store level data from a value fashion retailer, which is being analyzed using a (linear) mixed effects time series model with lead and lag effects.

Chapter 5 discusses the antecedents and consequences of Uniform Across the Board Promotions, to answer questions like "*who is the typical customer during UABPs*" and what impact do UABPs have on concepts such as customer loyalty, price image of the retailer or customer satisfaction. Furthermore this chapter discusses whether on a household level, customers really do change their behavior. These questions are being analyzed based on a combination of an original dataset of transaction data from 20,000 customers, out of which 1,641 have completed a questionnaire about their shopping behavior and attitude. These datasets have been combined and analyzed using a confirmatory factor analysis (CFA) as well as structural equation modeling (SEM). In addition to the store-level sales analysis conducted in Chaps. 3 and 4, this chapter gives more insight into the individual consumer and his attitude towards UABPs.

Chapter 6 summarizes the findings from the three studies carried out in Chaps. 3, 4, and 5 provides managerial implications. The goal of this chapter is to provide a quick overview of what UABPs are—what their impact on sales and the consumer is and what managers can do to best exploit their potential and avoid their risks.

References

- Böcking D (2012) Dramatische Hauptversammlung: Praktiker droht Aktionären mit Pleite. http:// www.spiegel.de/wirtschaft/unternehmen/praktiker-sanierungskonzept-sorgt-fuer-aerger-aufhauptversammlung-a-842634.html. Spiegel Online, last accessed 15 Apr 2013
- Bomsdorf C, Kerbusk KP, Scheele M (2003) Erst Gas geben, dann bremsen, Spiegel 49/2003. http://www.spiegel.de/spiegel/print/d-29341609.html. Last accessed 30 Nov 2012
- Hielscher H (2010) Praktiker verordnet sich Reperaturprogramm. http://www.wiwo.de/ unternehmen/harter-entzug-praktiker-verordnet-sich-reparaturprogramm/5687658.html. Wirtschaftswoche Online, last accessed 15 Apr 2013
- Homburg C (2012) Marketing management, 4th edn. Springer Gabler, Wiesbaden
- Läsker K (2012) Rabatte sind wie süße Drogen. http://www.sueddeutsche.de/wirtschaft/sparkusbei-der-baumarktkette-praktiker-rabatte-sind-wie-suesse-drogen-1.1322133. Süddeutsche Online, last accessed 15 Apr 2013

- Paperblog (2011) Penny Markt Frühshopping: 10% auf alles bis 10Uhr (z.B. 400g Nutella für 1€). http://de.paperblog.com/penny-markt-fruhshopping-10-auf-alles-bis-10uhr-zb-400g-nutellafur-1-172014/. Last accessed 10 Dec 2012
- Qipu (2012) 8% Cashback auf alle Produkte im Sony Store + 5% Rabatt im Outlet Bereich. http:// blog.qipu.de/deal/sony-cashback-gutschein-outlet/. Last accessed 27 Dec 2012
- Schnäppchenfuchs (2012) 20% Rabatt auf Alles bei Görtz auch auf reduzierte Ware. http://www. schnaeppchenfuchs.com/kleidung/20-rabatt-auf-alles-bei-gortz-auch-auf-reduzierte-ware. Last accessed 30 Nov 2012
- Sparfreunde (2012) Der grosse Rabatz 2012. http://www.sparfreunde.com/sites/default/files/ dealpics/aaa1_7.jpg. Last accessed 27 Dec 2012
- Sparwelt (2012) 20% Rabatt auf alles bei C&A. http://www.sparwelt.de/shopping/sale/cunda-hat-geburtstag-24-prozent-rabatt-auf-alles-31919. Last accessed 30 Nov 2012
- Style Now (2009) 20 % auf alles bei Esprit. http://www.style-now.de/20-auf-alles-bei-esprit/. Last accessed 30 Nov 2012

Chapter 2 Conceptual Framework

2.1 Pricing Strategies and Definitions of Price Promotions

"Nothing is more important in business than getting the pricing strategy right" summarizes the importance of choosing and implementing the right pricing strategy and tactics for a retailer (Tang et al. 2001). Generally retail managers can decide between two core pricing strategies, the *every-day-low-price* (EDLP) strategy or the *HiLo pricing* which together make up for over 56 % of pricing strategies used in the US, with other strategies (often derivatives of the two) such as *exclusive pricing*, *aggressive pricing* and *moderately promotional pricing* being used by ~44 % of US retailers (Bolton and Shankar 2003; Hoch et al. 1994; Lal and Rao 1997; Tellis 1986). In the following this thesis focuses on introducing EDLP and HiLo pricing.

When implementing an *every-day-low-price* strategy, the average price for every article is selected to be between a regular market price and a promotional price. By offering below regular market prices on all items, the EDLP-strategy aims at attracting price- and time-sensitive customers, who want to combine an attractive offering while not visiting multiple shops to hunt for the cheapest bargain (Lattin and Ortmeyer 1991; Seiders and Voss 2004). This strategy is very common as according to Bolton and Shankar (2003), approximately 45 % of US retailers, among them industry leaders such as WalMart, HomeDepot, CostCo and Aldi, have implemented this strategy (Bolton and Shankar 2003). In its pure form, no additional temporary price promotions would be granted to the customer.

The *HiLo pricing* strategy is defined as offering higher non-promotional prices mixed with temporary discounts on individual brands or categories to customers. These temporary discounts or variance in prices are a distinct characteristic and difference to EDLP-strategy. In contrast to the EDLP, the HiLo strategy attracts cherry-pickers, who are willing to invest additional effort in finding and visiting the stores with the cheapest price for a brand, even if this would require them to visit multiple stores (Bell and Lattin 1998). The HiLo strategy is less common than the EDLP strategy, as in the US only 11 % use this strategy, among them retailers like Lion, Safeway and Vons (Bolton and Shankar 2003). Within the HiLo pricing



Fig. 2.1 Overview of promotion types

strategy, retailers (as a matter of fact also manufacturers/wholesalers) are free to choose among various different designs of price promotions.

2.1.1 Definition of (Traditional) Price Promotions

For sales promotions one differentiates promotions targeted at the retailer and promotions targeted at the consumer. Sales promotions that are directed towards retailers are called trade promotions, and promotions that a directed towards consumers are called consumer promotions (Blattberg and Neslin 1990; Gedenk 2002, p.14) (Fig. 2.1).

Consumer promotions can further be classified as price promotions or non-price promotions. Non-price promotions are for example free samples, special packaging or promotion games, while price promotions are for example discounts or coupons (Gedenk 2002, p. 19) (Fig. 2.2). Price promotions can be defined as (1) temporary limited discounts to the regular market price, (2) sometimes supported by additional marketing measures (3) to increase sales for a retailer, wholesaler or manufacturer (Diller 1984; Gedenk 2002). Sales promotions and price promotions alike, do not necessarily have to focus on reaching short-term goals, such as the immediate increase in sales, but can also be targeted at e.g. increasing a retailer's image or increasing long-term sales (Gedenk 2002).



Fig. 2.2 Overview of consumer promotions

2.1.2 Definition of Uniform Across the Board Promotions (UABPs)

Uniform Across the Board Promotions, which is a term introduced to the literature with this thesis, are part of the broader category of *consumer price promotions*. As Uniform Across the Board Promotions have not been defined in the marketing literature before, this thesis introduces a definition, which is based on a qualitative pre-study further described in Chap. 3. For the avoidance of doubt, all other price promotions in this thesis are referred to as *traditional* or *other* price promotions.

Uniform Across-the-Board Promotions can be defined along four characteristics:

"Uniform Across the Board Promotions (1) are store-wide promotions on the entire assortment, that grant the customer (2) an undifferentiated discount on the entire assortment, i.e. no bigger or smaller discounts on any article. (3) This discount is only valid for a limited time period and (4) is expressed relative to the article prices, i.e. as a percentage to the absolute value."

Particularly the breadth of the promotion combined with the undifferentiated discount are differentiating criteria, as (a) other promotions are normally not applied to the entire assortment and (b) retailers generally do not offer the usually a bit cheaper white label products or the subsidized "category loss leaders" along the entire assortment. Even during sales periods, some articles are normally excluded from promotions or the depths of the discounts vary. As for traditional price promotions, UABPs can be supported by other forms of promotion and advertising, such as coupons, TV campaigns etc. Relative to the pricing strategies described in Sect. 2.1, UABPs combine elements from the EDLP strategy and the



Fig. 2.3 Classification of UABP among pricing strategies

HiLo pricing strategy and are hence targeted at efficient shoppers and cherry pickers alike. The fact that the price promotion is applied on the entire assortment is an EDLP characteristic, while the temporary restriction is an element used in HiLo pricing (Bell and Lattin 1998; Hoch et al. 1994). Figure 2.3 further displays the characterization of UABPs compared to other pricing strategies.

2.2 Theoretical Background

In the following some cornerstone theories and concepts from research in social psychology, micro-economics, and behavioral economics will be introduced, that form the basis for the empirical research conducted later in this thesis and provide explanations to the observed results. Concepts and theories discussed include:

- *Prospect theory/reference price concept* (2.2.1): Theory that goes back to adaption-level theory, that e.g. discusses, how promotions alter the price expectations of consumers and how this affects future buying behavior.
- *Price search and search cost* (2.2.2.): Concept with background in microeconomics, which states that consumers are always putting the search for better offers in context to the actual and opportunity costs related to the search.

2.2 Theoretical Background

- *Behavioral learning* (2.2.3): Theory from social psychology, stating that consumers are constantly learning and adapting their behavior, especially if they are conditioned to change through facing continuous stimuli.
- Attribution theory and self-perception theory (2.2.4): Has its background in social psychology and states that consumers always try to find an explanation for a specific state, meaning they e.g. put price promotions in a certain context. This context influences their attitude and behavior towards the observed state.

2.2.1 Prospect Theory and Reference Price Concept

The reference price concept states that consumers do not just evaluate observed prices on a stand-alone basis and as absolute but rather evaluate them in a relative context. This concept goes back to the prospect theory (Kahneman and Tversky 1979) and eventually the adaption-level theory (Helson 1964).

According to the adaption-level theory (Helson 1964), consumers compare a state, i.e. an observed price to a personal reference-level or an adaption-level. These adaption-levels are formed by the individual response to three classes of cues: focal, contextual and organic. Focal cues are stimuli that a consumer is directly responding to (i.e. the observed price), with the contextual cues describing every-thing in the background to those cues (i.e. comparison with prices of other products). Organic or residual cues are built through previous experiences of the consumer (i.e. past prices that consumers have paid) (Helson 1964). Building on the different cues used to form adaption-levels or reference prices, there is however a discussion that different types of consumers form different reference prices. While more brand-loyal customers focus on their internal reference price, which is built from previous experiences, brand switchers (deal seekers) focus on external (contextual) reference prices (Puto 1987; Rajendran and Tellis 1994).

Kahneman and Tversky (1979) have built on this concept when formulating the prospect theory. According to the prospect theory, which then forms the basis for the reference price concept, consumers do not just look at the actual prices of goods or services but, in line with the adaption-level theory, always put this price in context relative to a personal reference price. The price perception of a brand or retailer depends not only on the actual price of products offered but also on their reference price and the relationship between the two. A positive deviance of the actual price from this adaptation-level or reference price, i.e. a lower actual price than the reference price would in terms of mental accounting be considered a "profit" and lead to a positive price perception, while a negative deviance would be booked as a "loss" (Monroe 1973).

Kahneman and Tversky (1979) found that the utility function for those profits and losses is concave for profits and convex for losses. Moreover the curve is steeper for losses than it is for gains. This means that consumers are in general riskaverse and in absolute terms losses are more negative than the same profit would be positive. In terms of promotions this means that a temporary discount (promotion), with an observed price lower than the reference price will be considered a profit, while the return to the actual price compared to the "new reference price" would be a loss—with the net of the two effects being negative.

As consumers adjust their expectations to past experiences, e.g. promotions, the response of a consumer to a certain type of promotion can change over time, given the changed reference prices (Anderson and Simester 2004). Regarding promotions and in particular UABPs, historic experiences, and the increased predictability of finding an article at a lower price in the future, could make non-promotional prices less attractive and even provide a lower incentive to act on future promotions, given that the observed promotional price and the new reference price are converging (see Sect. 2.3.2 for a detailed discussion on long-term promotional effects).

Despite a general acceptance of this theory, the concept of the reference price and the rational consumer has come under some scrutiny as empirical research has shown that consumers often have little price knowledge and their internal reference price is often not very precise (Buzas and Marmorstein 1988; Dickson and Sawyer 1990). In reality, rather than having precise price knowledge, consumers use external cues while remembering whether a price was cheap or expensive. This heuristic is in most purchase situations sufficient for making purchase decisions or forming a view about a retailer and brand (Krishnamurthi and Raj 1988; Mazumdar and Monroe 1990).

For the applicability of the reference price concept on UABPs, it can be assumed that the "x percent promotion on everything" claim forms a strong enough external claim itself, so that the relative discount can be viewed as "reference" to what is normally charged for a product. This would mean that consumers do not compare the actual prices paid but rather use the heuristic of the existing "x percent promotion on everything" UABP discount as a reference value. This could lead to consumers expecting a certain (type) of discount rather than a precise price for an article.

2.2.2 Price Search and Search Costs

The behavioral pricing research differentiates between three phases of how consumers deal with prices. The first phase sets out how information on prices are obtained, the second one how those information are processed and the third one how the processed price information is stored by the consumer (e.g. Homburg and Koschate 2005). The *price information search*, i.e. the first phase, herein sets the basis for the processing and acting on the information.

The *interest in price* information, describes the general desire of a customer to search for price information and to consider these information in their decision making process. The greater the interest in price, the higher is the importance of price and the lower the willingness to pay an above average price (Diller 1999). The *interest in price* is increased through promotions, as they put the focus on the price

of a product which can also be tactically used by retailers, aiming at increasing the interest in price (Dickson and Sawyer 1990; Naik et al. 2005).

While the interest in price describes the original intention to search for prices, the *price search*, which is the undertaken effort to look and compare prices, influences the actual purchase process. In general price search can take place either between-stores (*between-store-search*), i.e. comparing prices across different stores for comparable products or within-stores (*in-store-search*), which refers to comparing prices for comparable products and brands within one store (Urbany et al. 1996). The promotion aspect of within-store search is less relevant for Uniform Across Board Promotions, which is why the focus will be more on *between-store* search elements. Understanding the factors influencing the intensity of between-store search will help to better understand in which retail settings UABPs might work better than in others (see also Chap. 3). Various older research to put only relatively little effort in the actual process for price search (Beatty and Smith 1987; Grewal and Marmorstein 1994). However price search is influenced by several influencing factors such as:

- Value of the product
- Observed price range
- · Price guarantees
- · Price knowledge
- · Search costs
- Promotions

Regarding the influence *value* has on the price search intensity the existing evidence is mixed. Darke and Freedman (1993) argue that there is a positive effect of a higher value on the price search intensity, as the same relative discount leads to bigger absolute savings, if an article has a higher base price. Grewal and Marmorstein (1994) have found different evidence in the retail market for electronic goods, as the same absolute promotions gives a higher relative benefit to the customer if the article has a lower value. They hence conclude that value has a negative correlation with price search intensity.

Consumers normally have a general idea of the *price range*, meaning the lowest and highest price, to which a certain product can be purchased. Similar to the reference price theory, consumers compare the observed price with this price range and hence try to form an opinion on the probability to shop for a lower price (Urbany 1986). The wider this price range, i.e. the "perceived price dispersion", the higher the chances of finding a cheaper price and the higher the price search intensity (Duncan and Olshavsky 1982).

Price guarantees, which are given by certain retailers, aim at stopping the price search of customers, as they are offered the difference between the price spent for a product and a potentially cheaper price in a different store. This reduces the uncertainty for the customer, and hence has a negative effect on the price search intensity (Jain and Srivastava 2000; Srivastava and Lurie 2001).

The literature concludes that *price knowledge* and the price search efficiency positively impacts price search intensity, potentially also because it indirectly lowers the price search costs (Alba and Hutchinson 1987; Urbany et al. 1996). However this is only valid, if consumers are generally still searching for prices and have not yet "subscribed" to only one particular brand or store (Alba and Hutchinson 1987).

Search costs, which includes all incurred and opportunity costs associated with finding the best price, impacts the price search intensity. Information economics, which focus on utility maximization, argue, that price search is conducted as long as the expected savings are still as high as or higher than the cost incurred by an additional shopping trip (Stigler 1961; Metha et al. 2003). However the subjective utility from the satisfaction of having found a lower price, which is one motivation of bargain hunters, also has to be considered (Marmorstein et al. 1992). As research in the area of psychology states, consumers prefer the heuristic processing of information (over the systematic processing), as decisions are made based on simple decision rules, which reduce their mental effort (Chaiken 1980). *Price promotions* in a store can be seen as such an easy decision rule, as the subjective probability of finding a lower price elsewhere is reduced and consumer hence terminate their price search. Promotions hence have a negative impact on the price search—the higher the absolute discount, the more negative the impact (Darke and Freedman 1993; Darke et al. 1995).

According to the theory of information economics, customers always put the *search costs* in relation to the expected savings (e.g. Metha et al. 2003; Stigler 1961) and one could expect that customers incur higher relative search costs if the expected savings are higher or if the expected relative savings are more certain. Regarding the relative promotion depth, UABPs in our example are generally comparable to other types of price promotions (compare e.g. Chap. 4, Table 4.14), while the certainty of finding the desired article (in a particular size, color and style) on sale and available during UABPs is significantly higher, as the promotion is applied to *every article* in the store. The hypothesis is that rationale consumers would hence be willing to incur higher search costs to shop during UABPs than they would occur for other types of promotions. In theory this could result in UABPs attracting more customers that are willing to incur higher promotional increases than for regular promotions. Part of this will be further analyzed in Chap. 3, when discussing the optimal retail settings for UABPs.

2.2.3 Behavioral Learning Theory

The central concept behind behaviorism or behavioral learning theory, introduced by Skinner (1953) and Thorndike (1911), states that positively reinforced behavior is more likely to reoccur than non-reinforced behavior. Nord and Peter (1980) as well as Rothschild and Gaidis (1981), have summarized the general concepts and

applied them to the field of marketing. Applied to marketing the behavioral learning theory suggests, that a transaction or purchase (response) takes place once a product (stimulus) is favorably received by the customer. If the customer is satisfied with the product (positive reinforced), the probability of repeat purchases increases. Marketers enhance the value of the product through additional stimuli, like adapting price, distribution or promotional variables. Such positive reinforcement, through favorable variables, further increases the purchase probability of the product. In the following some core components of behavioral learning in marketing will be quickly introduced and their potential impact on the purchase behavior for Uniform Across the Board Promotions.

The first step is called shaping, which derives a new behavior, such as choosing a new brand or store. This first step is essential as positive reinforcement, i.e. learning, can only occur, once behavior has already occurred. Rothschild and Gaidis (1981) suggest that shaping occurs through rewarding many small existing patterns, out of which a new, more complex behavior can occur. For example by advertising and promoting one can attract a consumer to a specific store once she already has the intention to buy in a certain product category. However the visit to the store is in this instance closely connected and reinforced by the specific additional stimulus (promotion, advertising). To ultimately achieve a change in behavior that is linked to the store rather than the additional stimulus, the ancillary incentives, such as the promotions and advertising need to fade out.

A common problem and significant risk in the use of those promotions stimuli is either the improper fading out or the over-usage of promotions/UABPs. If promotions/UABPs are not fading, i.e. being decreased over time, store sales may decrease significantly the moment those promotions are eventually dropped (see also Praktiker example and analysis in Chap. 4). This is caused by people switching back to a different store as the stimuli has become the main cause for choosing a certain store, rather than the store's products, features etc. Moreover an over usage of promotional or advertising aids may causes the consumer store choice to be contingent upon and overshadowed by those tools—once these aids are dropped, consumers might also switch back to a different store. To cause a long-term change in shopping behavior, promotional tools, hence also the UABPs can only be used to draw customers to a store, while continuously removing the correlation between the response (store selection) and the reward (UABP), to put the inherent store benefits back in focus (Rothschild and Gaidis 1981).

2.2.4 Attribution Theory

Attribution is a concept from the research in social psychology stating that individuals try to explain the causes for certain observed behavior and events. This concept has also been applied to marketing or more specific pricing research, arguing that consumers try to understand the reasons why a certain article or brand is on promotion (Folkes 1988; Lichtenstein et al. 1991). Consequently a

price promotion can, depending on the reason consumers expect behind it, be seen as positive or negative (Folkes 1988; Weiner 2000). For example the attitude towards are retailer and the value of a promotion is viewed more positively, when the customer thinks the promotions aim at winning market share and attracting customers rather than just for clearance of stock (Lichtenstein and Bearden 1989).

When consumers face unexpected promotions they try to assess whether this has something to do with the product (quality) or the store (Lichtenstein and Bearden 1989). A promotion which consumers cannot explain could be perceived as unfair even if beneficial for the consumer (Xia et al. 2004; Ordóñez et al. 2000). When consumers are facing very large discounts, they are discounting the promotion unless they are familiar with the brand (Moore and Olshavsky 1989). Regarding UABPs it will be interesting to see whether consumers are discounting the price promotion, impacting e.g. image of the brand, not just because of the actual depth but also because of the breadth (*"promotion of every article in the store"*) of the promotion. Especially in the long-term the impact of very deep discounts or very broad promotions, such as the UABP, has to be assessed as beyond a certain level the perceived value of a promotion is higher if the type of discount is rare—very frequent UABPs should hence lose in efficiency (Lichtenstein and Bearden 1989). The impact of UABPs on the brand and how they are perceived in general will be discussed in Chap. 5, where a connection to the attribution theory will be drawn.

2.3 Literature Review Related to the Impact of UABPs

2.3.1 Short-Term Sales Impact of Price Promotions

When addressing the impact of promotions on short-term consumer behavior, the marketing literature differentiates between two general effects: sales bump and promotional dips. The sales bump or immediate effect of price promotions refers to the increase in sales during a promotion campaign. Potential adjustment effects or pre- and post-promotion dips are understood as the "troughs" prior or post a promotion, during which sales are below the usually observed sales level, as consumers adjust their behavior to the promotions.

This topic was first analyzed in the late 70s, when promotional effects were decomposed and the first evidence for purchase acceleration was empirically observed. Research found, that consumer make sophisticated decisions and increase the quantities of consumer goods purchased (soap, coffee and orange juice) during promotions compared to non-promotional quantities ("sales bump"). Those purchases are often made at the expense of later purchases ("post-promotion dips") (Ward and Davis 1978). In the following the existing literature on the three short-term promotional effects will be reviewed and the potential implications for UABPs will be derived.

2.3.1.1 Immediate Effect of Price Promotions

This paragraph focuses on the immediate effect of price promotions, also called "sales bump", which refers to the increase in sales caused by a (temporary) reduction in price. The paragraph will summarize a review of the respective literature regarding key aspects of the immediate effect of price promotions and derive potential implications for Uniform Across the Board Promotions on:

- · Category incidence, brand choice/switching, purchase quantity
- Store choice
- Complementary effects (e.g. impact on customer mix, complementary products)

Effect on Category Incidence, Brand Choice and Store Switching

Earlier research has usually analyzed the promotional impact on brand choice (e.g. Guadagni and Little 1983), quantity decisions (e.g. Blattberg et al. 1981) and category incidence as separate topics. Gupta (1988) has then been among the first to measure the impact, each of those components have on sales simultaneously, as this helps to better understand the overall effectiveness of a promotion. For this reason this paragraph discusses the impact price promotions have on *category incidence, brand choice* and *purchase quantity* together.

Category incidence is the likelihood of consumers purchasing in a certain category, while *brand choice* is the impact promotions have on the selection and switching to a certain brand. Purchase quantity, which is often discussed together with acceleration of purchases or inter-purchase time, refers to consumers buying more (and earlier) of a product due to the product being on promotion. Especially for purchase quantity (and acceleration) it is important to understand, that those effects cannot be seen as stand-alone but are closely related to pre-and post-promotion dips, which will be further discussed in the following chapter.

It is undisputed, that store-sales generally respond positively to short-term price promotions, while the effect can be decomposed into within-category brand switching (e.g. Gupta 1988) or category expansion (e.g. Chintagunta 1993; van Heerde 1999). While within-category brand switching does not increase sales of a category but rather shifts those sales to a different brand, category expansion increases the demand for a certain category. Various researchers have further explored these effects, by decomposing the short-term sales increase into primary demand effects: purchase acceleration and increase in purchase quantity and secondary demand effects: brand switching.

As Table 2.1 shows, the elasticity decomposition of promotional impact as measured from household level data shows a broad range of results, for different categories. The average shows, that 74 % of the increase in sales can be attributed to secondary effects and more specific: brand switching (Bell et al. 1999; Bucklin et al. 1998; Chiang 1991; Gupta 1988). Van Heerde et al. (2003) who have analyzed

Study	Category	Brand switch.	Timing accel.	Quantity accel.
Gupta (1988)	Coffee	84 %	14 %	2 %
Chiang (1991)	Coffee (featured)	81 %	13 %	6 %
	Coffee (display)	85 %	5 %	10 %
Chintagunta (1993)	Yogurt	40 %	15 %	45 %
Bucklin et al. (1998)	Yogurt	58 %	19 %	22 %
Bell et al. (1999)	Various	49–94 %	1-42 %	0-45 %
Average		74 %	11 %	15 %

Table 2.1 Decomposition of promotional sales elasticities

store level data have only found this effect to account for 33 % of the short-term sales increase (see also Table 2.1, as inspired by van Heerde et al. 2003).

Walters (1991) in an earlier article has found that the negative impact of a promoted brand on its substitutes, depends firstly on the brand or category but also on the similarity of the perceived attributes of the two brands-this can likely explain some of the variance related to the degree of brand switching shown in Table 2.1. For retailers it is important to understand the brand switching effect, as the within store substitution might be desirable from a manufacturer point of view, however it might not increase overall store sales for the retailer. Based on the average results above (Table 2.1), 26 % of the sales increase is due to primary demand effects, i.e. short-term sales increases for the retailer, that are caused by customers buying more and buying earlier. Whether over a longer period of time, this leads to a net increase in sales for the retailer or causes post-promotion dips will be discussed in more detail later (see Sect. 2.3.1.2 for post-promotion dips). Especially if consumption rates react positively to promotions, i.e. customers consuming faster if they have bought an article at a reduced price, the category incidence increases and the net promotional effect for the retailer is likely even higher (Ailawadi et al. 2007; Bell et al. 1999).

When discussing the sales impact of UABPs against this background, it is important to note, that UABPs are targeted exactly at those primary demand effects, as there is usually no significant within-store substitution expected, given that the promotion is applied to every article in the store. On the contrary UABPs are not a mean to influence secondary effects or specifically brand choice but rather store choice and category incidence as will be discussed in the next chapter.

Effect on Store Choice

Part of the sales bump caused by promotions can be attributed to store switching. As stated before, UABPs are retail driven consumer promotions, i.e. unlike trade promotions or manufacturer driven consumer promotions that target an incremental increase in sales of a specific brand or product, UABPs aim at increasing overall retail store sales. They are hence an important marketing tool from a retailer's

perspective. The following summarizes, how and if price promotions generally affect consumer store choice.

Keng and Ehrenberg (1984) show, for a grocery retail setting, that consumers generally have low store loyalty and little segmentation between different stores and retail chains, when it comes to selecting a certain product. This is an indication, that consumers can theoretically be lured into switching stores by offering them attractive promotions.¹ This has been supported by work from Kumar and Leone (1988) and Walters (1991) that have found that pricing activity for a specific brand in one store has negative effects on sales in its competing stores or more specifically that price promotions of a brand in one store has a negative effect on the brand or its substitutes in a competing store. Bucklin and Lattin (1992) introduce two concepts of competition, direct and indirect, between retail stores that have different implications on the promotional sales effects. Direct effects are the influence promotions have on the actual store choice, i.e. that consumer switch stores to profit from a certain promotion. Indirect effects are the influence that promotions have on the sales in another store, without causing store switching. This is caused by consumers visiting various stores and increasing their sales for a certain brand in store x (caused by a promotion), with the subsequent decrease for the same brand when they visit store y (Bucklin and Lattin 1992). Direct store switching occurs especially when households use outside store-cues which leads to decreases in sales for competing stores (as opposed to "indirect "autonomous cross-shopping where results are mixed). While both effects help the retailer to increase sales without causing within-store substitution, direct effects are more favorable, as they drive more customers to a certain store, increasing the overall store's market share instead of just the market share in a certain (promotion supported) category.

As Uniform Across the Board Promotions are affecting the entire product portfolio of a store, they generally cause outside-store cues, and are hence likely to also cause direct store switching. While various studies agree that promotion induced store switching exists, they differ on the magnitude of this effect, which can partly be attributed to category specific effects. Kumar and Leone (1988) find that within-store substitution rates are by a factor of between two and three times higher than promotion induced store switching, while Ailawadi et al. (2006) find that store switching only makes up for 45 % (Ailawadi et al. 2006) of a brands temporary sales increase (with the rest coming from within-store substitution) (Ailawadi et al. 2006; Kumar and Leone 1988).

¹ In the case analysed in this thesis this might be somehow different, as products from the vertically integrated retailer are not available in any other store or chain.
Other Effects of the Short-Term Sales Increase

In addition to the impact on the actual size of the sales bump, promotions also affect the consistence of the "sales bump", i.e. what products will be purchased and by whom.

One general notion in retail, which also highlights the advantages of direct versus indirect effects in store switching, is that price promotions of a certain brand or category also stimulate the purchases of their non-promoted complements. Examples in research point to how for example promoted spaghetti sauce causes an increase in sales of the non-promoted spaghettis (Walters 1988). The degree of this effect varies strongly by brand or category (Mulhern and Leone 1991). Related to overall store sales, the promotional bump hence consists partly of incremental sales from the promoted brand but partly also of sales from the complementary products (net of less sales for the substitutes as previously discussed).

Another particularity of the sales bump is that sales in promotional periods can not necessarily be attributed to the regular and established customers a store, as promotions affect the customers mix visiting a store (e.g. more cherry pickers, less repeat purchases). Understanding the impact of the promotion on the customer group, will be important when discussing the long-term impact of price promotions (Anderson and Simester 2004).

2.3.1.2 Post-promotion Dip

The post-promotion dip or the questions whether "there is a trough after the deal" (Blattberg et al. 1995, p. 127) is one of the central topics in short-term promotional research and closely related to the previously discussed "*sales bump*".

When discussing post-promotion dips, this section focuses on four core questions:

- 1. Do post-promotion dips really exist (and what is the post-promotion dip paradox with regards to store-level data)?
- 2. What causes post-promotion dips?
- 3. What influences the degree of post-promotion dips?
- 4. What is the role of post-promotion dips for Uniform Across the Board Promotions?

Do Post-promotion Dips Really Exist?

The existence of post-promotion dips have among others been empirically shown by Gupta (1988) (*also discussed in paragraph on sales bumps*), who has decomposed the sales increase during a promotion and has found that 14 % of it is due to purchase acceleration, while 2 % is due to consumers buying more during a promotion (stockpiling). Hence 16 % of the promotional sales increase would be at the expense of other period's sales. Others such as Grover and Srinivasan (1992) find this effect to be up to 25 %. While Bell et al. (1999) on average find a similar degree (26 %) for sales being shifted from non-promotional to promotional periods, they observe a wider range of results among the 13 product categories they have observed (6–51 % of sales increase were attributable to stockpiling or purchase acceleration) (Bell et al. 1999; Grover and Srinivasan 1992; Gupta 1988). These findings support earlier research, which shows that consumers are doing sophisticated planning of purchases and adjust their buying behavior accordingly (Blattberg et al. 1981; Krishna 1992).

Even-though above research has found evidence for post-promotion dips in household-level data, earlier results analyzing store-related data is mixed. In addition to the above mentioned research from Neslin et al. (1985) as well as Jain and Vilcassim (1991) have found a dip following promotions, while Vilcassim and Chintagunta (1995) have found no trough.

Interestingly, Grover and Srinivasan (1992), who had found acceleration effects on household level data, have not found any effects on store-level data. Understanding this paradox and showing post-promotion effects also for store-related data, which is the type of data more managers base their analysis on (Bucklin and Gupta 1999), is important, in order to understand, whether the acceleration and stockpiling phenomenon really exists (on an aggregated basis). Neslin and Schneider Stone (1996) provide arguments, why post-promotion dips might not be observed in store-level data and why even the observation in household-level data might be complicated.

Later research addresses some of those issues, by adjusting the methodology to account for several of those factors hiding post-promotion troughs. Van Heerde et al. (2000) introduce three possible models, with which even complex post-promotion dips can be detected (see Chap. 4 for more details). These results are supported by later research from van Heerde et al. (2004), Macé and Neslin (2004) and van Heerde and Bijmolt (2005) who all find post-promotion dips in store-level data.

What Causes Post-promotion Dips?

Post-promotion dips could occur due to purchase acceleration, which can assume two forms: consumers buying *more* during a promotion (*"stockpiling"*) or them buying *earlier* i.e. shortening inter-purchase time. Neslin et al. (1985) have found that purchase acceleration is more often shown through stockpiling than through shortened inter-purchase time (e.g. Blattberg et al. 1995; Macé and Neslin 2004; Neslin et al. 1985).

What Influences the Degree of Post-promotion Dips?

The effectiveness of promotions and also the degree of acceleration and hence postpromotion dips can vary according to different product, category, promotion, store and customer characteristics (see e.g. Bell et al. 1999; Hoch et al. 1995; Macé and Neslin 2004). Related to product and category characteristics, post-promotion dips are greater for higher value and more established products with a higher general market share, which is explained by Macé and Neslin (2004, p. 346) as consumers "use promotions to ensure inexpensive supply (and consumption) of their favorite brands". Furthermore the storability of products and general selection within one product category (availability of different sizes) are positively related with the size of post-promotion dips (Macé and Neslin 2004). For products or categories with higher inventory holding costs and lower stock-out costs, consumers tend to hold fewer inventories, causing smaller sales bumps and also smaller post-promotion dips (Gönül and Srinivasan 1996).

Under the term *promotion characteristics*, the type of price promotion, advertising support and the general promotion schedule of a retailer, category or brand is summarized. Advertising of price cuts can increase quantity purchase and decrease inter-purchase time (i.e. accelerate purchases), leading to stronger post-promotion dips. Coupons also cause post-promotion dips, while those are smaller than those for advertised price cuts, as purchases are generally not accelerated through coupons (Neslin et al. 1985).

Krishna et al. (1991) have found that consumers form expectations on the future availability of coupons and promotions, which influence their purchase behavior. Results on the impact of predictability of promotions on the post-promotion dips are mixed. More frequent promotions have according to Macé and Neslin (2004) a positive impact on post-promotion dips as the consumer may learn to stockpile and decelerate purchases (deal-to-deal buying), which is contrary to earlier research, in which less predictable promotion schedules lead to the notion that there is a need to stockpile in order to make it to the next promotion (and ensure low cost availability of the product) (Meyer and Assuncao 1990). A possible explanation for this is, that consumers are conditioned to "lie-and-wait" for even more attractive promotions (Jacobson and Obermiller 1990; Mela et al. 1998) and that the two possible effects, stockpiling only during promotions and foregoing promotions hoping for more attractive ones, lead to the mixed empirical results.

Regarding customer characteristics especially older customer with bigger families that possess a car cause the largest post-promotion dips (Macé and Neslin 2004). One potential reason is that the availability of a larger car decreases inventory costs while bigger families might increase stockout costs, which would then be in line with the findings of Gönül and Srinivasan (1996). Customer loyalty does not necessarily lead to higher post-promotion dips, as findings have shown that this depends on the product category (Neslin et al. 1985). What is the Role of Post-promotion Dips for Uniform Across the Board Promotions?

As for traditional promotions, post-promotion dips decrease the net effectiveness of UABPs. There is no evidence, why the mere existence of post-promotion dips should be different than for traditional promotions, while the degree, to which they occur, could be affected by the breadth of the promotion. The claim that "every article is discounted" could increase the predictability of the promotion or rather the predictability of finding the desired article on promotion. Consumers could hence as per Macé and Neslin (2004) learn to stockpile and decelerate purchases, especially when frequently confronted with UABPs. Due to the generally higher predictability (of finding the desired article on promotion) of UABPs compared to traditional promotions, the post-promotion dip should be larger. This deal-to-deal buying for UABPs which is favored by the storability of goods will be closer analyzed in Chap. 3, using a dataset from a value fashion retailer.

2.3.1.3 Pre-promotion Dip

Even though the marketing literature has covered post-promotion dips i.e. anticipatory effects of promotions as early as the late 70s, pre-promotion dips or deceleration is not as well studied. Doyle and Saunders (1985) have been among the first, to stress the importance of lead effects when evaluating sales promotions and have found that those lead effects can be as important as lagged effects.

The marketing literature shows, that consumers form price expectations. When consumers anticipate marketing changes, e.g. if they expect significantly lower prices in the future, they react by adjusting their behavior and potentially deferring their purchases which will cause a pre-promotion dip (Kalwani et al. 1990; Winer 1986; van Heerde et al. 2000). Gönül and Srinivasan (1996), conducting research using house-hold level data for disposable diapers, find that consumers might defer purchases when they expect a coupon or promotion to be available in the next period. The likelihood of the deceleration and hence also the degree of the pre-promotion dips depends on the level of stockout-cost. When there is a reasonable substitute for a good), the purchase probability in a pre-promotion period decreases further.

Furthermore consumers assign a higher probability to coupon and promotion availability if there is no coupon available in the current period (Gönül and Srinivasan 1996). Mela et al. (1998) have found in household panel data for frequently purchased non-food products, that the formation of expectations is further influenced by the number of promotions. An increased availability of promotions (promotion rate) lets consumers decrease their baseline purchases in non-promotion times, as they expect to be able to purchase at even lower prices in the future. This holding-out further contributes to the existence of pre-promotion dips. In line with their findings when analyzing store level data for the existence of post-promotion dips, van Heerde et al. (2000) also find that pre-promotion dips exist.

As for their work on post-promotion dips, Macé and Neslin (2004) have found that pre-promotion dips are larger for frequently promoted products that have a higher share of wallet for the consumer. Moreover, they have found that storability and availability of different sizes has a positive impact on pre-promotion dips. Pre-promotion dips are further negatively correlated with age and income, which are the only relevant demographic effects for pre-promotion dips. As for post-promotion dips the degree to which pre-promotion dips exist for UABPs will be further discussed in Chap. 3.

2.3.2 Permanent/Long-Term Effects of Price Promotions

As stated earlier, the original Uniform Across the Board Promotions have been very successful in increasing short-term sales for the DIY retailer Praktiker. However they proved less efficient in the long-run and even worse, various experts claim that UABPs have caused severe problems for the retailer.

When assessing whether a certain type of price promotion campaign is beneficial for a retailer, it is hence critical to also understand the permanent effects on sales and consumer behavior. Specifically it is important to understand three interdependent areas of promotional impact. (1) whether the impact promotions have on sales and purchase incidence changes over time, (2) whether consumers become more price sensitive over time, which also affects their non-promotional buying behavior. This chapter will first review the literature on permanent effects on sales and purchase incidence and second discuss the impact price promotions have on brand or store loyalty. Last past research on how promotions affect the long-term price sensitivity and hence likelihood to stockpile of consumers is being reviewed. These topics give a direction, on how UABPs likely influence the expected mid-term sales bump, potential adjustment effects as well as consumer's attitude towards a brand or retailer and subsequently their long-term behavior. All aspects will be specifically discussed for UABPs in later chapters of this thesis (see Chaps. 4 and 5).

2.3.2.1 Impact on Long-Term Baseline Sales and Purchase Incidence

Overall the research on the long-term effects of price promotions is an area which according to Blattberg et al. (1995) is "the most debated in the promotional literature and one for which the jury is still out" (Blattberg et al. 1995, p. 127). The starting point in this discussion is what can be considered as "long-term", noticing, that the definition of long-term effects and adjustment effects are often quite similar. Mela et al. (1998) define long-term promotional impact as "the cumulative effect of previous promotional exposures (over quarters or years)"

(Mela et al. 1998, p. 250), while Dekimpe et al. (1999) find in their research, that long-term effects on sales can be defined as mostly stationary effects. Stationary effects assume that lagged effects exist but sales will eventually return to their pre-promotion mean, i.e. sales cannot be permanently affected by promotions (Dekimpe et al. 1999). When discussing the long-term impact on sales in this thesis, both a long-term non-reverting mean effect on sales and consumer behavior (Mela et al. 1998) will be discussed as well as a more mid-term adjustment effect that might eventually prove to be mean reverting (Dekimpe et al. 1999). The empirical evidence on whether long-term promotional effects exist is however regardless of the definition of the time-frame, mixed.

Long-Term Negative Impact of Promotions on Baseline Sales

The theoretical background sits close to the before mentioned prospect theory (Kahneman and Tversky 1979), adaption-level theory and the concept of reference prices. The reference price theory suggests, that (noticeable) promotions should have a negative impact on longer term (non-promotional) sales, as consumers get used to a lower price and hence expect to purchase at this price in the future (e.g. Greenleaf 1995; Jacobson and Obermiller 1990; Lattin and Bucklin 1989).

Depending on the type of promotion the promotional gain is separated from the price in a different mental account (Mazumdar and Yun 1993). Non-price promotions (e.g. lotteries) are booked as a separate gain, whereas price promotions are combined with the respective price and are booked as a loss reduction (Diamond and Johnson 1990). Only the later influences the reference price for a product. For UABPs, the discount is not directly applied to the product but rather as a bonus on the check-out on the entire basket, which is why it would be unclear, whether this would be considered as a direct gain or rather a loss reduction. For price promotions, an increased promotional exposure decreases the reference value for the product category during non-sales periods and hence the difference between category value and reference value increases which reduces the likelihood of category purchase incidence (Bell and Bucklin 1999).

According to Kalwani and Yim (1992) the promotion frequency and the depth of price discounts have a significant effect on the price expectations of consumers. They have found that frequent and sizeable promotions change the reference price, while infrequent or small, i.e. only around the current reference price, do not change price expectations of the consumers (Kalwani and Yim 1992). Frequency is hence important for promotions as they are a moderator in the degree to which promotions might change the reference price of the consumer. Reducing the once introduced frequency of promotions can have a negative effect on market shares as the net price to consumers is increased (Ailawadi et al. 2001).

In addition to the reference price concept discussed above, future price expectations or expectation about coupon availability also make consumers defer purchases to subsequent periods (Gönül and Srinivasan 1996), which is why the effect a long-term promotion exposure of a household has on purchase incidence is significantly negative (Mela et al. 1998). These findings are in line with earlier research that has found that promotions have a negative effect on long-term consumer behavior, i.e. purchase probability (Blattberg and Neslin 1990). During promotions, the average quantity bought increases, which affects future incidence rates and provides further evidence to the "sit and wait" attitude induced by promotions, that negatively affect long-term baseline sales (Mela et al. 1998).

Long-Term Positive Impact of Promotions on Baseline Sales

A second stream of research however suggests that promotions make consumers buy more and consume faster, which indicates a positive long-term impact (Ailawadi and Neslin 1998). While Blattberg and Neslin (1990) had found an overall negative impact, they recognize in line with learning theory, that promotion-induced trial purchases, or trial visits to retail stores, might increase familiarity with a brand or store and result in future repeat purchases, which would have a positive long-term impact (Blattberg and Neslin 1990). This is in line with research stating that promotions can be used to shape brand loyalty, which increases repeat purchases and hence baseline sales (Rothschild and Gaidis 1981). However as the self-perception theory suggests, consumers who have bought during promotions are likely to attribute the purchase to the presence of the promotion rather than their actual brand or retailer preference (Dodson et al. 1978).

Long-Term Neutral Impact of Promotions on Baseline Sales

Dekimpe et al. (1999) are among a group of researchers who argue that permanent effects of promotions do not structurally change the overall baseline sales of a brand or retailer and even if such effects exist for selected categories, they are usually very small (Dekimpe et al. 1999; Lim et al. 2005; Nijs et al. 2001). However the composition of sales, i.e. how much is bought at what point in time might be affected, as Mela et al. (1998) argue that customers will buy more at fewer occasions. This has been reiterated by Pauwels et al. (2002) who see the issue in measuring net impact in the different promotion induced effects cancelling each other out. In line with previous research, they argue that there is a negative impact on incidence and a positive one on quantities, which is why long-term impact has been difficult to measure in articles that can be stockpiled. In case that this argument holds, promotions can in the long-run be attractive for brand managers as consumers stay out of the market for competitive products—but also out of the market for other complements (which is bad for the retailers)—given the reduced frequency of shopping trips (Bell et al. 1999).

Summary of Impact of Promotions on Baseline Sales

As stated initially, one cannot definitely conclude whether promotions have a longterm effect on baseline sales, nor can one pass judgment on the direction such an effect would have. While overall more literature exists, suggesting negative longterm effects of promotions, various studies find positive or negative effects for one product category, which does not hold for other categories. For example the positive effect from promotion-induced repeat purchases (Blattberg and Neslin 1990), would be small or non-existent for mature categories and would only likely occur for new product categories or customers new in an area with different retail stores (Gijsbrechts 1993; Mela et al. 1997).

Overall the conclusion is that one has to (1) carefully analyze the type of price promotion run (*see detailed discussion on UABPs*), as well as the extent (depth and frequency) to which it is run. These factors impact the search costs for the consumer as well as the possibility to anticipate the promotion and the likelihood of a change in reference prices. Furthermore, the (2) category characteristics and (3) customer characteristics will have to be considered to pass a definite judgment on whether a promotion campaign has an impact on the future baseline sales of a brand or retailer.

2.3.2.2 Impact on Promotion Sensitivity and Promotion Effectiveness

This chapter discusses whether the long-term exposure to price promotions has an impact on the customer's decision to act on future promotions, i.e. whether their promotion sensitivity increases. Derived from the reference price theory, the general concept of consumers comparing an observed price to a reference price holds for regular sales periods (baseline sales) and promotional periods. The delta between those two prices is the perceived value for the consumer. If increased promotions lower this reference price and hence the price delta, even a lower price observed during promotions, is less "special" than it might have been if no previous promotions had been run. This reduces the purchase probability also during promotions and hence increases promotion sensitivity (e.g. Helson 1964, Sawyer and Dickson 1984).

As stated above reference prices are not just influenced by the absolute amount of a promotion (see Sect. 2.2.1 Prospect theory/reference prices) but also by the frequency of a promotion. Krishna et al. (1991) have found that consumers are good in assessing the frequency of regular promotions. This finding is important as empirical research has found that the impact of a promotion is significantly larger, if the promotion is unexpected (Kalwani and Yim 1992; Lattin and Bucklin 1989).

Empirical research supports this theory, as Mela et al. (1998) find, that in the long-term an increased use of promotions makes consumers more promotion as sensitive, which means that consumers are less likely to react on a promotion as they "sit and wait" for an even better deal to come by. This is in line with Gönül and Srinivasan (1996), who find that also as a function of current consumer inventory

levels, consumers might forego current promotions if they expect to receive future coupons that give them the same or even higher promotional benefits (Gönül and Srinivasan 1996). This entire line of research suggests that if consumers can form a view on future promotions, they might adapt their future shopping behavior accordingly, making promotions less effective. It is hence important to make promotions, or specifically Uniform Across the Board Promotions, as random as possible.

2.3.3 Summary of Theoretical Background and Literature Review and Outlook of Research Contribution

The focus of this was on two general topics: First, is the discussion of the theoretical background (Sect. 2.2), that sets the basis for many of the hypotheses developed in Chaps. 3, 4, and 5. The second focus was the review of existing pricing literature (Sect. 2.3) to introduce and position UABPs accordingly and put the later results into context. The research described in Sect. 2.2 are mostly findings based on experiments carried out in various fields of research, while Sect. 2.3 consists mostly of empirical studies that are based on relatively large data samples, mainly in the FMCG category and based on individual brands/categories.

The theoretical background (Sect. 2.2) dealt with four core concepts of which the highlights will be recalled in this paragraph:

- Prospect theory and reference price concept (Sect. 2.2.1): Which states that consumers compare prices to an internal or external reference price. Subject to the deviation of the observed price and reference price, the consumer forms a view on whether she likes an offer or not. This reference price can change when the consumer is confronted with promotions or the UABP.
- Price search and transaction costs (Sect. 2.2.2): This theory that stems from the field of microeconomics, suggests, that a consumer continues to look for a cheaper price, as long as the search costs are lower than the expected savings from finding a better deal. Expanding this to transaction costs states that consumers are making trade-offs between higher transaction costs and a potentially lower price. This becomes relevant, when the flexibility and breadth of UABPs will be discussed.
- Behavioral learning (Sect. 2.2.3): A concept from the field of social psychology which discusses whether consumers learn to adapt their behavior if repeatedly being confronted with a similar stimulus. A concept very relevant for the evolvement of the consumer's behavior and long-term impact of UABPs.
- Attribution theory (Sect. 2.2.4): This theory also has it's background in social psychology and claims that that individuals try to explain the causes for certain observed behavior and events, which will become relevant when discussing whether the UABP might change the attitude consumers have towards a retailer (Chap. 5).

In addition to the theoretical background, the literature review (Sect. 2.3) focused on how promotions influenced the short-term purchasing behavior (Sect. 2.3.1), specifically whether sales bumps are being caused by promotions and how promotions affect store choice, category choice or brand choice. Furthermore the adjustment effects around any potential sales bumps have been discussed, i.e. whether consumers adopt their purchase behavior through stockpiling and anticipation (Sect. 2.3.1.1)—both short-term and adjustment effects will be separately discussed for UABPs in Chaps. 3 and 4. The literature review concludes with discussing any potential long-term effects that might arise from promotions (Sect. 2.3.2) and that will be touched upon in Chap. 4.

The literature review mainly discusses the impact traditional promotions have on store sales and consumer behavior, while up to this point no study has particularly addressed the impact of Uniform Across the Board Promotions, which is a field that will be addressed by this study. Specifically the following three chapters will deal with this and related questions:

- Chapter 3: What is the impact of UABPs on the short-term sales performance of a retailer? In which type of stores and locations do they work best and how does this compare to other types of promotions?
- Chapter 4: How do UABPs affect any potential adjustment effect, i.e. do they cause pre-and post-promotion dips and what is their impact on the long-term sales performance of a retail store?
- Chapter 5: What are the antecedents of UABPs on a household-level, i.e. what type of consumers are more or less likely to shop during UABPs? Apart from impacting sales—what else is impacted (e.g. customer satisfaction and loyalty) and do households change their purchasing.

References

- Ailawadi KL, Neslin SA (1998) The effect of promotion on consumption: buying more and consuming it faster. J Market Res 35(3):390–398
- Ailawadi KL, Lehmann DR, Neslin SA (2001) Market response to a major policy change in the marketing mix: learning from Procter & Gamble's value pricing strategy. J Market 65 (1):44–61
- Ailawadi KL, Harlam BA, Cesar J, Trounce D (2006) Promotion profitability for a retailer: the role of promotion, brand, category, and store characteristics. J Market Res 43(4):518–535
- Ailawadi KL, Gedenk K, Lutzky C, Neslin SA (2007) Decomposition of the sales impact of promotion-induced stockpiling. J Market Res 44(3):450–467
- Alba JW, Hutchinson JW (1987) Dimensions of consumer expertise. J Consum Res 13(4):411-454
- Anderson ET, Simester DI (2004) Long-run effects of promotion depth on new versus established customers: three field studies. Market Sci 23(1):4–20
- Beatty SE, Smith SM (1987) External search effort: an investigation across several product categories. J Consum Res 14(1):83–95
- Bell DR, Bucklin RE (1999) The role of internal reference points in the category purchase decision. J Consum Res 26(2):128–143

- Bell DR, Lattin JM (1998) Shopping behavior and consumer preference for store price format: why "large basket" shoppers prefer EDLP. Market Sci 17(1):66–68
- Bell DR, Chiang J, Padmanabhan V (1999) The decomposition of promotional response: an empirical generalization. Market Sci 18(4):504–526
- Blattberg RC, Neslin SA (1990) Sales promotion: concepts, methods, and strategies, 1st edn. Prentice-Hall, Englewood Cliffs, NJ
- Blattberg RC, Eppen GD, Lieberman J (1981) A theoretical and empirical evaluation of price deals for consumer nondurables. J Market 45(1):116–129
- Blattberg RC, Briesch R, Fox EJ (1995) How promotions work. Market Sci 14(3):122-132
- Bolton RN, Shankar V (2003) An empirically driven taxonomy of retailer pricing and promotion strategies. J Retailing 79(4):213–224
- Bucklin RE, Gupta S (1999) Commercial use of UPC scanner data: industry and academic perspectives. Market Sci 18(3):247–273
- Bucklin RE, Lattin JM (1992) A model of product category competition among grocery retailers. J Retailing 68(3):271–293
- Bucklin RE, Gupta S, Siddarth S (1998) Determining segmentation in sales response across consumer purchase behaviors. J Market Res 35(2):189–197
- Buzas TE, Marmorstein H (1988) Consumers' knowledge of supermarket prices: the effects of manufacturer and retailer promotions. Adv Consum Res 15(1):360–363
- Chaiken S (1980) Heuristic versus systematic information processing and the use of source versus message cues in persuasion. J Pers Soc Psychol 39(5):752–766
- Chiang J (1991) A simultaneous approach to the whether, what and how much to buy questions. Market Sci 10(4):297–315
- Chintagunta PK (1993) Investigating purchase incidence, brand choice and purchase quantity decisions of households. Market Sci 12(2):184–208
- Darke PR, Freedman JL (1993) Deciding whether to seek a bargain: effects of both amount and percentage off. J Appl Psychol 78(6):960–965
- Darke PR, Freedman JL, Chaiken S (1995) Percentage discounts, initial price, and bargain hunting: a heuristic-systematic approach to price search behavior. J Appl Psychol 80 (5):580–586
- Dekimpe MG, Hanssens DM, Silva-Risso JM (1999) Long-run effects of price promotions in scanner markets. J Econometrics 89(1–2):269–291
- Diamond W, Johnson RR (1990) The framing of sales promotions: an approach to classification. Adv Consum Res 17(1):494–500
- Dickson PR, Sawyer AG (1990) The price knowledge and search of supermarket shoppers. J Market 54(3):42–53
- Diller H (1984) Das Zielsystem der Verkaufsförderung, Wirtschaftswissenschaftliches. Studium (WiSt) 13:494–499
- Diller H (1999) Entwicklungslinien in Preistheorie und –management. Marketing Zeitschrift für Forschung und Praxis 21(H1):39–60
- Dodson JA, Tybout AM, Sternthal B (1978) Impact of deals and deal retraction on brand switching. J Market Res 15(1):72–81
- Doyle P, Saunders J (1985) The lead effect of marketing decisions. J Market Res 22(1):54-65
- Duncan CP, Olshavsky RW (1982) External search: the role of consumer beliefs. J Market 19 (1):32–43
- Folkes V (1988) Recent attribution research in consumer behavior: a review and new direction. J Consum Res 14(4):548–565
- Gedenk K (2002) Verkaufsförderung, 1st edn. Verlag Vahlen, München
- Gijsbrechts E (1993) Prices and pricing research in consumer marketing: some recent developments. Int J Res Market 10(2):115–151
- Gönül F, Srinivasan K (1996) Estimating the impact of consumer expectations of coupons on purchase behavior: a dynamic structural model. Market Sci 15(3):262–279

- Greenleaf EA (1995) The impact of reference price effects on the profitability of price promotions. Market Sci 14(1):82–104
- Grewal D, Marmorstein H (1994) Market price variation, perceived price variation, and consumers' price search decisions for durable goods. J Consum Res 21(3):408–418
- Grover R, Srinivasan V (1992) Evaluating the multiple effects of retail promotions on brand loyal and brand switching segments. J Market Res 29(1):76–89
- Guadagni PM, Little JDC (1983) A logit model of brand choice calibrated on scanner data. Market Sci 2(3):203–238
- Gupta S (1988) Impact of sales promotions on when, what and how much to buy. J Market Res 25 (4):342–355
- Helson H (1964) Adaption-level theory, 1st edn. Harper & Row, Oxford
- Hoch SJ, Drèze X, Purk ME (1994) EDLP, Hi-Lo, and margin arithmetic. J Market 58(4):16-27
- Hoch SJ, Kim BD, Montgomery AL, Rossi PE (1995) Determinants of store-level price elasticity. J Market Res 32(1):17–29
- Homburg C, Koschate N (2005) Behavioral Pricing-Forschung im Überblick Teil 1: Grundlagen, Preisinformationsaufnahme und Preisinformationsbeurteilung. Zeitschrift für Betriebswirtschaft 75(4):383–423
- Jacobson R, Obermiller C (1990) The formation of expected future price: a reference price for forward-looking consumers. J Consum Res 16:420–430
- Jain S, Srivastava J (2000) An experimental and theoretical analysis of price-matching refund policies. J Market Res 37(3):351–362
- Jain DC, Vilcassim NJ (1991) Investigating household purchase timing decisions: a conditional hazard function approach. Market Sci 10(1):1–23
- Kahneman D, Tversky A (1979) Prospect theory: an analysis of decision under risk. Econometrica 47(2):263–291
- Kalwani MU, Yim CK (1992) Consumer price and promotion expectations: an experimental study. J Market Res 29(1):90–100
- Kalwani MU, Yim CK, Rinne HJ, Sugita Y (1990) A price expectation model of customer brand choice. J Market Res 27(3):251–262
- Keng KA, Ehrenberg ASC (1984) Patterns of store choice. J Market Res 21(4):399-409
- Krishna A (1992) The normative impact of consumer price expectations for multiple brands on consumer purchase behavior. Market Sci 11(3):266–286
- Krishna A, Currim IS, Shoemaker RW (1991) Consumer perceptions of promotional activity. J Market 55(2):4–16
- Krishnamurthi L, Raj SP (1988) A model of brand choice and purchase quantity price sensitivity. Market Sci 7(1):1–20
- Kumar V, Leone RP (1988) Measuring the effect of retail store promotions on brand and store substitution. J Market Res 25(2):178–185
- Lal R, Rao R (1997) Supermarket competition: the case of every day low pricing. Market Sci 16 (1):60–80
- Lattin JM, Bucklin RE (1989) Reference effects of price and promotion on brand choice behavior. J Market Res 26(3):299–310
- Lattin JM, Ortmeyer G (1991) A theoretical rationale for everyday low pricing by grocery retailers. Working papers (research paper 1144) Stanford Graduate School of Business, 1–34
- Lichtenstein DR, Bearden WO (1989) Contextual influences on perceptions of merchant-supplied reference prices. J Consum Res 16(1):55–66
- Lichtenstein DR, Netemeyer RG, Burton S (1991) Using a theoretical perspective to examine the psychological construct of coupon proneness. Adv Consum Res 18(1):501–508
- Lim J, Currim IS, Andrews RL (2005) Consumer heterogeneity in the longer-term effects of price promotions. Int J Res Market 22(4):441–457
- Macé S, Neslin SA (2004) The determinants of pre- and postpromotion dips in sales of frequently purchased goods. J Market Res 41(3):339–350

- Marmorstein H, Grewal D, Fishe RP (1992) The value of time spent in price-comparison shopping: survey and experimental evidence. J Consum Res 19(1):52–61
- Mazumdar T, Monroe KB (1990) The effects of buyers' intentions to learn price information on price encoding. J Retailing 66(1):15–32
- Mazumdar T, Yun S (1993) Consumer evaluations of multiple versus single price change. J Consum Res 20(3):441–450
- Mela CF, Gupta S, Lehmann DR (1997) The long-term impact of promotion and advertising on consumer brand choice. J Market Res 34(2):248–261
- Mela CF, Jedidi K, Bowman D (1998) The long-term impact of promotions on consumer stockpiling behavior. J Market Res 35(2):250–262
- Metha N, Rajiv S, Srinivasan K (2003) Price uncertainty and consumer search: a structural model of consideration set formation. Market Sci 22(1):58–84
- Meyer RJ, Assuncao J (1990) The optimality of consumer stockpiling strategies. Market Sci 9 (1):18–41
- Monroe KB (1973) Buyers subjective perceptions of price. J Market Res 10(1):70-80
- Moore DJ, Olshavsky RW (1989) Brand choice and deep price discounts. Psychol Market 6 (3):181–196
- Mulhern FJ, Leone RP (1991) Implicit price bundling of retail products: a multiproduct approach to maximizing store profitability. J Market 55(4):63–76
- Naik PA, Raman K, Winer RS (2005) Planning marketing mix strategies in the presence of interaction effects. Market Sci 24(1):25–34
- Neslin SA, Schneider Stone LG (1996) Consumer inventory sensitivity and the postpromotion dip. Market Lett 7(1):77–94
- Neslin SA, Henderson C, Quelch J (1985) Consumer promotions and the acceleration of product purchases. Market Sci 4(2):147–165
- Nijs VR, Dekimpe MG, Steenkamp JB, Hanssens D (2001) The category-demand effects of price promotions. Market Sci 20(1):1–22
- Nord WR, Peter JP (1980) A behavior modification perspective on marketing. J Market 44 (2):36–47
- Ordóñez LD, Connolly T, Coughlan R (2000) Multiple reference points in satisfaction and fairness assessment. J Behav Decis Making 13(3):329–344
- Pauwels K, Hanssens DM, Siddarth S (2002) The long-term effects of price promotions on category incidence, brand choice, and purchase quantity. J Market Res 39(4):421–439
- Puto CP (1987) The framing of buying decisions. J Consum Res 14(3):301-315
- Rajendran KN, Tellis J (1994) Contextual and temporal components of reference price. J Market 58(1):22–34
- Rothschild ML, Gaidis WC (1981) Behavioral learning theory: its relevance to marketing and promotions. J Market 45(2):70–78
- Sawyer AG, Dickson PH (1984) Psychological perspectives on consumer response to sales promotion. Research on sales promotion: collected papers: report: 84-104, Marketing Science Institute, Boston, 1–24
- Seiders K, Voss G (2004) From price to purchase. Market Manag 13(6):26-43
- Skinner BF (1953) Science and human behavior, 1st edn. Free Press, New York
- Srivastava J, Lurie N (2001) A consumer perspective on price-matching refund policies: effect on price perceptions and search behavior. J Consum Res 28(2):296–307
- Stigler GJ (1961) The economics of information. J Polit Econ 69(3):213-225
- Tang C, Bell D, Ho TH (2001) Store choice and shopping behavior: how price format works. Calif Manage Rev 43(2):56–74
- Tellis GJ (1986) Beyond the many faces of price: an integration of pricing strategies. J Market 50 (4):146–160
- Thorndike EL (1911) Animal intelligence, 1st edn. Macmillan, New York
- Urbany JE (1986) An experimental examination of the economics of information. J Consum Res 13(2):257–271

References

- Urbany JE, Dickson P, Kalapurakal R (1996) Price search in the retail grocery market. J Market 60 (2):91–104
- Van Heerde HJ (1999) Models for sales promotion effects based on store-level scanner data. Doctoral dissertation, University of Groningen
- Van Heerde HJ, Bijmolt THA (2005) Decomposing the promotional revenue bump for loyalty program members versus non-members. J Market Res 42(2):443–457
- Van Heerde HJ, Leeflang PSH, Wittink DR (2000) The estimation of pre-and post-promotion dips with store-level scanner data. J Market Res 37(3):283–395
- Van Heerde HJ, Gupta S, Wittink DR (2003) Is 75% of the sales promotion bump due to brand switching? No, only 33% is. J Market Res 40(4):481–491
- Van Heerde HJ, Leeflang PSH, Wittink DR (2004) Decomposing the sales promotion bump with store data. Market Sci 23(3):317–334
- Vilcassim NJ, Chintagunta PK (1995) Investigating retailer product category pricing from household scanner panel data. J Retailing 71(2):103–128
- Walters RG (1988) Retail promotions and retail store performance: a test of some key hypotheses. J Retailing 64(2):153–180
- Walters RG (1991) Assessing the impact of retail price promotions on product substitution, complementary purchase, and interstore sales displacement. J Market 55(2):17–28
- Ward RW, Davis JE (1978) A pooled cross-section time series model of coupon promotions. Am J Agr Econ 60(3):393–401
- Weiner B (2000) Attributional thoughts about consumer behavior. J Consum Res 27(3):382-387
- Winer RS (1986) A reference price model of brand choice for frequently purchased products. J Consum Res 13(2):250–256
- Xia L, Monroe KB, Cox JL (2004) The price is unfair! A conceptual framework of price fairness perceptions. J Market 68(4):1–15

Chapter 3 Impact of Uniform Across the Board Promotions on Short-Term Store Revenues

3.1 Introduction and Context to Short-Term Effects of UABPs

As already briefly touched upon in Sect. 1.1, until 2003 retailers had relatively few options with regard to the design of price promotions due to a "federal promotion regulation" in Germany. Particularly, promotions were only allowed in short and specific timeframes and the type, width and depth of promotion was centrally regulated. For instance any form of coupons or uniform discounts were prohibited.

When this situation changed in 2003, after the regulation was altered and retailers were given more flexibility concerning price promotions—Praktiker, Germany's second largest DIY chain was among the pioneers in using this new flexibility. They became famous for offering their customers uniform discounts on the entire assortment, captured in the slogan "20 % off on any article in the store—except pet food". This uniform percentage discount on every article in the store was in fact the first of a new type of promotion in Germany, which is referred to in this thesis as "Uniform Across the Board Promotion (UABP)" and which is has been defined in Sect. 2.1.2 of this thesis (Bomsdorf et al. 2003).

Despite the growing importance and usage of UABPs in the German and European retail environment (see Sect. 1.1), this new form of promotion and its impact on the consumer has not been covered in the marketing literature thus far. This research gap is important, because the performance impact of UABPs is far from clear as Praktiker faced severe financial problems which the German business press has at least partly attributed these difficulties to the excessive use of UABPs (Handelsblatt/Reuters 2012). Against this background, it is the goal of this thesis, to investigate the performance impact of UABPs with the first step being the analysis of the short-term impact UABPs have on sales, which will be discussed in this chapter.

The key dependent variable is store level performance. This focus distinguishes this research from the more established research on brand promotions. In particular, as UABPs are applied to all products at the same time and to the same degree, within store cannibalization effects that are typical for brand promotions are likely to be negligible. Therefore it becomes possible to clearly attribute the impact of UABPs on overall retail store sales. This being said, there is the possibility of different articles having different price elasticities and hence there might be a change in the composition of the shopping basket. This is however not a focus of this study and as this thesis focuses on overall store sales the net effect on the store is assumed to be zero and will be neglected going forward.

Moreover, it becomes relevant to understand how the impact of UABPs is influenced by store characteristics. The latter is a central topic as UABPs are often used by larger retail chains with a heterogeneous store portfolio (e.g. Praktiker, Esprit, and Replay). This chapter discusses the impact of the following store variables on the success of UABPs: (1) the location of the retail stores, (2) the net trading area, (3) the age, and (4) the general condition of the store.

To analyze the impact of UABPs and the role the store trading characteristics play weekly store level data from a German value fashion retailer has been used. The dataset contains weekly sales level data from 747 stores (S = 747) over a time span of 225 weeks (T = 225) in the period from 2008 to 2012. Methodologically a Markov Chain Monte Carlo simulation in an integrated hierarchical Bayes Model is being used to estimate the impact of promotions on retail store sales as well as the influence of the retail store characteristics on the promotional impact.

The main contribution to research made in this chapter is twofold: First this chapter introduces Uniform across the Board Promotions to the marketing literature and defines their key characteristics based on a qualitative pre-study. Second the short-term sales impact of UABPs is analyzed based on a proprietary data sample and compared with other forms of promotions and advertising. Third the store characteristics of a retailer that influence the impact of UABPs on the short-term sales impact are explored and compared with the impact those store characteristics have on other promotions and advertising.

3.2 Conceptual Framework

3.2.1 Definition and Characteristics of Uniform Across the Board Promotions

To arrive at a clear understanding of what constitutes a "Uniform Across the Board Promotions" a small qualitative pre-study is conducted. In particular promotion materials have been collected over the last years and more systematically in the fourth quarter of 2012 using internet search engines (content and image search) based on the material gathered before. In addition the topic has been discussed with retail managers and experts that have practical experience with those campaigns and/or have launched them in their companies. As a result, a small database containing information on more than 90 UABPs in the period between 2009 and

2012 has been built. Table 1.1 (Chap. 1) provides information for a representative subset of promotions included in this database.

Based on this qualitative pre-study, one can define uniform across-the-board promotions along four characteristics which have been briefly introduced in Sect. 2.1.2: (1) They are store-wide promotions on the entire assortment, sometimes with minor exclusions of certain categories. (2) They involve an undifferentiated discount on the entire assortment, i.e. no bigger or smaller discounts on any article. (3) The discount is expressed relative to the article prices, i.e. as a percentage to the absolute value. (4) They are only valid for a limited period of time.

Compared with other price promotions, the first three characteristics combined are differentiating criteria. In particular, other promotions are normally not applied to the entire assortment and even during sales periods, some articles are excluded or the depths of the discounts vary. Table 1.1 further shows that UABPs are not restricted to certain industries, sales channels or article types. They do vary in their offered promotion depth, duration of the offering and restrictions such as the obligation to use a coupon or minor restrictions in the choice the customer has.

Conceptually, UABPs aim at attracting efficient shoppers as well as cherry pickers, as they combine elements from the every-day-low-price and the HiLo pricing strategy. Providing the customers with lower prices on the entire assortment, and hence not restricting the choice the customer has, is an EDLP element and attracts efficient shoppers, while only doing this temporarily and sometimes supported by featured advertising is an element from the HiLo strategy, which aims at attracting cherry pickers (Bell and Lattin 1998; Hoch et al. 1994).

Another advantage of UABPs for the retailer is the likely absence of within-store substitution effects resulting in higher overall store sales. This distinguishes UABPs from traditional brand promotions that often only fractionally increase store sales due to within-store substitution (Gupta 1988; Sun 2005; Sun et al. 2003; van Heerde et al. 2003). For instance, Kumar and Leone (1988) find that within-store substitution rates are by a factor of between two and three times higher than promotion induced store switching (Kumar and Leone 1988). Directionally consistent, but different in the magnitude, Ailawadi et al. (2006) find that within-store substitution and temporal shifts account for 55 % of the sales bump, while induced store switching only makes up for 45 % (Ailawadi et al. 2006). The different magnitude of the individual drivers for the sales bump could be caused by the differences in categories researched by Kumar and Leone (1988), who have only focused on disposable diapers in ten stores, while Ailawadi et al. (2006) have conducted their research on a large span of drugstore articles.

Given the characteristics of UABPs described above and in the previous chapters, one could expect that the sales impact for the retailer (from store switching) is even stronger for UABPs than for traditional promotions, as it allows consumers to not have to switch to a less preferred brand while still profiting from the promotion. The theory of information economics underlines the advantage of UABPs, as customers always put the search costs in relation to the expected savings. Given the promotion breadth of UABPs, the probability of finding the desired article at a discount, and hence the expected savings (given the same depth of promotion) is



Fig. 3.1 Illustration of research framework

higher for UABPs than for other promotions. In consequence the willingness to incur higher search costs, i.e. drive to a retailer further away, should be higher than during normal promotions (Metha et al. 2003; Stigler 1961).

3.2.2 Introduction of Store Level Antecedents

As UABPs affect the total store sales, it is essential for a retailer to understand the store characteristics that influence the success of UAPBs. This area has been touched in previous research as for example Montgomery (1997) has dealt with customizing pricing according to specific retail settings in order to maximize gross profit in a heterogeneous store portfolio. Macé and Neslin (2004) have analyzed the impact of store-trading-area characteristics on the success of brand specific price promotions. They focused in detail on the store customer parameters such as education, ethnicity, age, income profile, car ownership etc. and have found that pre- and post-promotion dips are influenced by those parameters.

Please refer to Fig. 3.1 for an illustration of the research framework. Variables listed therein are described in Tables 3.1 and 3.2.

Variable	Description
Uniform Across the Board Promotions	All articles in the store are discounted with uniform discounts for a specific promotion period. For example 25 % on all articles in the store. The discount is being granted at the checkout and the UABPs are restricted to a certain time period
GlamourShoppingWeek	Cooperation with a German magazine to grant their readers uniform discounts within a certain week. Promotion form is a mix of UABP and advertising and given the broad distribution of the advertising not exactly comparable with the regular UABPs, which is why it is measured separately
WinterSales	Item specific clearance sales at the end of the winter season. Gener- ally lasts for around 8 weeks—discounts might increase in the later weeks of the sales period. Depth of promotion varies by article
SummerSales	Item specific clearance sales at the end of the summer season. Generally lasts for around 6 weeks—discounts might increase in the later weeks of the sales period. Depth of promotion varies by article
Buy2Get1Free	When selecting three articles the cheapest of the three articles is for free
Promotion days	Assortment specific sales to promote a sub-category with special sales offers. Examples are Jeans days or Lingerie days
Dot days	Specific items in a store a marked with different color dots—each color represents a certain percentage discount on the "dotted" article
Mailing	Existing customers receive a mailing with special discounts on certain articles
Leaflet	Leaflets are distributed to all households in the proximity of a store. Special article based offers are promoted
TV	TV campaigns are run on national television to promote the retailers brand and certain articles

Table 3.1 Promotion types and advertising types

3.3 Method

3.3.1 Store-Level Promotional Data for Short-Term UABP Impact

The analysis is based on weekly store data for 747 retail stores of a German value fashion retailer. Out of the dataset provided by the retailer individual stores had been eliminated due to obvious measurement errors (e.g. no sales history for 10 days, than 1 day with extremely high sales). The time span for the data is 225 weeks in the period 2008–2012. The weekly store sales data has been de-seasonalized and de-trended for a linear temporal trend using a regression model similar to the one by Abraham and Lodish (1993) (see Appendix A) in which weekly sales have been adjusted for seasonal fluctuations and a positive temporal linear trend resulting from same store sales growth over the time period.

Characteristics	Description	Mean	SD	Min	Max
Location	The retail stores are located in different types of locations: Shopping center, strip mall, inner city, sublet from hypermarket	Shopping mall (zone) (n = 6) marke	g center (n = 470 (n = 86) (03), subl et (n = 5	(n = 69)); common comm enter common comm common common); strip nercial ity per-
Size	Trading area of the store in square meters	560.9	127.0	147.0	1,238.0
Age of store	Age of the store since opening (in years)	11.65	5.25	5.00	29.00
Age since refurbishment	Some stores had a make-over. This mea- sures the age of the store since make- over or opening (in years)	3.92	4.31	2.00	29.00
Assortment female	Share of assortment (sales) per store in female category	50 %	3 %	37 %	74 %
Assortment male	Share of assortment (sales) per store in male category	17 %	3 %	0 %	25 %
Assortment kids	Share of assortment (sales) per store in kids category	21 %	3 %	11 %	36 %
Assortment accessories	Share of assortment (sales) per store in accessories category	13 %	2 %	4 %	17 %
Value per article	Average value per article sold (in EUR)	7.91	0.49	4.22	9.09

Table 3.2 Store characteristics

For the promotions dummy variables have been introduced that indicate whether a certain promotion has been applied in the retailer's stores in a certain week according to the central marketing plan of the company. The store characteristics have been provided by the company for all 747 stores.

3.3.2 Methodology to Analyze Short-Term Promotional Impact

In the marketing literature there are various different models proposed to analyze the impact of promotions. The majority of these models deal with the promotional impact on certain brands and categories, i.e. their specific price elasticity, rather than the impact on the overall retail store sales (e.g. Bell et al. 1999; Lim et al. 2005; Nijs et al. 2001). A popular approach to model the short-term promotional "sales bump" caused by promotions and advertising would be the use of the SCAN*PRO model which is widely accepted among managers and also often used in academics (e.g. Hoch et al. 1995; Montgomery 1997; Montgomery and Rossi 1999; van Heerde et al. 2000) but which would have to be adjusted similar to the use in Blattberg and George (1991) to focus on store wide promotions rather than on brand specific promotions. To measure the impact of the various promotional and advertising campaigns in this example would let each store have its own OLS regression model and then in a second step perform a cross-sectional regression in which the relevant parameters (e.g. impact of UABP, impact of 2 for 1 promotion etc.) would be treated as dependent variables and the store characteristics as their antecedents.

However, especially in heterogeneous store environments the resulting store estimates might have large standard deviations in the promotional and advertising parameters and larger error terms across stores (Montgomery 1997). Assuming non-heterogeneity across stores, and pooling stores, could however be misleading from a managerial point of view and the expected promotional impact of the UABP would be over- or underestimated (Andrews et al. 2008).

To overcome this issue, a hierarchical Bayes (HB) model is being applied to analyze the promotional impact for all stores simultaneously and to derive the store characteristics that have the strongest influence on the promotional sales impact. In recent years, the use of hierarchical Bayes models has become more common as usage constraints—particularly computational time—were overcome through advances in the power of PC's while standardized software packages (e.g. plug in for R) have become easier accessible (Rossi et al. 2005).

3.3.3 Hierarchical Bayes Model

3.3.3.1 General Application of a Hierarchical Bayes Model

The term *hierarchical* stems from the fact that the model consists of multiple levels. This way the methodology accommodates nested data structures. In this study, two levels are considered. Weekly store sales and weekly store promotional activity at the lower level are nested in stores with specific store characteristics at the higher level. With hierarchical Bayes models the distribution of the effects of the different promotional activities (i.e., beta coefficients) across stores at the lower level is thought to follow a normal distribution [see also (3.3a)]. In other words, for each promotional activity a separate effect is estimated for each store. At the same time, it is modeled whether the way the store is set up affects these store-specific estimates.

The method requires initial betas ("priors") for each promotional activity. Priors can be "informative", i.e. based on prior knowledge or analysis of the objects or they can be more or less random "guesses". In this case the informative betas were estimated using OLS regressions for each individual store. These initial estimates, i.e. the individual betas as well as the mean and covariances of their distribution are then continuously updated using a process called "Gibbs sampling". This estimation process for each beta is always conditional on the other parameters (i.e. betas) estimated in the same "draw". Eventually these estimates converge to a stable and correct estimate for all individual betas (*posterior* distribution). As the results from every iteration/draw have been saved, every beta has an entire distribution of results. This process shows another difference to OLS regressions, as HB models fully account for uncertainty and provide results conditional on the observed data (Rossi et al. 2005).

In other words, the prior is the belief one has in a certain beta or parameter, without knowing the data. The posterior is the belief in the distribution conditional on the observed data. The likelihood function is the probability that the data can be generated by the model, believing in the parameters, while the evidence is the distribution of the data. Basically one believes in a parameter distribution according to a function, and by testing this against the data, the original belief will be continuously updated in order to fit with the observed data (Kruschke 2011, p. 301):

 $posterior = likelihood \times prior/evidence$

3.3.3.2 Application of a Hierarchical Bayes Model to Estimate the UABP Impact

In this example, a hierarchical Bayes model is being applied, in which the promotional impact and the store characteristics are modeled in one integrative framework, without treating each store as separate and unrelated. The integrative framework of the HB model allows estimating the immediate effects of the promotions and advertising and relating those effects to the individual store characteristics. While under certain *prior* assumptions the integrated HB model could yield results identical to those from individual store model regressions (Montgomery 1997) the method introduced by Blattberg and George (1991) is being applied, in which the results from the parameters of the individual OLS models are used as samples from a common prior distribution. These results, i.e. their means and co-variance matrix, will form the prior for the analysis.

The weekly sales for store

42

j'(j = 1, 2, ..., 747) in week t (t = 1, 2, ..., 225) is denoted by the vector

$$\mathbf{S} = (\mathbf{s}_1, \mathbf{s}_2 \dots, \mathbf{s}_{747}) \tag{3.1}$$

where s_i^1 equals sales of store *j* in week 1, and

$$x_1 = (x_{1,1}, x_{1,2}, \dots, x_{1,747}) \tag{3.2}$$

are matrixes of the individual promotional $(X_1, X_2, X_3, \dots, X_{11})$ dummy variables described in (3.1).

 $x_{i,j}^{1}$ is the value of promotional campaign i (i = 1, 2, ..., 11) in store j in week 1. The first column in the X matrix is set identical to 1 ($x_{j}^{1} = 1$) to account for an intercept. Using the standard regression notation for the sales response model, one can write the standard hierarchical model for all j stores as

$$S_{j,t} = \sum_{i=1}^{11} \beta_{j,i}^t x_{j,i}^t + \varepsilon_j, \varepsilon_j \sim N(0,\sigma)$$
(3.3)

And priors:

$$\beta_i \sim N(\gamma, V_\beta) \tag{3.3a}$$

$$\gamma \sim N(\overline{\gamma}, A^{-1})$$
 (3.3b)

$$V_{\beta \sim IW(v, V)} \tag{3.3c}$$

The priors (3.3a), (3.3b), and (3.3c) are independent. The informative priors of β could be input either directly or by estimating those using a least squares estimate (Boatwright et al. 1999). As described above, the latter approach was chosen here. The resulting covariance matrix of the β -prior was obtained from the empirical regression results of running 747 separate OLS regressions for each store in a method based on Montgomery (1997) [see also Casella (1985), Carlin and Louis (2008), for similar approaches].

The second stage priors of the β -priors are set diffuse to $\gamma \sim N(\bar{y}, A^{-1})$ and $V_{\beta} \sim IW(v, V)$ to allow for broad coverage of the parameter space (Rossi et al. 2005). This broad coverage is important to not restrict the results of the HB model by setting priors limited to a certain parameter space. As suggested by Rossi et al. (2005) the inverse Wishart distribution was used as prior distribution for the variance of the beta coefficients with a three degrees of freedom.

The error term is normally distributed with $\varepsilon_j \sim N(0, \sigma)$ and σ is modeled to be $\chi^2 - distributed$ with three degrees of freedom (df = 3). These degrees of freedom and the choice of the Wishart distribution were based on the default setting in the R package "*bayesm*" (Rossi 2012; Rossi et al. 2005) that was used for the analysis.

The hierarchical Bayes model treats γ_j as the effect of the store parameters z on the promotional parameters β_i^i :

$$\gamma_{i} = \varphi z_{i} + u_{j}; u_{j} \sim N(0, V_{\gamma})$$

$$(3.4)$$

 z_i is the matrix containing the store parameters which are listed in Table 3.2.

In this model setup, the posterior distribution depends on both types of variables, store and promotional variables. The parameters were simulated in a Markov Chain Monte Carlo approach (MCMC) using a Gibbs sampler with 6,000 iterations, and a burn in of 500 iterations. The results converged nicely already after a couple thousand iterations.

3.3.4 Model Variables

This paragraph deals with the specifications of variables used in the hierarchical Bayes model. To analyze the impact of UABPs versus other types of promotions and in various store settings, one can distinguish between (1) promotional variables, which relate to nine types of promotions and advertising used by the German value fashion retailer (see Table 3.1) and (2) six different store characteristics (see Table 3.2).

3.3.4.1 Promotional Variables

As described above, the central marketing plan served as input for the exogenous promotion and advertising variables described below. Note that the retailer runs the same promotion and advertising campaigns in all stores in a given week. Promotions and advertising are usually not run in the same week, nor are different types of promotions run at the same time. A summary of promotions are listed in Table 3.1.

The average promotion depth for each promotion campaign over time varies due to external factors such as inventory levels and competitive market behavior and has been on average around the same level as the promotion depth the UABP offers (~20 %) within the research period. In addition to UABPs six other types of promotions as well as three types of advertising have been analyzed. Please refer to Table 3.1 for further details.

3.3.4.2 Store Characteristic Variables

The performance of the individual store promotions is tested against various store parameters to determine whether the sales impact of UABPs (or other promotional forms) is influenced by store characteristics and if so, in which retail setting the relevant promotion type might work best. The analysis will be performed on 747 fashion retail stores located in Germany. The store trading area characteristics analyzed include the following:

1. *Location of the store* which serves as a proxy for the related search and transactions costs for the customer. The location mix is heterogeneous and the stores of the fashion retailer have been clustered in five different location types by the company according to typical characteristics such as footfall, average commute for the customer and general outlay of the premises. The five location types are comparable regarding the size of the trading areas, interiors and product carriers and show the following typical characteristics: (a) strip malls/commercial zones: Typically lower footfall, out of city locations, which require longer commutes for the customer, typically accessed by the retailer's customer to shop for specific items, as the associated transaction costs are higher; (b) shopping centers/inner city: Comparably high footfall, convenient locations for the customer; (c) hypermarket sublet: Typically stores in larger hypermarkets that carry food but also a fashion and non-food assortment with moderate footfall and often non-prime locations within the hypermarkets. Dummy variables are being used for *location* with *inner city* used as reference point.

3.4 Empirical Results

- 2. *Size of the store* refers to the net trading area measured in square meters. While all stores carry the same core product portfolio, larger stores can offer a slightly broader and deeper choice in products when it comes to "sizes and fits". The size of the store hence serves as a proxy for whether UABPs work relatively better, if customers are given a broader product selection.
- 3. *Age of the store* is a proxy for how established a store is in a community—this is assumed to also directionally affect the number of people who know of the store and who actively follow its promotions. Given that the retailer had recently refurnished all stores, this does not reflect the shape a store is in.
- 4. The age since refurbishment, is hence a better proxy for the condition of the store. Refurbishments typically include a major refurbishment of the store including floors, walls, product carriers etc. This retailer has spent considerable funds to refurbish its stores, which is why this criterion is relevant to be considered.
- 5. Gender focus, is rather a retail assortment specific characteristic, which is measured as the percentage of weekly sales being made in any of the three categories: female, male or kids. A 50 % female share would hence mean that during a week 50 % of the retail value of sold articles comes from the category "female". A third product category referred to as "accessories" serves as the reference point. The analysis of these variables aims at understanding whether the UABP works particularly well with a certain gender in the discussed retail setting.
- 6. *Value per article is* used to analyze the impact a higher average price of articles sold in the store has on promotions. It has to be noted, that the core assortment is identical in every store and as seen in Table 3.2 the standard deviation for this variable is relatively small, which is why any impact would only be directional and would have to be further researched.

3.4 Empirical Results

3.4.1 Empirical Results: Short Term Performance Outcomes

Table 3.3 shows the impact the promotions have on the short-term sales. The impact is shown as normalized, de-trended and de-seasonalized sales as described above.

The first key observation is that in the integrated model all types of promotions have a positive impact on retail store sales. On a *by individual store* basis, one can however observe that UABPs strongly impact sales with a mean of 1,533 and a standard deviation of 103 per store per week. These results are not to be confused with the impact of the promotions on each individual store, which can vary, as in some stores; promotions have for store specific reasons stronger or less strong impacts. Particularly for UABPs one can observe that the range of impact for UABPs is between 211 and 6,396 for individual stores, which makes the analysis for store specific criteria that influence the UABP, even more relevant.

e		
Mean	Median	SD
-862.8	-870.1	60.9
1,533.4	1,546.2	103.9
321.4	320.6	35.2
764.8	776.1	71.6
443.1	447.6	57.0
275.7	276.6	55.3
378.4	388.0	56.1
4,242.6	4,259.7	310.9
325.7	334.9	79.2
954.3	963.2	74.7
737.6	739.8	52.9
	Mean -862.8 1,533.4 321.4 764.8 443.1 275.7 378.4 4,242.6 325.7 954.3 737.6	Mean Median 862.8 870.1 1,533.4 1,546.2 321.4 320.6 764.8 776.1 443.1 447.6 275.7 276.6 378.4 388.0 4,242.6 4,259.7 325.7 334.9 954.3 963.2 737.6 739.8

Table 3.3 Impact of promotions on marginal sales-individual store values

Comparing this to any other type of promotion or advertising, the relative impact becomes evident as UABPs increase weekly store sales by a magnitude of $2.0-5.0 \times$ compared to other promotion types with comparable average promotion depth of the individual promotion types. Only the Glamour Shopping Week, which in fact has many commonalities with UABPs, as across the board promotions are given, coupled with significant advertising in a popular magazine and some in store additional promotions, has a bigger relative impact on weekly store sales. When further comparing the impact of UABPs and various types of advertising on weekly store sales, UABPs influence short-term sales about twice as much as the national TV campaign and 60 % more than the broadly distributed leaflets.

Reviewing the impact UABPs on an *across-store* basis (see Table 3.4), i.e. how the various types of promotions affect individual stores, one can observe that the impact of UABPs is positive for the entire store portfolio, while the range of its impact across stores varies significantly. The minimum marginal sales increase is 211.3, with the maximum being 6,395.7. The standard deviation of the sales impact is 533.1 on an average impact of 1,533.4. The relative large standard deviation and range across stores bears the question, which parameter influences the relative impact of those promotions, which is what will be discussed in the next paragraph.

3.4.2 Empirical Results: Store Level Antecedents

Table 3.5 shows parameter estimates of store-level effects in the hierarchical Bayes model. Overall only the location parameters shown have a strong impact on the success of UABPs, while some other parameters give an indication of the likely direction of impact while still having relatively high standard-deviations relative to their mean.

The strong impact of locations on sales during UABPs is an important finding as it would allow retailers to either only choose a sub-sample of stores in which UABPs will be offered to their customers or would even have some retailers

3.4 Empirical Results

Parameter	Mean	Median	SD	Minimum	Maximum
Intercept	-862.8	-823.6	312.4	-3,992.0	-222.5
UABP	1,533.4	1,493.5	533.1	211.3	6,395.7
WinterSales	321.4	310.2	138.9	-8.2	1,914.7
SummerSales	764.8	744.7	245.0	56.9	3,310.0
Buy2Get1Free	443.1	356.5	57.0	-363.8	3,817.9
Promotion days	275.7	253.8	55.3	-364.8	2,989.6
Dot days	378.4	402.0	116.8	-348.9	691.7
GlamourShoppingWeek	4,242.6	4,218.3	1,071.3	368.5	11,820.8
Mailing	325.7	309.5	238.6	-202.4	2,687.6
Leaflet	954.3	922.4	333.0	168.7	4,060.0
TV	737.6	697.3	316.3	-409.5	4,141.1

Table 3.4 Impact of promotions on marginal sales-across-store values

Table 3.5 Impact of store characteristics on sales impact of UABPs

Parameter for UABP	Mean	Median	SD	Minimum	Maximum
Store characteristics					
Size	0.48	0.50	0.61	-1.45	2.37
Age of store	7.32	7.35	6.33	-15.85	28.74
Years since renovation	-5.23	-5.43	6.01	-24.99	18.06
Location					
Shopping center	-186.6	-190.8	104.5	-599.0	144.5
Strip mall	311.2	315.4	91.3	16.0	608.1
Sublet	103.9	107.7	137.0	-384.0	685.8
Commercial zone	339.4	345.3	127.5	-84.3	734.0
Customer type					
Share of female	502.3	498.7	868.0	-3,048.7	3,489.7
Share of male	-157.9	-161.9	983.4	-4,041.1	3,300.9
Share of kids	-248.0	-253.5	862.4	-3,357.8	4,362.2
Value per article	364.9	363.5	255.6	-468.9	1,202.7

reconsider, whether UABPs would even be the best suited form of promotion given their store portfolio.

Store Characteristics Regarding store size and age since renovation relatively high standard deviations compared to their means indicate that their impact on store sales during UABPs is not very robust. The impact of the age of the store seems to be marginally more robust, indicating that the older, i.e. more established a store is, the stronger the impact on sales during UABPs.

Location The respective means and standard deviations suggest a strong impact of location on UABPs, showing that out-of city, lower-footfall locations [strip malls (μ =311.2; σ =91.3) and commercial zones (μ =339.4; σ =127.5)] have a strong and positive impact on the performance of UABPs relative to inner city locations (which is the reference variable), while shopping center locations have a negative relative impact on retail store sales during UABPs (μ =-186.6; σ =104.5). The

sales impact of sublet location is less obvious as shown by the relatively high standard deviation compared to its mean ($\mu = 103.9$; $\sigma = 137.0$), which indicates no strong impact of sublets on retail store sales during UABPs. According to the data, location does influence sales during UABPs, however when it comes to measuring the exact impact, further research would have to be conducted, taking into account the exact footfall which also varies within location types according to positioning within a shopping center, inner city etc. Furthermore it is to be noted that these impacts are on a relative basis, i.e. UABPs have a positive impact on all locations (as discussed before) but a more positive impact on certain locations.

Customer Type As Table 3.5 shows, the value per article, i.e. whether articles sold in a store are more costly, significantly impacts the influence UABPs have on sales ($\mu = 364.9$; $\sigma = 255.6$)—hence UABPs are more successful in stores with relatively pricier items. Furthermore Table 3.5 reports that *gender*, i.e. whether a store has for example more female than male customers, does not significantly influence the impact of the UABP on sales. This is shown by the large standard deviations that are larger than the respective means for each "gender category".

3.4.3 Comparison of Store Level Antecedents to Other Promotion Types

In addition to analyzing the store characteristics that impact the UABPs, particularly the several very robust results such as *location* and to a lesser extent the *age of the store* or the *average value of the articles purchased*, it is interesting to discuss, whether those were distinct findings for UABPs or generally true for various other types of promotions.

Location When comparing the impact of *location* among different types of promotions, Table 3.6 shows that the location impact is only evident for UABPs, showing an average of the absolute impact of 235.3 and a standard deviation of 115.1. The impact of *location* on all other types of promotions and advertising show significantly higher standard deviations compared to their mean, suggesting that location does not influence the relative success of promotion and advertising campaigns for every type of promotion to the same degree as it does for UABPs.

Value of Article Table 3.7 shows that the impact of the value per article for the different promotion and advertising types is mostly positive, suggesting that in general promotions work better if the articles purchased in a store are more expensive. However, relative to the respective standard deviations, this statement only suggests robust results for winter sales ($\mu = 234.4$; $\sigma = 126.1$) as well as TV ($\mu = 414.4$; $\sigma = 232.3$) and flyer advertising ($\mu = 322.1$; $\sigma = 188.7$). The impact of the value per article is slightly lower for UABPs ($\mu = 364.9$; $\sigma = 255.6$).

3.5 Summary and Discussion

Promotion type—location	Mean	SD	Mean/SD
Uniform Across the Board Promotions	235.3	115.1	2.04
GlamourShoppingWeek	350.4	402.9	0.87
WinterSales	30.3	100.0	0.30
SummerSales	93.9	130.4	0.72
Buy2Get1Free	117.2	171.6	0.68
Promotion days	191.4	190.1	1.01
Dot days	121.4	179.7	0.68
Leaflet	127.4	91.9	1.39
Mailing	116.5	224.0	0.52
TV	108.6	115.5	0.94

 Table 3.6 Impact of location on different promotion types

 Table 3.7 Impact of value per article on different promotion types

Promotion type—value of article	Mean	SD	Mean/SD
Uniform Across the Board Promotions	364.9	255.9	1.43
GlamourShoppingWeek	624.5	848.5	0.74
WinterSales	234.4	126.1	1.86
SummerSales	98.9	238.7	0.41
Buy2Get1Free	334.8	292.0	1.15
Promotion days	214.2	396.2	0.54
Dot days	-271.4	302.8	0.90
Leaflet	322.1	188.7	1.71
Mailing	245.3	413.8	0.59
TV	414.4	232.3	1.79

Age of the Store The age of the store, which is used as a proxy for how established a retail location is, has only moderate positive impact on UABPs compared to e.g. *location or value of the article* as Table 3.5 shows. This effect is similarly small for other promotion types and one can conclude that the moderately positive effect of the store age is no distinct characteristics of UABPs as outlined in Table 3.8.

3.5 Summary and Discussion

In this chapter, a new form of promotion, Uniform Across the Board Promotions (UABP), is introduced and their impact on short-term retail sales and the impact store characteristics have on the success of UABPs are investigated.

Promotion type—age of store	Mean	SD	Mean/SD
Uniform Across the Board Promotions	7.32	6.33	1.16
GlamourShoppingWeek	-6.71	20.26	0.33
WinterSales	1.12	3.82	0.29
SummerSales	9.07	6.21	1.46
Buy2Get1Free	-11.00	8.86	1.24
Promotion Days	0.72	11.66	0.06
Dot Days	0.08	8.65	0.01
Leaflet	5.15	4.46	1.15
Mailing	9.57	11.21	0.85
TV	-2.61	6.12	0.43

 Table 3.8 Impact of age of store on different promotion types

3.5.1 Academic Implications from Short-Term Effects UABPs

This chapter discusses conceptually, that UABPs appear to have advantages for retailers and consumers alike. This has been supported by the results of the analysis as UABPs show a stronger positive short-term impact on retail sales than any other promotion type and advertising type analyzed. The stronger increase in overall retail sales can partly be attributed to UABPs not causing any within-store substitution effects, the way other promotions do (Gupta 1988; Sun et al. 2003). Furthermore the increase in flexibility, which reduces the information and transaction costs for the customer (Metha et al. 2003; Stigler 1961), while not causing any reactance, makes the promotion more attractive for customers and can hence cause promotion induced store switching, which positively affects the higher short-term sales impact of UABPs recorded in these findings.

In addition to the impact promotions and specifically UABPs have on sales, it is also analyzed which store characteristics influence the success of UABPs. It is being tested, whether those criteria are unique for UABPs or shared with other promotion types. The analysis shows that location has a stronger impact on the performance of UABPs than other types of store characteristics. This is being shown by the fact, that, e.g., inner city locations and shopping centers perform much worse on a relative basis, than out-of-city locations such as commercial zones and strip malls. A possible explanation lies within the theory of information economics, as the guaranteed discount from the UABPs on the entire assortment is an incentive for the consumer to incur the higher search and transaction associated with the longer commute to an out of city location, which causes more positive relative sales increase. The data shows that this is a unique feature of UABPs, as all promotions seem to directionally perform relatively better in commercial zones and strip malls than in inner city and shopping center locations. One could hence figure, that location only affects the impact of UABPs on relative retail store sales, while the relative performance of other promotion and advertising types are not strongly affected by location in the example. Regarding the store characteristics the data shows that the average value per article influences the success of UABPs. This impact is positive, which is intuitive as the higher the potential savings, which are always relative to the value per article, the higher the incentive to visit a store during UABPs. The theoretical explanation lies again within the information economics which compare the savings with induced search and transaction costs.

This chapter has closed a research gap, introducing UABPs, their impact on overall retail sales as well as the influence certain store characteristics have on the promotional success. Due to the large and extensive dataset it has been possible to compare these results to the impact other forms of promotions have on overall retail sales and the role store characteristics play for those promotions, while previous research has mostly focused on the impact promotions have on certain brands or categories, rather than overall retail sales. Regardless of these contributions, moving forward, one needs to work on several topics to enhance this research:

First, this chapter has dealt with promotions of a value fashion retailer, whereas researchers should be encouraged to expand this to other categories and sales channels as promotions in the fashion business are very common and customers are used to reacting to them and potentially adjusting their shopping behavior accordingly. Also fashion articles are a durable good and spending on fashion is mostly discretionary—both characteristics are likely to influence the results. Furthermore the value of an individual article of the retailer is on average below 10 euros, and hence on the lower end of the general fashion market. As shown above, the actual value per article influences the customer reaction on UABPs (while the ranges tested for those articles are still within the 10 euro range) and testing reactions to UABPs for significantly more expensive goods will be an interesting area for future research.

Second: The impact of UABPs on sales for store characteristics and a few product-store characteristics (such as price) has been tested in this chapter. While the articles the UABPs have been applied on are basically non-branded goods of a vertically integrated retailer, it will be interesting to conduct similar research when testing the influence of more product specific characteristics, such as brand image, on the success of UAPBs, as conducted in previous research for other forms of promotions (e.g. Nijs et al. 2001; Macé and Neslin 2004; Montgomery 1997).

Third: This chapter only considers the impact on sales, whereas retailers are also interested in the impact promotions have on gross profit. While the exact gross profit, which given the operational leverage in many retail businesses is a good short-term proxy for EBITDA (Earnings before Interested Depreciation and Amortization) is influenced by the price elasticity and the gross margins of the actual goods purchased during UABPs, a simple calculation shows (Table 3.9), that a 20 % across the board discount on a net sales price level, when assuming a typical non-promotional gross margin of 50 % (as discussed with management), would require the retailer to sell approx. 67 % more, on a volume basis, to break even on a gross profit level. This obviously is a strong simplification as articles have different gross margins and promotional expenses. To estimate the gross profit impact of UABPs future research should look at overall retail sales in combination with product specific gross margins and price elasticities.

	Non-promotion	(Break-even) UABP	% vs. non-promotion
Price per article	100.0	80.0	-20.0 %
Volume	1.0	1.67	67.0 %
COGS per article	50.0	50.0	0.0~%
Net sales	100.0	133.3	33.0 %
COGS	-50.0	-83.3	67.0 %
Gross profit	50.0	50.0	0.0%
GP margin	50.0 %	37.5 %	-12.5 %

Table 3.9 Illustrative calculation of required volume increase to break-even during UABPs

Fourth: Regardless of whether one discusses the impact on sales or profitability, future research would have to analyze the long-term impact UABPs have on the retailer's image and customer behavior. This impact could be positive or negative, in a sense that positively, the increased short-term traffic in retail stores, might help to promote the store in general and hence also serves as a sort of advertising, increasing long-term traffic to the store. Negatively, consumers could be conditioned to only visit the retailer's stores given the provided flexibility through UABPs, which would result in an overall decline of gross margins and a decrease in sales in non-promotion periods. Only after these additional topics have been further analyzed will one be able to pass a judgment on whether Uniform Across the Board Promotions are a sustainable and favorable alternative to traditional price promotions.

3.5.2 Retailer Implications from Uniform Across the Board Promotions

This research shows that UABPs have a strong impact on short-term sales and are hence an effective promotional tool for retailers. The advantage of not causing substitution effects within a store goes with the disadvantage that the retailer is not able to use UABPs to steer the customer towards certain brands or assortments, i.e., use UABPs as a tool to sell-off or promote specific items, as used in end-of season sales. UABPs should hence be used to cause an overall sales increase or general de-stockage and potentially drive new customers to the stores that are attracted by this form of promotion. As UABPs aim at promoting the entire store, store trading characteristics influence the success of UABPs, which is why they need to be considered when taking decisions on whether and in which scope UABPs are to be launched. Especially when retailers chose to apply a micro-marketing strategy (Montgomery 1997) and only launch UABPs in a subsample of their stores, a consideration of store characteristics is important to find the most efficient subsample of stores. As mentioned above, going forward it is important for managers to gain further understanding on the sales and profit impact of UABPs for additional sales channels, product categories and analyze their long-term impact on themselves, non-promotional sales and overall customer behavior.

References

- Abraham MM, Lodish L (1993) An implemented system for improving promotion productivity using store scanner data. Market Sci 12(3):248–269
- Ailawadi KL, Harlam BA, Cesar J, Trounce D (2006) Promotion profitability for a retailer: the role of promotion, brand, category, and store characteristics. J Market Res 43(4):518–535
- Andrews RL, Currim IS, Leeflang P, Lim J (2008) Estimating the SCAN*PRO model of store sales: HB, FM or just OLS? Int J Res Market 25(1):22–33
- Bell DR, Lattin JM (1998) Shopping behavior and consumer preference for store price format: why "large basket" shoppers prefer EDLP. Market Sci 17(1):66–68
- Bell DR, Chiang J, Padmanabhan V (1999) The decomposition of promotional response: an empirical generalization. Market Sci 18(4):504–526
- Blattberg RC, George EI (1991) Shrinkage estimation of price and promotional elasticities: seemingly unrelated equations. J Am Stat Assoc 86(414):304–315
- Boatwright P, McCulloch R, Rossi P (1999) Account-level modeling for trade promotion: an application of a constrained parameter hierarchical model. J Am Stat Assoc 94 (448):1063–1073
- Bomsdorf C, Kerbusk KP, Scheele M (2003) Erst Gas geben, dann bremsen, Spiegel 49/2003. http://www.spiegel.de/spiegel/print/d-29341609.html. Last accessed 30 Nov 2012
- Carlin PC, Louis TA (2008) Bayesian methods for data analysis, 3rd edn. Chapman & Hall, London
- Casella G (1985) An introduction to empirical Bayes data analysis. Am Stat 39(2):83-87
- Gupta S (1988) Impact of sales promotions on when, what and how much to buy. J Market Res 25 (4):342–355
- Handelsblatt/Reuters (2012) Praktiker setzt alles auf eine Karte. http://www.handelsblatt.com/ finanzen/aktien/aktien-im-fokus/die-rettung-praktiker-setzt-alles-auf-eine-karte/7079140. html. Last accessed 30 Nov 2012
- Hoch SJ, Drèze X, Purk ME (1994) EDLP, Hi-Lo, and margin arithmetic. J Market 58(4):16-27
- Hoch SJ, Kim BD, Montgomery AL, Rossi PE (1995) Determinants of store-level price elasticity. J Market Res 32(1):17–29
- Kruschke JK (2011) Doing Bayesian data analysis, 1st edn. Elsevier, Oxford
- Kumar V, Leone RP (1988) Measuring the effect of retail store promotions on brand and store substitution. J Market Res 25(2):178–185
- Lim J, Currim IS, Andrews RL (2005) Consumer heterogeneity in the longer-term effects of price promotions. Int J Res Market 22(4):441–457
- Macé S, Neslin SA (2004) The determinants of pre- and postpromotion dips in sales of frequently purchased goods. J Market Res 41(3):339–350
- Metha N, Rajiv S, Srinivasan K (2003) Price uncertainty and consumer search: a structural model of consideration set formation. Market Sci 22(1):58–84
- Montgomery AL (1997) Creating micro-marketing pricing strategies using supermarket scanner data. Market Sci 16(4):315–337
- Montgomery AL, Rossi PE (1999) Estimating price elasticities with theory-based priors. J Market Res 36(4):412–423
- Nijs VR, Dekimpe MG, Steenkamp JB, Hanssens D (2001) The category-demand effects of price promotions. Market Sci 20(1):1–22
- Rossi PE (2012) Bayesian Inference for Marketing/Micro-Econometrics. R package version 2.2
- Rossi PE, Allenby GM, McCulloch R (2005) Bayesian statistics and marketing, 1st edn. Wiley, West Sussex
- Stigler GJ (1961) The economics of information. J Polit Econ 69(3):213-225
- Sun B (2005) Promotion effects on endogenous consumption. Market Sci 24(3):430–443

- Sun B, Neslin SA, Srinivasan KA (2003) Measuring the impact of promotions on brand switching under rational consumer behavior. J Market Res 50(4):389–405
- Van Heerde HJ, Leeflang PSH, Wittink DR (2000) The estimation of pre-and post-promotion dips with store-level scanner data. J Market Res 37(3):283–395
- Van Heerde HJ, Gupta S, Wittink DR (2003) Is 75% of the sales promotion bump due to brand switching? No, only 33% is. J Market Res 40(4):481–491

Chapter 4 Adjustment and Permanent Effects of Uniform Across the Board Promotions

4.1 Introduction to Adjustment and Permanent Effects of UABPs

To introduce the adjustment and permanent effects of UABPs and put them into context, it makes sense to review and extend on the Praktiker (German DIY retailer) example referred to in Sects. 1.1 and 3.1. While Chap. 3 mainly describes how Praktiker has been a pioneer in introducing UABPs, how they have gained importance in the retail environment and how strong their short-term impact on retail store sales can be, this chapter fast forwards to the situation, where UABPs have been in use for a longer period of time.

Back to Praktiker—Following 3 years of negative like-for-like sales, an overall drop of revenues from €4 bn to €3 bn, Praktiker, the second largest German DIY retailer with more than 200 stores faced losses in excess of €500 m for the financial year 2011, forcing Kay Hafner, the CEO of Praktiker, to reach out to his shareholders and plead for the acceptance of a major restructuring and refinancing package which was the last and only alternative to an immediate insolvency of Praktiker (Böcking 2012). Current and former management, retail experts, investors and the media blame the excessive use of Uniform Across the Board Promotions (UABPs), to have caused or at least significantly contributed to this downfall. One of the core investors of Praktiker, Thomas Fox, called the UABPs "sweet drugs", others stated that "Praktiker can now not do with nor without them" (Hielscher 2010; Läsker 2012).

After introducing UABPs to the German market in 2003, Praktiker only ran few days of UABP campaigns a year. Encouraged by their short-term success, management increased the UABP promotion days to 107 in 2007 until they feared that customers adapted their shopping behavior to the once successful UABPs, which in sum negatively affects the overall gross margin of Praktiker. In order to reverse this trend, Praktiker decreased the number of UABP days down to 62 in 2010 and eventually down to 0 in 2011 (Böcking 2012; Hielscher 2010). This contributed to

the described decrease in sales and the eventual collapse of the share price from 34.50 Euro in 2007 to 1.24 Euro in 2012 (Hirzel 2012).

While the German business press and management alike sees the root cause for Praktiker's financial difficulties in the excessive use of UABPs, this hypothesis has not been validated by marketing research. Even though Sect. 1.1 and Chap. 3 introduces this new form of promotion to the marketing literature, and among other things analyses the short-term impact of UABPs on retail store sales, the adjustment effects or permanent effects on customer behavior has not yet been covered in the literature. Against this background, it is the goal of this chapter to analyze adjustment and permanent effects of UABPs on retail store sales.

The key dependent variable is store level performance: before, during and after UABPs. The analysis of the sales bump during UABPs and the respective pre- and post-promotion dips show how customers adapt their shopping behavior to profit from those promotions. Investigating, how those dips and bumps evolve over time, helps to understand the long-term impact of UABPs on potential acceleration and stockpiling effects, promotion sensitivity and their subsequent impact on long-term retail store performance.

To analyze the long-term impact of UABPs on pre-and post-promotion dips as well as baseline sales, i.e. customer behavior, store level data from a German value fashion retailer is being used. The dataset contains daily sales level data from 421 stores (S = 421) over a time span of 1,107 days (T = 1,107) in the period from 2008 to 2012. Methodologically a mixed-effect times series model is being applied on the panel dataset to reflect fixed promotion effects, random store intercepts, and time based interaction variables, with the adjustment effects of UABPs models as lead and lag effects (Macé and Neslin 2004; Judge et al. 1985; Van Heerde et al. 2000). The main contribution to research is twofold: First the existence and degree of pre-and post-promotion dips for UABPs is being shown and second it is analyzed how those dips evolve over time and whether the long-term baseline sales of the retailer is being affected by the UABPs.

4.2 Conceptual Framework

Before developing the hypotheses for discussing the immediate and dynamic adjustment effects (pre-and post-promotion dips) as well as long-term effects of UABPs some key concepts the following paragraph (1) recalls some of the main characteristics of UABPs and discusses, what (2) "adjustment effects" and (3) long-term effects are defined as in this context.
4.2.1 Definitions of Relevant Concepts When Discussing Adjustment and Permanent Effects

As previously discussed, UABPs are a form of temporary price promotion that is characterized by its flexibility for the customer which is caused by its promotional breadth, as the same relative discount is being applied to any article in the store. In Chap. 3 it is already shown, that Uniform Across the Board Promotions have a substantial immediate impact on overall retail store sales, significantly higher than the short-term sales increase caused by other traditional promotions and advertising analyzed. To judge whether these particularities have broader effects on the net performance of UABPs in the long-run, this chapter focuses on the analysis of the *adjustment effects* of UABPs as well as the *permanent*, *i.e. long term effects*.

4.2.1.1 Definition of Adjustment Effects

Under adjustment effects, the existence of pre and- post-deal troughs around potential promotional bumps is being summarized. Post-deal troughs are often attributed to timing and purchase (quantity) acceleration effects, as customers try to profit from a promotion, by building up stock or anticipating purchases and hence reducing their purchases in subsequent periods (Bell et al. 1999; Blattberg et al. 1995; Grover and Srinivasan 1992). Pre-deal troughs are anticipatory effects of promotions which are caused by consumers anticipating marketing changes, e.g. lower prices in the future, and adjust their behavior by deferring purchases accordingly (Kalwani et al. 1990; Winer 1986; Van Heerde et al. 2000). The magnitude of those dips can for instance be influenced by explicit advertising for a promotion as well as by learning (Doyle and Saunders 1985). Adjustment effects can in general lead to either the reversion to previous sales levels or a new sales level, while the adjustment effects themselves can be positive or negative (Blattberg and Neslin 1990; Greenleaf 1995).

4.2.1.2 Definition of Permanent/Long-Term Effects of UABPs

While the definition of long-term promotion effects differs widely in the marketing literature, the focus is on the existence of three potential effects that could influence the performance of UABPs at any time. Those effects could show as either (1) reversion to a different sales level after the promotion and adjustment period are over (2) less positive immediate sales effects of the promotion over time potentially caused by a decrease in promotion sensitivity or (3) stronger adjustment effects over time, leading to an overall neutral or even negative sales impact of UABPs. Figure 4.1 illustrates the focus of this research which will be further outlined in four key hypotheses.



Fig. 4.1 Framework of impact of Uniform Across the Board Promotions on store sales

4.2.2 Hypothesis Regarding the Adjustment and Permanent Effects of UABPs

Adjustment effects of UABPs Empirical evidence on whether adjustment effects exist is mixed. Price perception theory suggests that price promotions impact the reference prices of consumers to which they compare prices during non-promotion times and which hence increases the reluctance to buy during non-promotional periods (Helson 1964; Lattin and Bucklin 1989). Moreover, Blattberg et al. (1995) argue that promotions damage the brand's quality image, which could in turn affect future sales and cause post-promotion troughs (Blattberg et al. 1995). Nevertheless, post-promotion troughs are rarely detected in store level sales data, which has been explained by behavioral theories, as for instance new consumers, attracted by lower prices, will try the store or product during promotions and might repurchase in the future (Mela et al. 1997) or else promotions can remind existing consumers to return to a brand or store and reinforce their attraction to it (Erdem 1996). Despite those arguments, one could expect that in line with the short-term results, the flexibility of UABPs increase the incentive to shop during UABPs over non-promotional periods and that pre- and post-promotional dips will be visible in store-level data. Moreover UABPs focus on the entire store and not just on individual brands, which offsets Neslin and Schneider Stone's (1996) argument that competitive promotions within a store might mask the post-promotional dips of promotions on a store-level (Neslin and Schneider Stone 1996).

 H_1 : Uniform Across the Board Promotions cause adjustment effects, i.e. pre- and postpromotion sales dips, which reduce the net sales effect of UABPs

Permanent/long-term sales impact of UABPs When discussing the potential long-term effects of UABPs, the focus lies on the analysis of three areas as mentioned above: (1) impact on baseline sales, (2) impact on immediate sales effects over time and (3) adjustment effects over time.

First (impact of UABPs on baseline sales) Even though, as stated earlier, repeated promotions increase consumer price sensitivity and decrease the reference price while also decreasing brand equity over time, results on the long-term impact of promotions are mixed (Blattberg and Neslin 1990). Kalwani and Yim (1992) find that there seems to be a tipping point, as consumers that have often purchased goods during promotions are likely to adapt their behavior to shift all purchases to promotion times. However, a number of studies exist, arguing that there are no negative long-term effects on the baseline sales, as sales will eventually return to their pre-promotion level (Assuncao and Meyer 1993; Krishna 1992, 1994). Even if there is a non-mean reverting effect of promotions on sales, this long-term effect is generally small and applicable only to a few categories (Dekimpe et al. 1999; Lim et al. 2005; Nijs et al. 2001).

Against this background, it is unclear, whether UABPs will generally have negative long-term sales effects, i.e. a mean-reversion to a sales level below the pre-promotional level. However given the product characteristics of this data sample, which focuses on durable goods, that are easy to stockpile (as opposed to consumables, for which out of stock costs are sometimes higher and that are more difficult to stockpile), one could expect to see negative long-term effects on baseline sales—in a later part of the chapter it will be discussed whether this can be made a general finding.

H₂: Uniform Across the Board Promotions have a negative long-term effect on baseline sales

Second (impact on immediate sales effects of UABPs over time) In addition and related to the long-term impact on non-promotional periods, this chapter discusses how consumers react to subsequent promotions or specifically UABPs and how this affects the adjustment effects. Due to the decreased reference price and increased price sensitivity, the probability to react to future promotions decreases (Mela et al. 1997). Moreover households are less likely to act on a promotion, if they are conditioned to "lie and wait" for even more attractive promotions in the future (Jacobson and Obermiller 1990; Mela et al. 1998). Gönül and Srinivasan (1996) find that consumers form expectation about the availability of promotions (coupons) in the future and that depending on those expectations they defer purchases to subsequent periods.

H3: The positive sales impact of Uniform Across the Board Promotions decreases over time

Third (Impact on adjustment effects over time) An increase in adjustment effects over time could lead to a neutral or even negative effect of UABPs on sales over time. While Pauwels et al. (2002), do not find that the negative pre-and post-promotion dips outweigh the positive immediate sales effect, they do state, that

these findings strongly depend on the product category and its storability. This argument also supports earlier research that reported mixed results on the net impact of promotions. While some researchers found positive effects (e.g. Davis et al. 1992; Guadagni and Little 1983), later studies found predominantly negative net effects on sales (Jedidi et al. 1999; Mela et al. 1998). As mentioned before the flexibility of UABPs for consumers, combined with the low out of stock costs of the category fashion (or most durable goods) are an indication that consumers learn how to adjust their shopping behavior, which would lead to larger adjustment effects. However and in line with the existing research, there are strong indications that any of the measured net effects, or increasing adjustment effects might be strongly affected by the actual category which is why it is again being addressed in the discussion section of this chapter.

 H_4 : Adjustment effects from UABPs increase over time (potentially leading to neutral or negative net effects of UABPs)

4.3 Method

4.3.1 Store-Level Data to Analyze Adjustment and Permanent Effects

The analysis is being performed based on daily store data for 421 retail stores of a German value fashion retailer. Out of the original sample, stores with incomplete or flawed sales history (e.g. a week missing and then 1 day with ten times the regular sales) had been eliminated. The time span for the data is 1,107 days in the period 2008–2012. The daily store sales data has been de-seasonalized and de-trended for a linear temporal trend using a regression model based on to the one by Abraham and Lodish (1993) with additional adjustments for the weekday and store level sales data. For the promotions dummy variables are introduced that indicate whether a UABP has been run in the retailer's store during a certain day according to the central marketing plan of the company—lead and lag variables as well as time interaction effects have been added accordingly.

4.3.2 Methodology to Analyze Adjustment and Permanent UABP Effects

Even though the existence of pre- and post-promotion dips is evident in household level data, it is often not observed when analyzing store data. Various arguments imply the difficulty of detecting such dips and refer to the complex patterns in brand and store sales (Neslin and Schneider Stone 1996; Van Heerde et al. 2000). To

overcome these difficulties, Van Heerde et al. (2000) have introduced the use of a flexible economic model, which is based on the SCAN*PRO model (Christen et al. 1997) to regress brand-level sales on immediate, lagged and lead price promotion effects.

This model (called an "*unrestricted dynamic effect model*" or also dynamic effects regression model) is being adapted to suit the purposes as further described below. The original model already provides very accurate results and its advantages further lie in the high degree of flexibility, which will also allow us to include parameters to reflect potential long-term effects of UABPs. Furthermore it is relatively straight forward to implement from a managerial perspective (Macé and Neslin 2004; Van Heerde et al. 2000).

4.3.2.1 Dynamic Effects Regression Model

To analyze pre- and post-promotion dips for Uniform Across the Board Promotions as well as long-term effects, the econometric model used by Van Heerde et al. (2000) is being adapted. The model used in this chapter mainly differs in three aspects from the original model. First, instead of applying the model on weekly brand sales, the results are being estimated using OLS estimation on overall daily store sales, calculating an absolute sales impact of the promotion rather than promotion brand price elasticities. Second contrary to Van Heerde et al. (2000), who were pooling across stores, the model is being estimated at a store-specific level as done by Macé and Neslin (2004), by introducing a store-specific intercept. Third, a time variable is being included i.e. the impact of the cumulative number of UABPs used on the promotional bumps, lead and lag effects as well as non-promotional sales is being considered.

The original model, which serves as basis for the model used in this chapter looks like the following (compare Van Heerde et al. (2000) and Macé and Neslin (2004))

$$S_{t} = \alpha x_{t} + \sum_{u=1}^{s} \beta_{u} x_{t-u} + \sum_{\nu=1}^{s'} \gamma_{\nu} x_{t+\nu} + u_{t}$$

Where S_t are sales in period *t*. Lagged and lead effects are included with the relevant lag (t - 1, t - 2, t - 3) and lead (t + 1, t + 2, t + 3) predictors on the UABP which is included as the dummy variable x_t . u_t is a store specific intercept with $u_t = \alpha_0 + \theta_n$ with $\theta_n = (0; \sigma)$ where *n* is an indicator for the specific store. In addition to this basic model, the model is being enhanced as mentioned before to fit the required purposes, which is shown in the model below. For store *n* with $n = 1, \dots, 421$, the full model is:

$$S_{n,t} = \alpha_{n,t} x_t + \sum_{u=1}^{3} \beta_{n,t-u}^{UABP} x_{t-u} + \sum_{\nu=1}^{3} \gamma_{n,t+\nu}^{UABP} x_{t+\nu} + X_t * \left(\alpha'_{n,t} x_t + \sum_{u=1}^{3} \beta'_{n,t-u}^{UABP} x_{t-u} + \sum_{\nu=1}^{3} \gamma'_{n,t+\nu}^{UABP} x_{t+\nu} \right) + \delta_{n,t} X_t + u_n$$

 $S_{n,t}$ is daily sales of store n for period t

Indicators and intercept

 x_t is the immediate sales effect indicator for UABPs in period t. ($x_t = 1$ if in period t an UABP is taking place and 0 otherwise).

 x_{t+v} is the lead effects indicator for UABPs in period t - v. ($x_{t+v} = 1$ if an observation is a lead in period t + v and 0 otherwise).

 x_{t-u} is the lag effects indicator for UABPs in period t-u. ($x_{t-u}=1$ if an observation is a lag in period t-u and 0 otherwise).

 X_t is the mean-centered overall number of UABPs performed in a specific store until period t (on total 23 UABPs considered, hence average is 11.5; this number is deducted from the respective UABP, i.e. for the twentieth UABP, X equals 8.5 (20 - 11.5)).

 u_n is a store specific intercept, which reflects the impact of all store characteristics on sales.

Parameters

 β_{t-u}^{UABP} , is the parameter for adjustment effect (lag) of UABP on sales α_t^{UABP} , is the parameter for the immediate promotion impact of UABPs on sales γ_{t+v}^{UABP} is the parameter for adjustment effect (lead) of UABP on sales β_{t-u}^{i} , are the parameters for the long-term effects of UABPs on its lag parameters α'_{t+v}^{UABP} , are the parameters for the long-term effects of UABPs on their immediate promotional increase

 γ'_t^{UABP} , is the parameter for the long-term effects of UABPs on its lead parameters $\delta_{n,t}$, is the parameter for the long-term effect of UABPs (modeled as mean-centered cumulative UABPs) on sales

The model has been estimated using the statistical software R and specifically the *nlme package* (Pinheiro et al. 2013), from which the use of the linear-mixed effects function (*lme*) has been applied to conduct an OLS regression considering fixed effects as well as the random-store effects (Macé and Neslin 2004). To perform the regression, the store data has been shaped as panel data (long-format) and lead and lag variables have been included, using mainly functions from the *plyr package* in R (Wickham 2009).

4.3.2.2 Model Variables

This section explains the specifications and operationalization of variables used in the regression model. To analyze the impact of UABPs on adjustment and permanent effects one distinguishes between five main types of indicators and parameters: (1) Impact of UABPs on sales in the respective period, (2) the lead and lag structures as described above (*line 1 of full regression model*), (3) the interaction between the cumulative UABPs run and the lead and lag indicators (*line 2 of the full regression model*) and (4) the direct impact of the cumulative UABPs on sales (*line 3 of full regression model*). In addition (5) store-specific characteristics are being considered (*line 4 of full regression model*).

In the model the criterion variable is the daily sales of a particular store. As described above, daily sales have been de-trended for a linear time trend as well as for seasonal and week-day fluctuations that are typical in this fashion retail setting. As predictors five different type of variables described in the following have been included:

- 1. *Uniform Across the Board Promotions*: the impact of Uniform Across the Board Promotions on daily sales has been modeled using dummy variables on a daily basis for whenever a UABP has been taken place in the observation period.
- 2. Adjustment effects/lead and lag effects: as outlined above, these effects show the impact UABPs have on the sales in the periods exactly before or after this type of promotion has been run. After discussing with management the lead and lag effects are chosen to be 3 days (s = 3; s' = 3) prior or post the use of an UABP. These lead-and lag effects are modeled using dummy variables, as suggested by Van Heerde et al. (2000). While when testing for model alternatives these lag and lead periods could be extended, the reason 3 days are chosen is that first the pre- and post-promotion periods do not interact with any other form of promotion or advertising and second the leakage/announcement of an upcoming UABPs does typically not exceed 3 days prior to the promotion.
- 3. Long-term effects of UABPs on promotion and adjustment effects. The flexibility of the model allows to include the variable of (mean-centered) cumulative number of UABPs run by the fashion retailer which interacts with the dummy variables described under (1) and (2) namely the (1) the immediate effect UABPs have on sales and (2) the adjustment effects of each UABP. This allows measuring how the sales bump and dips are impacted by the term "cumulative UABPs over time" which is used as a proxy for the long-term promotional impact of UABPs.
- 4. Long-term impact on non-promotional store sales. The effect of running UABPs on the daily store sales can be measured by analyzing the direct impact the (mean-centered) cumulative numbers of UABPs have on average on sales. This is modeled as a linear impact—with the respective implications to be discussed in a later part of the chapter.

5. *Store specific intercept*. As promotions are being carried out on a store level, a store specific intercept is being introduced to reflect e.g. different store sizes and locations.

4.4 Empirical Results

Immediate and adjustment effects of UABPs The results in Table 4.1 confirm the significant positive impact (5,448.5) UABPs have on daily store sales. These immediate effects and particularly their adjustment effects will be discussed in the following.

Hypothesis 1: Pre-and post-promotion dips for UABPs Figure 4.3 basically zooms into the sales pattern and focuses on the adjustment effects and further portrays the average lead and lag effects around the sales increase. These de-trended and de-seasonalized effects on daily store sales are a visualization of the regression results reported in Table 4.1. Figure 4.3 shows that the UABP induced effect on sales in the period leading up to the UABP (3 trading days) is negative, indicating a pre-promotion dip. In the period following the UABP (again 3 trading days), the figure shows the same pattern, while the effects in the outer *leads* (Lead 2 and Lead 3) are relatively small (Fig. 4.2).

Table 4.1 summarizes these findings, showing that all lag effects are significant and negative (Lag 3 to Lag 1: -382.3, -313.4 and -1,091.2), while all lead effects are also negative (Lead 3 to Lead 1: -1,013.4, -58.5 and -11.9) but only lead 1 and lead 2 are statistically significant. Overall hypothesis 1 can be supported as there are significant adjustment effects for UABPs with most variables being significantly below 0. These effects are in line with results from Macé and Neslin (2004) that have applied a similar approach for various categories during traditional price promotions (Macé and Neslin 2004). Despite significant adjustment effects, the net effect of the UABP is still positive, as the increase in sales during the UABP outweighs the adjustment effects.

Long-term impact of UABPs When assessing the long-term impact of UABPs on promotions three distinct areas are covered: First it is discussed how the increased use of UABPs affects the expected baseline store sales of the retailer. Second the focus is on how the effectiveness of UABPs changes over time, i.e. whether the impact of later UABPs on sales is different compared to earlier UABPs—a concept closely related to research on promotion sensitivity. Third it is investigated how adjustment effects develop over time, i.e. are the pre-and post-promotion troughs getting larger or smaller once consumers get more used to UABPs. Table 4.2 shows the long-term effects of UABPs on the three discussed concepts above.

Hypothesis 2: UABPs have a negative long-term effect on baseline sales The effect of running UABPs does have a significantly negative effect on sales in periods where no UABPs are run (-3.59). While this effect is statistically significant (H₂ can be supported), the effect is relatively small on a daily basis compared to the

Variables	Term	Result
Intercept		-153.63***
		(1.74)
UABP-lag $(t - 3)$	$\beta_{n+2}^{UABP} x_{t-3}$	-383.85***
	<i>i n</i> , <i>i</i> = 5 · · · · 5	(9.47)
UABP-lag $(t-2)$	$\beta_{n,t}^{UABP} x_{t-2}$	-314.1***
	, <i>n</i> , <i>t</i> = 2	(11.00)
UABP-lag $(t-1)$	$\beta_{n,t-1}^{UABP} x_{t-1}$	-1,091.72***
-	$r_{i,i} = 1$	(10.89)
UABP (t)	$\alpha_{n,t} x_t$	5,448.62***
		(10.90)
UABP-lead $(t+1)$	$\gamma_{n,t+1}^{UABP} \chi_{t+1}$	-1,014.29***
	$n_{i}t + 1$	(10.88)
UABP-lead $(t+2)$	$\gamma_{n,t+2}^{UABP} \chi_{t+2}$	-59.60***
	$n_{i}t + 2$	(10.86)
UABP-lead $(t+3)$	$\gamma_{n,t+3}^{UABP} \chi_{t+3}$	-12.90
	$n, n \neq 5$	(9.17)

 Table 4.1
 Adjustment effects for UABPs (pre- and post-promotion dips)

N = 461,246 observations from 421 stores ****p < 0.000



Fig. 4.2 Pre-and post-promotion dips for Uniform Across the Board Promotions

overall sales effect of UABPs, however adds up over time given the relatively longer non-UABP periods. The interpretation of this number is, that the baseline, to which daily store sales revert is slightly lower (by 3.58 Euros sales per day) after running one additional Uniform Across the Board Promotions (see also Table 4.2). Significant but small negative effects (that are however category specific) from promotions on baseline sales are in line with findings in the recent marketing literature (Lim et al. 2005; Nijs et al. 2001).

Variables	Term	Result
Long-term impact on UABP-lag $(t - 3)$	$X_t \beta' {}^{UABP}_{n t-3} x_{t-3}$	-10.30***
	1836 5 1	(0.50)
Long-term impact on UABP-lag $(t-2)$	$X_t \beta' {}^{UABP}_{n,t-2} x_{t-2}$	3.20***
	1836 Z	(0.61)
Long-term impact on UABP-lag $(t-1)$	$X_t \beta' {}^{UABP}_{n,t-1} x_{t-1}$	49.75***
	$n_{jl} = 1$	(0.62)
Long-term impact on UABP (t)	$X_t \beta'_{n,t}^{UABP} x_t$	-82.04***
		(0.62)
Long-term impact on UABP-lead (t+1)	$X_t \gamma' UABP_{n,t+1} X_{t+1}$	47.94***
	n, i+1	(0.62)
Long-term impact on UABP-lead (t+2)	$X_t \gamma' UABP_{n,t+2} X_{t+2}$	-18.46^{***}
	n, n + 2	(0.63)
Long-term impact on UABP-lead (t+3)	$X_t \gamma' UABP_{x_t+3}$	7.49***
	<i>n</i> , <i>t</i> +5 + 5	(0.51)
Long-term impact on non-UABP sales	$\delta_{n,t}X_t$	-3.58***
		(0.11)

Table 4.2 Impact of running more UABPs on immediate and adjustment effects

N = 461,246 observations from 421 stores ***p < 0.000



Fig. 4.3 Visualization of long-term evolvement of immediate and adjustment effects

Hypothesis 3: The positive sales impact of UABPs decrease over time The longterm impact of UABPs on sales decreases significantly with additional UABPs performed over time (-82.04), which means that this type of promotion continuously wears off. This has to be put into perspective with the actual sales bump caused by UABPs—strongly simplifying and assuming a linear decline over the cumulative number of UABPs suggests that after around the 66th UABP run (5,448.63 immediate impact divided by 82.04 wear-off effect ~66 UABPs), the positive impact of the UABP on sales would have vanished. This has been visualized in Fig. 4.3, in which simply the main effects from the promotional impact had been considered and the impact of running additional UABPs multiplied by the respective number of UABPs has been added/subtracted from that number (*e.g. immediate sales impact of 10th UABP* = $5,448.63 - 82.04 \times (10) = 4,628$). Each UABP is shown with its lead and lag effects as a different line in the Fig. 4.3.

Hypothesis 4: Adjustment effects from UABPs increase over time As visible in Fig. 4.3, adjustment effects do change over time. Table 4.2 reports that the impact of UABPs on the long-term adjustment effects (e.g. Long-term impact on UABPlag (t-1) is mixed. The impact of running additional UABPs on the direct adjustment (lag1 (t-1) and lead1 (t+1)) are significantly positive (49.75 and 47.94). This means that the pre-and post-promotion troughs directly surrounding UABPs, are getting smaller with an increased usage of UABPs over time. The effects on the other lag (lag2 and 3: 3.20/-10.30) and lead (lead2 and 3: -18.4/7.49) variables are mixed regarding the direction of their impact. While being smaller in magnitude the effects are still statistically significant. The initial hypothesis that adjustment effects increase over time cannot be supported. Furthermore despite the net effect of promotions being negatively affected over time as the sales impact of UABP is decreasing, this effect is partly offset by the net decreasing of promotional dips. These results combined with the long-term sales impact on the promotion period itself indicates a generally declining, yet in the beginning still positive, effect of UABPs over time, which will later be discussed in detail.

4.5 Summary and Discussion

Overview This chapter discusses the long-term impact Uniform Across the Board Promotions (UABP) have on sales of a value fashion retailer. Specifically the focus is on the (1) sales bump as well as their respective pre- and post-promotion dips, and (2) and it is analyzed how they develop over time. Furthermore (3) the impact an increased use of UABPs have on long-term store-level baseline sales is being analyzed in this chapter.

4.5.1 Academic Implications

Despite mixed results in the marketing literature regarding adjustment and permanent effects of promotions, relatively clear implications for the three topics regarding Uniform Across the Board Promotions have been found.

 Sales bumps and pre-and post-promotion dips: As already discussed in Chap. 3, UABPs are an effective means to increase short-term sales, leading to a significant sales bump during the UABP which is also evident in the daily sales level data of the retailer. While the net effect is still positive, the sales impact of UABPs is reduced by significant pre-and post-promotion dips, that are larger than those for traditional price promotions in this dataset. Particularly the period leading up to the UABP period experiences significant dips.

As discussed earlier this can be due to "stockpiling" or in this case rather purchase acceleration. The magnitude of the effect is likely positively influenced by the particularities of the category (fashion items), which is characterized by better storability and "lower out of stock costs" than e.g. consumables is. Moreover the increased flexibility and increased certainty of finding the desired article on sale, provides the consumer with advantages that lower the search costs and offer a higher incentive to shift purchases to UABP periods (Urbany et al. 1996). This can lead to higher sales bumps and deal troughs on a store level, compared to traditional price promotions.

Directionally, on average over all stores and throughout the entire year, the UABP decreases sales in the 3-day period preceding the UABP by on average 28 % per day assuming baseline sales of ~€2,000 for an average store (largest decrease around 50 % the day prior to the UABP) and by approximately 18 % per day in the 3 day period following the UABP (again the day directly following the UABP faces the biggest decrease: ~50 %). During UABPs sales are by approximately 275 % higher each day than during regular periods. The net average sales gain solely attributable to the UABP is approximately 76 %, after subtracting pre-and post-promotion dips. Hence approximately 28 % of the original increase (76 % divided by 275 %) is the net percentage gain from a UABP (compare to Macé and Neslin (2004) for methodology).

When comparing to the results of previous research that focused on traditional price promotions, one has to make a caveat for category characteristics as those tend to play quite an important role, as shown by Macé and Neslin (2004) who have found promotion bumps ranging between 23 % (gum) and 200 % (machine dish detergent) for individual categories. However this study would like to directionally do just that, as Macé and Neslin (2004) have shown that traditional promotions (with same promotion depth of 20 %) lead to an average increase of 85 % (compared to the 275 % for the UABP). The net average sales gain, again considering pre- and post-promotion dips in their research is 57 % (vs. 76 % for UABPs). The respective percentage net gain is 67 % (vs. 28 % for UABPs) (see also Table 4.3).

This indicates, that while the overall impact of UABPs in this example is stronger (in line with the results introduced in Chap. 3), the relative dips also seem to be larger for UABPs (indicated through lower percentage net gain), which, while partly category dependent, is in line with the original expectations, that UABPs cause larger changes in consumer purchase behavior than traditional promotions do. As clearly shown by the large range of outcomes in the paper of Macé and Neslin (2004) it is strongly encouraged to repeat this study based on different categories. Furthermore customer and store specific characteristics should be included as Chaps. 3 and 5 shows that those have a significant impact on the relative performance of UABPs.

2. Impact of UABPs on promotion sensitivity (i.e. evolvement of sales bumps and deal troughs: The long-term effects of UABPs on their future effectiveness is

Promotion type	Current-period avg. sales gain ^{b,c}	Net avg. sales gain ^{b,c}	Percentage net gain ^{b,c}
Uniform Across the Board Promo- tion (this thesis)	275 %	76 %	28 %
Traditional price promotion ^a	85.1 %	56.8 %	66.7 %

Table 4.3 Estimated incremental store sales gains

^aAs per 20 % price decrease (comparable to UABP), Macé and Neslin (2004)

^bAverage sales impact of a 20 % price reduction, with baseline sales indexed at 100. 85.1 % current-period increase in sales: after subtracting pre- and post-promotion dips, the net sales gain is 56.8 %. This implies that 66.7 % (58.8/85.1) of the current-period average sales gain represents truly incremental sales (Macé and Neslin 2004)

^cFor UABPs assuming 3 days of running the promotion; lag period and lead period also 3 days each. The main effects which are attributed to the promotion (Table 4.1) have been set in relation to the median daily sales of all stores to calculate the respective figures

negative, i.e. the more UABPs are being offered to customers over time, the smaller the immediate sales increase from each one of them is. This is in line with the literature stating that consumer form expectations about future promotion availability which can lead to them deferring purchases on subsequent occasions. In the long-run they are becoming less promotion or specifically UABP sensitive, i.e. they can forego a deal as they expect to get a better one in the future (Gönül and Srinivasan 1996; Mela et al. 1998).

As Uniform Across the Board Promotions have a strong impact on short-term sales (outperforming traditional price promotions), managers whose performance is measured on like-for-like sales growth, are often forced to continue offering UABPs to their customers once they have introduced them. This makes the use of UABPs predictable and customers learn to adopt their behavior accordingly. This change in behavior can further be shown by some customers coming to the stores a few days ahead of an UABP are sometimes allowed to "reserve" items instead of buying them instantly, which further increases sales bumps and deal troughs, according to discussion with the retail managers of the fashion stores. The next section will address the practical implications of these results.

With regards to the adjustment effects, the immediate pre-and post-promotions dips (one period before and after) of UABPs are also decreasing over time, which is a sign of the overall decreasing impact of UABPs on the sales level. Eventually, the impact of UABPs on sales could become negligible as consumers get used to them and form lower reference prices for the store and its articles. Sales increases could then only be motivated by offering ever lower prices at lower gross margins further reducing consumer reference prices and potentially being detrimental to the customer attitude—which is a special topic to be investigated given this background.

3. As discussed earlier, the interaction term between UABPs over time and the overall sales level is negative, implying that UABPs reduce baseline sales for a retailer in periods where no promotions are run. Albeit this negative trend is relatively small compared to the positive short-term impact of UABPs—it is

significant and further supported by the existence of a long-term negative impact on the outer pre-and post-promotion dips. This is explained by the adaption level theory and reference price concept, as consumers get used to lower prices and hence expect those prices in the future (e.g. Jacobson and Obermiller 1990; Lattin and Bucklin 1989; Greenleaf 1995). While anecdotal evidence suggested similar results for DIY articles ("*Praktiker example*") future research on the long-term effects of UABPs should try to base this on household-level data, to check whether different customer groups react differently or whether different customer groups are being attracted during the promotion (e.g. Neslin and Shoemaker 1989) and whether these result also hold for other retailers with a different product portfolio. Furthermore a linear decline over the number of promotions has been assumed, which going forward, can, be extended to more sophisticated non-linear, long-term decay models.

4.5.2 Retailer Implications from Long-Term Effects of Uniform Across the Board Promotions

Despite the existence of adjustment effects (pre-and post-promotion dips), UABPs have a positive net effect on sales around the actual promotion period and can hence be considered a useful tool to increase short-term sales (see also Table 4.1). As over time their effectiveness decreases, so do the adjustment effects—as a result the net effect continues to be positive, albeit on an eventually much smaller level. This of course only considers the net sales impact—the net gross profit might be negative, depending on the retailer's margin, as already shown in Sect. 3.5.1.

To hence achieve the same sales or gross profit impact as when UABPs are first run, retailers need to increase the use of UABPs, which however will have negative effects on the baseline sales of the retailers. Once UABPs are introduced and retailers try to match sales levels of previous periods, they ceteris paribus enter in a vicious circle that they cannot escape, given the long-term negative impact on promotion sensitivity and baseline sales. However reducing once introduced frequency of promotions can have a negative effect on market shares as the net price to consumers is increased (Ailawadi et al. 2001).

Getting of the UABP drugs, while not losing too much sales, i.e. market share requires retailers to apply UABPs (and other forms of promotions) more creatively, segmenting their customers and getting the more promotion prone customers off a bit softer than the non-promotion prone. Furthermore to decrease the adjustment effects and mitigate a potential change in behavior of the customers, retailers need to a.) be disciplined with restricting UABPs to the actual promotional period and to a pre-selected target group but also b) be discrete and unpredictable as to when the next UABP is going to take place.

As for any type of short-term performance enhancing drug, another solution is of course not starting with them (UABPs) at all, as they eventually catch up to you. Not starting with a regular use of UABPs or swiftly finding suitable means of

reducing UABPs would even be stressed, if UABPs also have a negative impact on further customer and retailer KPIs such as for example customer loyalty, customer satisfaction, attitude towards the retailer, brand equity and so forth. This, along with other household-level studies on UABPs is an area which will be discussed in Chap. 5.

References

- Abraham MM, Lodish L (1993) An implemented system for improving promotion productivity using store scanner data. Market Sci 12(3):248–269
- Ailawadi KL, Lehmann DR, Neslin SA (2001) Market response to a major policy change in the marketing mix: learning from Procter & Gamble's value pricing strategy. J Market 65 (1):44–61
- Assuncao JL, Meyer RJ (1993) The rational effect of price promotions on sales and consumption. Manag Sci 39(5):517–535
- Bell DR, Chiang J, Padmanabhan V (1999) The decomposition of promotional response: an empirical generalization. Market Sci 18(4):504–526
- Blattberg RC, Neslin SA (1990) Sales promotion: concepts, methods, and strategies, 1st edn. Prentice-Hall, Englewood Cliffs, NJ
- Blattberg RC, Briesch R, Fox EJ (1995) How promotions work. Market Sci 14(3):122–132
- Böcking D (2012) Dramatische Hauptversammlung: Praktiker droht Aktionären mit Pleite. http:// www.spiegel.de/wirtschaft/unternehmen/praktiker-sanierungskonzept-sorgt-fuer-aerger-aufhauptversammlung-a-842634.html. Spiegel Online, last accessed 15 Apr 2013
- Christen M, Gupta S, Porter JC, Staelin R, Wittink DR (1997) Using market-level data to understand the promotion effects in a nonlinear model. J Market Res 34(3):322–334
- Davis S, Inman J, McAllister L (1992) Promotion has a negative effect on brand evaluations or does it? Additional disconfirming evidence. J Market Res 29(1):143–148
- Dekimpe MG, Hanssens DM, Silva-Risso JM (1999) Long-run effects of price promotions in scanner markets. J Econometrics 89(1–2):269–291
- Doyle P, Saunders J (1985) The lead effect of marketing decisions. J Market Res 22(1):54-65
- Erdem T (1996) A dynamic analysis of market structure based on panel data. Market Sci 15 (4):359–378
- Gönül F, Srinivasan K (1996) Estimating the impact of consumer expectations of coupons on purchase behavior: a dynamic structural model. Market Sci 15(3):262–279
- Greenleaf EA (1995) The impact of reference price effects on the profitability of price promotions. Market Sci 14(1):82–104
- Grover R, Srinivasan V (1992) Evaluating the multiple effects of retail promotions on brand loyal and brand switching segments. J Market Res 29(1):76–89
- Guadagni PM, Little JDC (1983) A logit model of brand choice calibrated on scanner data. Market Sci 2(3):203–238
- Helson H (1964) Adaption-level theory, 1st edn. Harper & Row, Oxford
- Hielscher H (2010) Praktiker verordnet sich Reperaturprogramm. http://www.wiwo.de/unternehmen/ harter-entzug-praktiker-verordnet-sich-reparaturprogramm/5687658.html. Wirtschaftswoche Online, last accessed 15 Apr 2013
- Hirzel J (2012) Baumarktkette Praktiker fürchtet um seine Existenz. http://www.focus.de/ finanzen/news/unternehmen/es-geht-ums-ueberleben-baumarktkette-praktiker-fuerchtet-umseine-existenz aid 777230.html, Focus Online/Reuters/dpa. Last accessed 15 Apr 2013
- Jacobson R, Obermiller C (1990) The formation of expected future price: a reference price for forward-looking consumers. J Consum Res 16:420–430

- Jedidi K, Mela CF, Gupta S (1999) Managing advertising and promotion for long-run profitability. Market Sci 18(1):1–22
- Judge GG, Griffiths WE, Hill RC, Lütkepohl H, Lee TC (1985) The theory and practice of econometrics, 2nd edn. Wiley, Singapore
- Kalwani MU, Yim CK (1992) Consumer price and promotion expectations: an experimental study. J Market Res 29(1):90–100
- Kalwani MU, Yim CK, Rinne HJ, Sugita Y (1990) A price expectation model of customer brand choice. J Market Res 27(3):251–262
- Krishna A (1992) The normative impact of consumer price expectations for multiple brands on consumer purchase behavior. Market Sci 11(3):266–286
- Krishna A (1994) The impact of dealing patterns on purchase behavior. Market Sci 13(4):351-373
- Läsker K (2012) Rabatte sind wie süße Drogen. http://www.sueddeutsche.de/wirtschaft/sparkusbei-der-baumarktkette-praktiker-rabatte-sind-wie-suesse-drogen-1.1322133. Süddeutsche Online, last accessed 15 Apr 2013
- Lattin JM, Bucklin RE (1989) Reference effects of price and promotion on brand choice behavior. J Market Res 26(3):299–310
- Lim J, Currim IS, Andrews RL (2005) Consumer heterogeneity in the longer-term effects of price promotions. Int J Res Market 22(4):441–457
- Macé S, Neslin SA (2004) The determinants of pre- and postpromotion dips in sales of frequently purchased goods. J Market Res 41(3):339–350
- Mela CF, Gupta S, Lehmann DR (1997) The long-term impact of promotion and advertising on consumer brand choice. J Market Res 34(2):248–261
- Mela CF, Jedidi K, Bowman D (1998) The long-term impact of promotions on consumer stockpiling behavior. J Market Res 35(2):250–262
- Neslin SA, Schneider Stone LG (1996) Consumer inventory sensitivity and the postpromotion dip. Market Lett 7(1):77–94
- Neslin SA, Shoemaker RW (1989) An alternative explanation for lower repeat rates after promotion purchases. J Market Res 26(2):205–213
- Nijs VR, Dekimpe MG, Steenkamp JB, Hanssens D (2001) The category-demand effects of price promotions. Market Sci 20(1):1–22
- Pauwels K, Hanssens DM, Siddarth S (2002) The long-term effects of price promotions on category incidence, brand choice, and purchase quantity. J Market Res 39(4):421–439
- Pinheiro J, Bates D, DebRoy S, Sarkar D (2013) Linear and nonlinear mixed effects models. R package version 3.1
- Urbany JE, Dickson P, Kalapurakal R (1996) Price search in the retail grocery market. J Market 60 (2):91–104
- Van Heerde HJ, Leeflang PSH, Wittink DR (2000) The estimation of pre-and post-promotion dips with store-level scanner data. J Market Res 37(3):283–395
- Wickham H (2009) Plyr: tools for splitting, applying, and combining data. R package version 1.9
- Winer RS (1986) A reference price model of brand choice for frequently purchased products. J Consum Res 13(2):250–256

Chapter 5 UABPs: Antecedents and Consequences on a Household Level

5.1 Introduction and Context to UABP Antecedents and Consequences on a Household Level

As discussed in previous chapters, Uniform Across the Board Promotions (UABPs), are a relatively new type of promotion concept that has received negative publicity in the international business press, as it has partly been blamed for the insolvency of Praktiker, a major European DIY retailer (Böcking 2012). The anecdotal evidence of a negative long-term impact of UABPs from "the Praktiker example" is supported by the analyses of store-level data of an international value fashion retailer, where strong pre-and post-promotion troughs, decreasing positive sales effects during the promotion as well as negative long-term effects on baseline sales can be observed (compare Chaps. 3 and 4). Many of those effects also exist directionally for traditional price promotions, however to a smaller extent (van Heerde et al. 2000). Despite these findings, many retailers still use UABPs and for those retailers, it is important to further understand, whether those effects also exist on a household level or whether other drivers impact the observations above.

While the previous chapters and existing research has not yet covered the household level impact of Uniform Across the Board Promotions specifically, extensive research exists that covers related topics: First a broad stream of research explains and shows *lead effects, stockpiling* and increasing *promotion sensitivity* on a household level, often explained by changes in *reference prices*. (Anderson and Simester 2004; Doyle and Saunders 1985; Mela et al. 1998). Second, another stream of research empirically analyses the adjustment effects of traditional promotions and how they are affected by category and customer characteristics, i.e. which type of customers or categories are most likely being affected by promotions (Bell et al. 1999; Macé and Neslin 2004). The third stream of research discusses the impact of price promotions and discounts on consumer behavior, e.g. the perception of a brand or the customer loyalty. Two exemplary studies are Grewal et al. (1998a, b) who analyze the promotion consequences using structural equation modeling based on an experiment with a subsequent questionnaire and

Dodson et al. (1978) who use a long-term household panel data, to show the impact of promotions on customer loyalty and brand switching.

While the research carried out and introduced in this chapter has various similarities with the above, it differs and adds value in three major points: (1) this thesis, i.e. chapter specifically focuses on UABPs. Given the distinct characteristics of UABPs as well as their increased popularity it is important to specifically analyze the impact UABPs have on consumers. This has become even more evident, as store-level results for the impact of UABPs differed in their magnitude and partly direction from traditional promotions (see Chap. 3). (2) While various studies covered either antecedents or specific consequences of promotions on a household level, no study covers the two aspects as extensive and in one single model. (3) Historical studies covering the antecedent or consequences of promotions have mostly either used panel or experimental data, while this thesis uses transaction and survey data combined. It is expected to hence provide realistic results when linking survey results for attitudes and intended behavior to actual past behavior.

In addition to the points above, this chapter also aims to provide guidance to managers using UABPs by discussing three relevant questions: (1) Can the effects observed on a store-level be confirmed by the behavior of individual consumers or are other factors affecting the performance of UABPs? (2) which customers are most likely to react to UABPs—who is the "typical UABP customer" to target? (3) what other "soft" consequences, apart from the direct impact on sales does the use of UABPs have on store image and customer behavior?

To achieve these objectives, this chapter is divided in two parts: the first part discusses the *antecedents of UABP user intensity* and the second part focuses on how the *UABP user intensity impacts four central retailer KPIs*: (1) Customer satisfaction, (2) Price and promotion image of the retailer, (3) Customer loyalty and (4) Expected promotion driven change in behavior

As stated above, the analysis will be based on a combination of household survey data, conducted among 1,641 customers of a value fashion retailer, complemented by the respective transaction data of each of these households during any Uniform Across the Board Promotion campaign carried out over the period between 2008 and 2012. Methodologically the study pursues in two steps. First a confirmatory factor analysis is used to ensure the reliability and validity of the relevant concepts as laid out in the questionnaire. Second a structural equation model is built in order to analyze the hypotheses regarding antecedents and consequences of the UABPs.

5.2 Conceptual Framework: Antecedents and Consequences

In the following the model that further explains the antecedents and consequences of UABP usage is being introduced. The first part discusses which customers, depending on socio-demographics and attitude are most likely to react to Uniform



Fig. 5.1 Antecedents and consequences of UABPs

Across the Board Promotions while the second part explains how the participation in UABPs might affect the attitude towards the retailer and future shopping behavior. The left side of Fig. 5.1 lists the antecedents and how they influence the *UABP user intensity*, which is the participation intensity of a consumer for UABPs.

When analyzing the antecedents of UABPs the focus is on three different areas. First *general socio-demographic variables*, that are commonly used to describe the composition of customers in a retail store (e.g. Bell et al. 1999; Macé and Neslin 2004; Montgomery 1997). These are helpful to segment customers and differentiate their exposure to marketing instruments and price, as suggested by Montgomery (1997). The variables used are:

- Age
- · Family size
- Income
- Gender.

There is no information available for this study on some of the typical sociodemographic information that is often derived from store-trading area characteristics (e.g. Hoch et al. 1995) as for example the education level or the ethnicity of the customers.

Second additional *search cost related variables* are being discussed, in order to describe the cost-benefit ratio of a promotion from the viewpoint of a consumer. This is based on the transaction or search cost theory that is derived from information economics that state that customers put the search (and transaction) costs in relation to the expected savings (Metha et al. 2003; Stigler 1961). Consumers hence

continue to search for a better price as long as the expected savings are still as high as or higher than the cost incurred by an additional shopping trip. Price promotions decrease the probability of finding a lower price elsewhere and consumers are more likely to terminate their price search. Promotions hence have a negative impact on the price search—the higher the absolute discount, the more negative the impact (Darke and Freedman 1993; Darke et al. 1995). The variables used for transaction cost/price search are:

- Distance to the retailer
- · Car ownership
- · Category budget.

Distance to the retailer and the ownership of a car are part of the cost side of the equation, i.e. how much effort is required to visit a retailer and search for deals, whereas the category budget, which is the money spent on fashion, implies whether any savings in this category has an overall relevant impact on the household budget or whether it is negligible (if category budget is low).

The third variable considered is *price knowledge and involvement*. This variable refers to the questions whether the customer is interested in comparing prices and whether she has a general idea about the prices in the market, as this also influences the search cost behavior of the customer.

Hypotheses regarding the impact of antecedents on UABPs The impact of consumer-demographics as a predictor of usage of promotions has been frequently discussed in the marketing literature while reporting mixed results given that many attempts to find a correlation between the two has in the past only found weak or no correlation for the majority of parameters (e.g. Bell et al. 1999; Jolson et al. 1987; Macé and Neslin 2004). Nevertheless empirical research shows, that some consumer demographics do influence the deal proneness of consumers and their likelihood to shop during promotions (e.g. Hoch et al. 1995; Macé and Neslin 2004). Some of those demographics will be discussed in the following:

Age The age of the consumer has been among one of the most frequently discussed characteristics and as for the overall subject, results are mixed (e.g. Jolson et al. 1987; Macé and Neslin 2004; Teel et al. 1980). There have been suggestions, that older customers have more time and should hence be more deal prone (Bell et al. 1999), that promotion elasticities increase with age (Macé and Neslin 2004) or that deal-prone housewives tend to be older (Webster 1965). Other research has however found that older customers are less price sensitive (Ainslie and Rossi 1998; Hoch et al. 1995) and that young customers are more likely to react to coupons (Teel et al. 1980). Macé and Neslin (2004) explain the difference in results above by the fact that they and Bell et al. (1999) consider UPC characteristics, which e.g. Hoch et al. (1995) do not control for.

In the case of UABPs and this particular product category, fashion, it can be expected that older customers tend to be more promotion prone, as UABPs offer them a very broad selection of items. Moreover older customers might be more likely to carry out shopping for their children and husbands as opposed to customers in their late teens and early twenties.

H1: Age is positively correlated with UABP user intensity

Family size As mentioned above, the breadth of the promotion across the entire assortment is a specific characteristic of UABPs, which differs from traditional brand based promotions. This allows customers to purchase fashion items for the entire family during one shopping trip, hence reducing search costs and increasing efficiency. Macé and Neslin (2004) have found that post-promotion dips, an indicator for stockpiling and purchase acceleration and hence promotional activity are positively correlated with household size. This is in line with Hoch et al. (1995) who find that larger families are more price sensitive and hence try to profit from promotions while optimizing their search costs.

H₂: Family size is positively correlated with UABP user intensity

Income One could expect a negative correlation between income and deal proneness as the opportunity costs for less affluent households are often lower, allowing them to invest more time searching for items on promotion. Additionally, for less affluent consumers, money saved due to a promotion might be higher on a relative basis, than the same absolute savings would be for affluent consumers. However existing research has often not found any significant effects between income and promotion sensitivity, which might have been driven by the relative small absolute savings on the consumable categories tested (Bell et al. 1999; Macé and Neslin 2004). Surprisingly, which was also against the expectation of the authors, Kwon and Kwon (2007) have found a positive income effect, i.e. higher incomes have a higher likelihood to use coupons. Despite the findings in the literature this study still expects less affluent consumers to try to shift more of their purchases into UABP periods, as (1) this study tests for fashion items, for which savings are also in absolute terms relevant for the customer and (2) the customers can time their purchases accordingly, as the out of stock costs on fashion are negligible (compared to e.g. groceries).

H₃: Income is negatively correlated with UABP user intensity

Gender (% of females) While Jolson et al. (1987) have found no correlation between deal proneness and gender, this study assumes a positive relationship, which is retailer specific. For the value fashion retailer used for this study, most customers are female and once can hence assume that they keep up more with retailer specific UABP promotions. This might however be more category or retailer specific than finding a (stereo-type) male-female explanation for this hypothesis.

H₄: Women exhibit a higher UABP user intensity

Distance to the retailer The results of Chap. 3 have shown that retailers in rural areas i.e. where customers have a longer commute, perform relatively better during UABPs. The absolute performance might however be higher in more central locations, given the higher natural footfall partly owned to the lower opportunity costs of visiting the store. While customers with a longer *distance to the retailer* might shop relatively more during UABPs compared to non-promotional periods, customers with a short commute might in absolute terms shop more during UABPs and during non-promotional periods given their lower search and transaction costs.

H₅: Distance to the retailer is negatively correlated with UABP user intensity

Car ownership Further reduces the opportunity costs for a customer, while facilitating the ability to stockpile and hence profiting more from the UABP. This has been confirmed by Macé and Neslin (2004) who find a positive relationship between stockpiling, promotion participation and car ownership.

H₆: Car ownership is positively correlated with UABP user intensity

Category budget Macé and Neslin (2004) find that post-promotion dips are larger in categories, with a high budget share, i.e. where customers spend a high percentage of their income. This can again be explained by transaction costs and information economics, as if the expected savings, meaning the category budget times the expected percentage discount, is higher, customers are willing to time their purchases accordingly and/or visit the stores in which they are being offered discounts.

H₇: UABP user intensity is positively correlated with category budget

Price knowledge and involvement Very often consumers do not have precise knowledge of prices (Dickson and Sawyer 1990) but rather use heuristics, i.e. simplified external cues, such as whether an article is on promotion, to form a purchase decision. Very often those external cues are sufficient to form a view about a price, retailer or brand (Krishnamurthi and Raj 1988; Mazumdar and Monroe 1990). One could hence assume that in absence of perfect price knowledge or involvement, the customer will use the UABP as a cue that positively stimulates her buying decision. Price knowledge is consequently negatively correlated with UABP participation.

H8: UABP user intensity is negatively correlated with Price knowledge and involvement

5.2.1 Consequences of UABPs

To gain a better understanding of whether it is advisable for a retailer to perform UABPs, it is important to not just look at the impact this type of price promotion has on sales but also consider non-monetary consequences the Uniform Across the Board Promotions have on consumer behavior and attitude. This is relevant as Anderson et al. (1994) show that customer satisfaction and long-term economic performance are positively correlated. Moreover several studies have found that a strong link between customer satisfaction and loyalty exists, which in turn is influenced by the (quality and price) image of the retailer (Anderson and Sullivan 1993). Retailers hence need to consider the impact on those variables as those can be indicators for future performance.

Specifically this chapter discusses the impact of UABPs on (1) the customer satisfaction and the (2) price image of the retailer. Moreover it will be discussed whether UABPs have an impact on (3) customer loyalty and (4) whether UABPs will influence the shopping behavior of existing customers.

Customer satisfaction and its antecedents have been extensively studied in the marketing literature and can be broadly characterized as the post-purchase evaluation given pre-purchase expectations (Anderson and Sullivan 1993). The expected price of a product plays an important role and Oliver and Winer (1987) show that consumers adapt their expectations according to past prices. A positive deviation from the expected, historic price and quality levels (Helson 1964) has a positive impact on customer satisfaction. This is supported by the reference price concept which suggests that a positive deviation from the regular or expected price is booked as a "profit" in the consumer's utility function, which hence increases the satisfaction of the customer (Kalyanaram and Little 1994; Mayhew and Winer 1992). It can hence be expected that UABPs, where customers receive the discount on the desired article without having to compromise on selection or quality positively influences the customer satisfaction.

H₉: UABP user intensity is positively correlated with customer satisfaction

Price image: of the retailer Another critical topic is how UABPs affect the price image of the retailer. In the literature, there are particularly two factors that influence the price image: regular price-quality ratio and the depth and frequency of price promotions (Nyström 1970). Regarding the latter the results are mixed. While some empirical research shows, that price promotions have a positive influence on the price image (Desai and Talukdar 2003; Schindler and Rogulic 1998), others did not find a correlation between the two (Cox and Cox 1990). However all of those studies focused on the impact promotions of individual articles have on the retailer's image—while during UABPs not just selected items are put in focus to signal a certain price image for the retailer but rather the entire retail assortment serves as a signal, which is why it can be expected that UABPs will have an impact on the price image of the retailer.

H₁₀: UABP user intensity is positively correlated with the price image of the retailer

Customer loyalty It is empirically proven that promotions can be used to shape *customer loyalty* (Rothschild and Gaidis 1981) which is furthermore affected by customer satisfaction (which is expected to be positively influenced by UABPs). However according to the self-perception theory consumers might attribute their buying during UABPs to the presence of the UABP itself, rather than their actual

affiliation with the retailer (Dodson et al. 1978). While it is hence unclear whether buying will solely be attributed to the UABP, one could still expect some positive impact on customer loyalty which is further supported by the indirect positive effects from loyalty and price image.

H11: UABP user intensity is positively correlated with customer loyalty

Expected promotion driven change in behavior The promotion driven change in behavior refers to the phenomenon that customers might shift purchases from non-promotion periods to UABP periods, causing large pre-and post-promotion dips on a store-level (see Chap. 4). This change in behavior can be explained by changes in the reference price of the customer and a learning process in which customers adapt their shopping behavior to frequently repeated promotions. Furthermore the fact that during UABPs all articles will be promoted reduces the uncertainty of finding the right article and hence reduces the risk of potential stock-out costs for the consumer, making the deferring of purchases even more prominent than during normal promotions (Gönül and Srinivasan 1996).

 H_{12} : UABP user intensity is positively correlated with change in shopping behavior (stockpiling and anticipation)

5.3 Method

5.3.1 Household Level Survey and Transaction Data

The company provided actual transaction data of a random sample of 20,000 customers from an international value fashion retailer's customer database. The sample has the same distribution in terms of gender and age as the overall population of the customer database and was randomly distributed geographically. Every customer in the sample had provided an email-address and did participate in at least one UABP in the period 2008–2012. For each customer, certain socio-demographic information that the customers have provided when signing up for the newsletter were available as well as the information whether she had participated in a specific UABP and what the total amount of purchases has been during each UABP. There is no information available on customer spending in non UABP periods, nor on the composition of the shopping basket.

All of those customers had been sent a personalized e-mail request by the retailer to complete a questionnaire, which was designed using professional survey software, carrying logos of the retailer and the university. The participants of the survey had been offered the opportunity to win a gift certificate of the retailer valued between ≤ 10 and ≤ 100 , overall prizes in a total value of ≤ 500 had been given away. The survey consisted of questions related to, for example, their general attitude towards the retailer, customer satisfaction, their shopping behavior as well as

UABP specific information. Items related to each construct were formulated as statements which were to be answered on a seven-point Likert scale (l = strongly disagree, 7 = strongly agree)—further details are provided in the Appendix B. Furthermore the customer was asked to provide some socio-demographic information about herself and her household (see Appendix B for more information on the question categories).

The questionnaire consisted in total of around 75 questions; with the median time for answering to all of them was slightly above 10 min (606 s). Respondents not completing the entire questionnaire or obviously answering the questions arbitrarily (e.g. family size 100 members, age 0 years) had been excluded from the sample, leaving 1,641 adult completes, for which all questions had been answered properly and the actual transaction history during UABPs was available. It is to be noted, that there might be a gender and slight age bias in the data sample, with above 90 % of respondents being female and 70 % of respondents being between 30 and 50. With the gender bias being slightly over-proportionate to the total share of male customers for the retailer both values are still directionally in line with the target customer group of the store.

5.3.2 Methodology to Analyze Antecedents and Consequences: CFA and SEM

Latent variables i.e. constructs, derived from the items in the questionnaire and the transaction data serve as an input to testing the hypothesis H1-H12 on the antecedents and the consequences of UABPs using structural equation modeling (SEM). The validity and reliability of the constructs has been assessed with a confirmatory factor analysis (CFA) in which each item is prescribed to be loaded on one specific latent variable; thus, a customer loyalty item is related to the customer loyalty factor and not to any other factor. A standardized solution has been calculated using the lavaan-package in the statistical software R (Rosseel 2012), which shows that the items load highly on their corresponding constructs and are mostly above the recommended 0.40 item reliability threshold. The few exceptions that are below the threshold are in the customer satisfaction category as well as one in the customer loyalty variable. Those items are still kept to preserve conceptual comprehensiveness and as keeping those in still improved construct reliability and average variance extracted values (AVE). Overall the measures exhibit good reliability and validity of the constructs with composite reliabilities above the recommended 0.70 threshold (see Table 5.1).

Furthermore the fit indices of the structural equation model show a good fit of the measures, indicated by a Comparative Fit Index (CFI) of 0.90, which is at the recommended threshold of 0.90, a RMSEA of 0.06, which is in between the 0.05 for good and 0.08 for moderate fit (MacCallum et al. 1996) and a SRMR of 0.06, which is below the 0.08 threshold and indicates good fit (Hu and Bentler 1998). The NFI

Construct	Item	IR	CR	AVE
UABP user intensity	Actual participation (absolute numbers)	0.55	0.78	0.55
	Relative participation (% since first participation)	0.53		
	Recent participation (has participated in at least two recent UABPs)	0.56		
Customer satisfaction	"I am satisfied with the selection of the assortment"	0.33	0.85	0.47
	"I am satisfied with the friendliness of staff"	0.54		
	"I am satisfied with the store design"	0.70		
	"I am satisfied with the cleanliness of the store"	0.62		
	"I am satisfied with the accessibility of stores"	0.28		
Price image	"The assortment of the retailer is worth its price"	0.74		
	"The prices at the retailer are justified"	0.86		
Price image	"The quality of the goods justifies their price"	0.84	0.93	0.81
Customer loyalty	"I would recommend the retailer to my friends"	0.81		
	"I will visit this retailer more often in the future"	0.35		
Customer loyalty	"I will continue to visit this retailer in the future"	0.50	0.79	0.56
Expected promotion driven change in	"I deliberately buy at times when UABPs take place"	0.58		
behavior	"I try to plan my purchases according to when UABPs take place"	0.49		
Expected promotion driven change in	"I expect to be able to continue to buy at this retailer at reduced prices"	0.36	0.62	0.36
behavior Price knowledge and	"I am very interested in the price of articles and I do compare prices"	0.50		
involvement	"I know quite precisely what a similar article would cost at one of the retailer's competitors"	0.68		
Price knowledge and involvement	"I am very involved regarding my decision on where to buy clothes"	0.92	0.77	0.46
	"I am often contemplating where to best buy fash- ion articles"	0.67		

Table 5.1 Indicator and construct reliabilities from CFA

and NNFI are both at 0.88, which is around the recommended threshold of 0.90 indicating a moderate fit (Bentler and Bonett 1980) of the model.

The Chi-square test was used to explore that the relationships proposed in the model provide a plausible explanation to the relationships that exist in the data. The χ^2/df of 5.9 is above the 5.0 threshold which is the upper bound of the threshold being discussed in the literature (Wheaton et al. 1977). However this measure of fit is less important in the outlined case, given the large sample size, which makes the chi-square test less relevant (Burnham and Anderson 2002). Due to these facts and despite the relatively high χ^2/df ratio no concepts or variables have been altered or deleted.

The Fornell/Larcker criteria (1981), which states that the squared correlation of each construct with any other construct needs to be below the average variance extracted for the respective construct is met. This means that there is sufficient discriminant validity in the model, as concepts that are supposed to be unrelated are in fact unrelated. Table 5.2 gives an overview of the individual constructs and their correlation among each other.

Variable	М	SD	CR	AVE	1	2	3	4	5
UABP user intensity	2.99	1.81	0.78	0.55	1.00				
Customer satisfaction	5.46	0.92	0.83	0.50	0.13	1.00			
Price image	5.03	1.21	0.93	0.82	0.05	0.55	1.00		
Customer loyalty	5.36	1.21	0.79	0.56	0.15	0.72	0.70	1.00	
Expected promotion driven change in	4.94	1.20	0.72	0.47	0.24	0.18	0.20	0.24	1.00
behavior									

Table 5.2 Correlations and measurement information

5.4 Empirical Results

The structural equation model shows, that five out of eight hypotheses regarding the impact of antecedents on UABPs can be supported at the 5 % level and one hypothesis can be supported at the 10 % level while two hypotheses cannot be supported at the 10 % level and hence those antecedents have no significant impact on the likelihood to participate in Uniform Across the Board Promotions. Out of the four hypotheses tested regarding the consequences of UABPs, three can be supported, while one cannot be supported at the 10 % level. Table 5.3 and Fig. 5.2 give an overview of the impact of the individual antecedents and consequences that will be further discussed below.

5.4.1 Results Regarding the Antecedents of UABPs

As highlighted in Table 5.3, the analysis shows that there is a positive relationship between the age of the shopper and the UABP participation, i.e. the older the customer, the more likely she is to participate in UABPs (hence supporting H_1). Regarding family size, H_2 is supported, as larger families participate more actively in UABPs than small families do. Income does not influence the participation probability in Uniform Across the Board Promotions, as other than expected, households with a lower income do not participate more intensively in the UABPs (not supporting H_3) than affluent ones. In this example gender does have an impact, as women seem to be more UABP prone than men are, as they have a significantly higher likelihood of participating in this type of promotion (supporting H_4). As this might be retailer specific, it will be further discussed later in this chapter.

Concerning the hypotheses regarding the variables that could be influenced by transaction costs and search costs, the results are mixed. As expected, the distance a customer has to commute to the retailer, i.e. the effort it takes her to participate in an UABP, has a negative impact (supporting H_5) while having a car, which eases the effort and allows for stockpiling has a positive impact on the likelihood for her to participate (supporting H_6). However the category budget, i.e. what a household

		Dependent variable					
Нуро-	Independent	User	Customer	Price	Customer	Change in	
thesis	variable	intensity	satisfaction	image	loyalty	behavior	
H ₁ (+)	Age	0.208*					
		(0.116)					
H ₂ (+)	Family size	0.450***					
		(0.115)					
H ₃ (-)	Income	$-0.081^{n.s.}$					
		(0.118)					
H ₄ (+)	Gender (%	1.317*					
	females)	(0.486)					
$H_{5}(-)$	Distance to	-0.290 **					
	retailer	(0.115)					
H ₆ (+)	Car	1.683***					
	ownership	(0.450)					
H ₇ (+)	Category	$-0.069^{\text{n.s.}}$					
	budget	(0.115)					
${ m H}_{8}(-)$	Price	0.341**					
	knowledge	(0.115)					
H ₉ (+)	UABP user		0.027***				
	intensity		(0.007)				
H ₁₀	UABP user			0.014 ^{n.s.}			
(-)	intensity			(0.010)			
H ₁₁ (+)	UABP user				0.020**		
	intensity				(0.007)		
$H_{12}(+)$	UABP user					0.076***	
	intensity					(0.013)	

 Table 5.3
 Antecedents and consequences of Uniform Across the Board Promotions

n.s. not significant

Notes: standardized parameters are shown, and standard errors are in parentheses p < 0.10; p < 0.05; p < 0.001

spends on fashion per month does not influence its likelihood to participate in UABPs (not supporting H_7), which is surprising, as the potential savings for those households should in theory be larger. Against the initial assumption, price knowledge and involvement does not negatively impact the likelihood of costumers to participate in UABPs—in fact price knowledge has a positive impact on UABP user intensity (not supporting H_8).

The standardized parameters in Table 5.3 further highlight the magnitude of the impact on the likelihood to participate in UABPs. Taken aside gender and car ownership, as these variables might have a particularly strong impact due to category and retailer specific characteristics discussed separately, family size has the strongest impact on the likelihood to participate in UABPs ($\gamma_{family size} = 0.454$), followed by the general price knowledge and age of the customer. The negative impact of income and category budget is negligible.

5.4 Empirical Results



Fig. 5.2 Results of model estimation. * $p \le 0.10$; ** $p \le 0.05$; *** $p \le 0.001$; n.s., not significant; notes: completely standardized parameters are shown

5.4.2 Results Regarding the Consequences of UABPs

As further highlighted in Table 5.3 and Fig. 5.2, UABPs have a significant influence on some important retailer KPIs which is discussed in hypotheses H_9-H_{12} .

First UABPs have a positive impact on the customer satisfaction, meaning customers who often shop during UABPs are significantly more satisfied with the retailer than customers who do not use UABPs (supporting H₉). However UABPs do not positively influence the overall price image of the retailer, nor do they seem to hurt the price image (not supporting H₁₀). The direct effect of UABPs on customer loyalty is also positive (supporting H₁₁), indicating that customers that are more likely to participate in UABPs are also more likely to re-visit the stores of the retailer. Despite the higher likelihood of customers returning to the store, those customers are in fact less likely to visit the retailer during regular price periods as they expect to shift their purchases to UABP periods and expect to always have the opportunity to shop during those type of promotion (supporting H₁₂).

When also considering the indirect effects of UABPs on customer loyalty and expected change in behavior, the total impact of UABPs become even more evident, as both total effects are significant at the 1 % level. The total, direct and indirect impact of UABPs on customer loyalty is 0.082 (SD: 0.013), and the total impact on the expected change in behavior is 0.045 (SD: 0.011). Particularly the total effect of UABPs on customer loyalty benefits from the significant relationship between customer satisfaction, price image and customer loyalty, which has been a

subject of various studies in the marketing literature (e.g. Anderson and Sullivan 1993). The R^2 -values for customer loyalty and expected change in behavior are 0.66 and 0.10 respectively, indicating especially for the expected change in behavior that this is influenced by many additional factors, which will be further discussed in the following.

5.5 Summary and Discussion

5.5.1 Academic Implications

Overall, especially given the public discussions of how UABPs can hurt a retailer, its image and future business perspectives, some of the results of this analysis came rather surprising. When looking at the consequences of Uniform Across the Board Promotions, it appears that they have a positive influence not only on customer satisfaction but also on customer loyalty, meaning customers who participate in UABPs more frequently are significantly more satisfied with the retailer and are more likely to visit the retailer in the future. Customers hence seem to differentiate less between UABPs and regular price periods than previous research has found (Dodson et al. 1978).

However, whilst being satisfied and loyal customers, those customers do want to shift their shopping behavior to periods where UABPs take place and they do expect to be able to buy at reduced prices going forward. The positive attitude towards UABPs as well as the intention to shift purchases through anticipation and stockpiling fully supports the results earlier discussed when analyzing store-level data (Chaps. 3 and 4). The extraordinary promotional sales bumps followed by dips seen in store-level data will likely also exist on a household level, given the results of this study. As stated above, any potential negative long-term effects that have been found analyzing store-level data, can however be complemented by some positive effects UABPs have on a household level. Before going ahead and discussing what this implies for retailers, some results are to be shared and some caveats to be discussed in order to give guidance for future research.

In this particular context, the customers that are most likely to participate in the UABPs are older, female customers that typically shop for a larger families and are in possession of a car to go on a shopping trip. On the basis of these results as well as the consequences of UABPs, it is possible that the nature of the data sample might contain a selection bias. The data sample is based on one single retail chain (with more than 1,500 stores), with mostly (a) female loyalty customers of (b) durable goods from (c) mostly non-high-street fashion stores, whose (d) USP is clearly price. Those characteristics could hence be biased towards generally more deal prone customers compared to the overall population. Furthermore the fact that the majority of the retailer's stores are less convenient to be reached by car and that 75–80 % of its assortment is targeted towards women and children explains some of the results.

Of course these results are generally supported by literature and theory (e.g. transaction cost/price search theory), but might be influenced by category and retailer specific characteristics. In this context the relative high correlation between the latent variables, such as price image, customer satisfaction and loyalty, which are directionally not surprising but in their magnitude outstanding, also makes more sense, given that price and promotions play an integral role for the shopping decision of those customers with this particular retailer. Researchers should hence be encouraged to extend on this study to also include stores in different price segments and additional categories. Furthermore a broader study could analyze whether the UABP induced changes in consumer behavior, does also affect her purchase behavior and price expectations towards other retailers in the same segment.

It is further to be noted, that especially the R^2 of the latent variable "expected change in behavior" is relatively small (0.1), which shows that apart from UABPs there are additional factors that influence the buying behavior of customers. This concept is closely related to the reference price theory. When consumers observe a positive deviation, they are more likely to buy in the future and factors such as internal and external cues (e.g. prices observed at other retailers) will further impact the reference price in addition to the impact the UABPs have.

5.5.2 Retailer Implications from Long-Term Effects of Uniform Across the Board Promotions

As mentioned before, UABPs can help retailers to improve customer satisfaction and loyalty—however they also "mis-educate" customers and might turn them from "a normal customer of retailer A" into "a loyal and happy customer of retailer A but only during times when they are running their Uniform Across the Board Promotions". Does this mean retailers should always shy away from using this type of promotion: no! But it does mean that the general marketing strategy ought not to be based solely on UABPs, as frequently using them would condition customers in the wrong way and stopping those promotions immediately will have negative effects on sales and likely also on loyalty and satisfaction.

Currently certain sectors are facing a situation, where on-going and frequent use of UABPs in will likely change the long-term shopping behavior of customers up to a point where retailers might not be able to further increase sales "by just using more" UABPs to sustainably win or keep any customers. What does this mean for retail managers? A couple of messages could be distilled:

1. You can use UABPs for short-term benefits and as tactical measures: UABPs are a powerful tool for retailers, given how well they perform with customers and there are situations in which using them makes sense. For example UABPs could be used to attract customers that otherwise would not have visited the retailer or retailers could use them for required short-term sales increases (e.g. to generally clear stock and generate cash). However it is important not get hooked on them and use them with extreme care, as people will adopt their shopping behavior accordingly leading to a downward spiral.

- 2. *Make UABPs unpredictable, restrictive and inconvenient*: Even when only occasionally using UABPs, shoppers will try to adapt their shopping behavior. Making UABPs unpredictable (e.g. short-term notice, no clear rhythm), inconvenient (e.g. only during certain shopping hours) and restrictive (e.g. only certain customers invited or stores included) will mitigate some of those negative long-term effects and ensures that only the most promotion-prone shoppers (that might otherwise not buy at all) are actually fully adjusting their shopping behavior.
- 3. *Track the use of UABPs carefully*: Be aware what share of your business is generated during UABPs and know which customers (or at least what share of customers) predominantly shop during UABPs—set yourself clear KPIs to what an acceptable ratio would be and stick to this ratio. Therefore you have to link UABPs to a loyalty program, which again helps you to target specific customer segments for the next UABPs
- 4. *Mix UABPs with marketing instruments to sustainably win loyal customers*: While shoppers increasingly come to stores during UABPs—retailers should find alternative ways to create loyalty and satisfaction among those customers. Especially during UABPs customer can be convinced by other instruments such as a great assortment, general good price-quality ratio or friendliness of staff, so they do not want to wait for the next UABP—but want to come back sooner.
- 5. If you are really addicted—get of them slowly but surely: As with drugs UABPs will eventually erode your sales and customer base and it is essential to get off them. However in reality it is difficult to cut a major promotion campaign as sales will likely drop sharply. To get off these "drugs" slowly, in addition to doing 2, 3 and 4 as described above, retailers could segment their target customers and retail locations (Chap. 3) to assess in which sub-group or sub-stores UABPs will continue to be run—while steadily decreasing their frequency and substituting them with other types of promotions and advertising.

References

- Ainslie A, Rossi PE (1998) Similarities in choice behavior across multiple categories. Market Sci 17(2):91–106
- Anderson ET, Simester DI (2004) Long-run effects of promotion depth on new versus established customers: three field studies. Market Sci 23(1):4–20
- Anderson EW, Sullivan MW (1993) The antecedents and consequences of customer satisfaction for firms. Market Sci 12(2):125–143
- Anderson EW, Fornell C, Lehmann DR (1994) Customer satisfaction, market share, and profitability: findings from Sweden. J Market 58(3):53–66
- Bell DR, Chiang J, Padmanabhan V (1999) The decomposition of promotional response: an empirical generalization. Market Sci 18(4):504–526

- Bentler PM, Bonett DG (1980) Significance tests and goodness-of-fit in the analysis of covariance structures. Psychol Bull 88(3):588–600
- Böcking D (2012) Dramatische Hauptversammlung: Praktiker droht Aktionären mit Pleite. http:// www.spiegel.de/wirtschaft/unternehmen/praktiker-sanierungskonzept-sorgt-fuer-aerger-aufhauptversammlung-a-842634.html. Spiegel Online. last accessed 15 Apr 2013
- Burnham KP, Anderson DR (2002) Model selection and multimodel inference: a practical information-theoretic approach, 2nd edn. Springer, New York
- Cox AD, Cox D (1990) Competing on price: the role of retail price advertisements in shaping store-price image. J Retailing 66(4):428–445
- Darke PR, Freedman JL (1993) Deciding whether to seek a bargain: effects of both amount and percentage off. J Appl Psychol 78(6):960–965
- Darke PR, Freedman JL, Chaiken S (1995) Percentage discounts, initial price, and bargain hunting: a heuristic-systematic approach to price search behavior. J Appl Psychol 80 (5):580–586
- Desai KK, Talukdar D (2003) Relationship between product groups' price perceptions, shopper's basket size, and grocery store's overall store price image. Psychol Market 20(10):900–933
- Dickson PR, Sawyer AG (1990) The price knowledge and search of supermarket shoppers. J Market 54(3):42–53
- Dodson JA, Tybout AM, Sternthal B (1978) Impact of deals and deal retraction on brand switching. J Market Res 15(1):72–81
- Doyle P, Saunders J (1985) The lead effect of marketing decisions. J Market Res 22(1):54-65
- Fornell C, Larcker DF (1981) Structural equation models with unobservable variables and measurement error: algebra and statistics. J Market Res 18(3):382–388
- Gönül F, Srinivasan K (1996) Estimating the impact of consumer expectations of coupons on purchase behavior: a dynamic structural model. Market Sci 15(3):262–279
- Grewal D, Monroe K, Krishnan R (1998a) The effects of price-comparison advertising on buyers' perceptions of acquisition value, transaction value and behavioral intentions. J Market 62 (2):46–59
- Grewal D, Krishnan R, Baker J, Borin N (1998b) The effect of store name, brand name and price discounts on consumers' evaluations and purchase intentions. J Retailing 74(3):331–352
- Helson H (1964) Adaption-level theory, 1st edn. Harper & Row, Oxford

Hoch SJ, Kim BD, Montgomery AL, Rossi PE (1995) Determinants of store-level price elasticity. J Market Res 32(1):17–29

- Hu L, Bentler PM (1998) Fit indices in covariance structure modeling: sensitivity to underparameterized model misspecification. Psychol Methods 3(4):424–453
- Jolson MA, Wiener JL, Rosecky RB (1987) Correlates of rebate proneness. J Advert Res 27 (1):33-44
- Kalyanaram G, Little JD (1994) An empirical analysis of latitude of price acceptance in consumer package goods. J Consum Res 21(3):408–418
- Krishnamurthi L, Raj SP (1988) A model of brand choice and purchase quantity price sensitivity. Market Sci 7(1):1–20
- Kwon KN, Kwon YJ (2007) Demographics in sales promotion proneness: a socio-cultural approach. Adv Consum Res 34:288–294
- MacCallum RC, Browne MW, Sugawara HM (1996) Power analysis and determination of sample size for covariance structure modeling. Psychol Methods 1(2):130–149
- Macé S, Neslin SA (2004) The determinants of pre- and postpromotion dips in sales of frequently purchased goods. J Market Res 41(3):339–350
- Mayhew GE, Winer RS (1992) An empirical analysis of internal and external reference price using scanner data. J Consum Res 19(1):62–70
- Mazumdar T, Monroe KB (1990) The effects of buyers' intentions to learn price information on price encoding. J Retailing 66(1):15–32
- Mela CF, Jedidi K, Bowman D (1998) The long-term impact of promotions on consumer stockpiling behavior. J Market Res 35(2):250–262

- Metha N, Rajiv S, Srinivasan K (2003) Price uncertainty and consumer search: a structural model of consideration set formation. Market Sci 22(1):58–84
- Montgomery AL (1997) Creating micro-marketing pricing strategies using supermarket scanner data. Market Sci 16(4):315–337
- Nyström H (1970) Retail pricing: an integrated economic and psychological approach. Doctoral dissertation, Stockholm School of Economics
- Oliver RL, Winer R (1987) A framework for the formation and structure of consumer expectations: review and propositions. J Econ Psychol 8(4):469–499
- Rosseel Y (2012) Lavaan: an R package for structural equation modeling. J Stat Software 48 (2):1–36
- Rothschild ML, Gaidis WC (1981) Behavioral learning theory: its relevance to marketing and promotions. J Market 45(2):70–78
- Schindler H, Rogulic B (1998) Preismanagement im Einzelhandel: vom Preiskampf zur Steuerung des Preisimages. Absatzwirtschaft 12(98):68–75
- Stigler GJ (1961) The economics of information. J Polit Econ 69(3):213-225
- Teel JE, Williams RH, Bearden WO (1980) Correlates of consumer susceptibility to coupons in new grocery product introductions. J Advert 9(3):31–46
- Van Heerde HJ, Leeflang PSH, Wittink DR (2000) The estimation of pre-and post-promotion dips with store-level scanner data. J Market Res 37(3):283–395
- Webster FE (1965) The 'deal-prone' consumer. J Market Res 2(2):186-189
- Wheaton B, Muthen B, Alwin DF, Summers G (1977) Assessing reliability and stability in panel models. Social Methodol 8(1):84–136

Chapter 6 Summary of Results

The aim of this thesis was to introduce a relatively new type of price promotions, Uniform Across the Board Promotions, that have become more and more popular among retailers of various categories. Despite their growing popularity they had not been covered in the marketing literature thus far. In addition to introducing UABPs and putting them in context with the relevant marketing literature and theoretical background, this thesis discussed the impact UABPs have on retail sales as well as consumer behavior and attitude. More specifically, the aim was to answer the following questions:

- How do Uniform Across the Board Promotions affect short-term retail sales and how does this compare to other types of price promotions (Chap. 3)?
- Do Uniform Across the Board Promotions perform better in specific retail settings and if so, what are these? Is this UABP specific or typical for all types of price promotions (Chap. 3)?
- Are there any adjustment effects for UABPs, and if so, what is the net impact of UABPs on retail sales (Chap. 4)?
- Do these adjustment effects change over time or do they remain constant (Chap. 4)?
- Do Uniform Across the Board Promotions also affect long-term baseline sales (Chap. 4)?
- Are there any customer specific characteristics/antecedents that are an indicator for consumers to be more likely to shop during UABPs (Chap. 5)?
- What are the consequences on a household level of individuals participating in UABPs—does this e.g. affect customer satisfaction and loyalty (Chap. 5)?

The data on which the empirical research in Chaps. 3, 4, and 5 was based consisted of:

- Daily store-level sales data of more than 700 retail stores over the period of 2008–2012.
- A detailed marketing plan of the retailer, outlining every single type of promotion and advertising used over the last 4 years.

- A description of the individual stores including store characteristics (some of which had been researched).
- Transaction data of 20,000 customers having shopped during Uniform Across the Board Promotions.
- An extensive customer survey among ~1,600 customers having shopped during UABPs (and for which transaction data was available).

Methodologically, the data has been analyzed with various statistical methods, namely (1) a hierarchical Bayes model (Chap. 3) to analyze short-term sales effects and impact of store characteristics, (2) a (linear) mixed effects time series model with lead and lag effects (Chap. 4) to analyze adjustment and permanent effects of UABPs, and (3) a confirmatory factor analysis and structural equation model to discuss antecedents and consequences of UABPs (Chap. 5). Section 6.1 aims at summarizing some of the key results from Chaps. 3, 4, and 5, while also providing the core caveats of the research as well as further areas researchers are encouraged to address when discussing UABPs. A more specific summary can be found in Sects. 3.5, 4.5, and 5.5, respectively.

6.1 Summary and Discussion of Results

With regards to the short-term impact of Uniform Across the Board Promotions it was not surprising to find that UABPs have a very strong positive impact on short-term retail sales—stronger than other type of price promotions being used by the retailer. As discussed earlier, there are two likely reasons for the strong increase in overall retail sales. First, the fact that UABPs are not causing any within-store substitution effects, the way other promotions do (Gupta 1988; Sun et al. 2003) and second the increase in flexibility, which reduces the information and transaction costs for the customer (Metha et al. 2003; Stigler 1961). The latter makes UABPs more attractive for customers and hence encourages store switching which in turn positively affects the sales for the retailer.

Furthermore the analysis in Chap. 3 has shown that there are some very specific store characteristics, which favor the impact UABPs have on sales: the strongest influence on the relative performance of UABPs is the *location of the retail store*. The analysis shows that on a relative basis, inner city locations and shopping centers perform much worse than out-of-city locations such as commercial zones and strip malls, which has partly been explained by the theory of information economics. Furthermore, Chap. 3 shows that the average value per article positively influences the success of UABPs, which can be explained using the transaction cost theory. Other tested store characteristics have not been found to significantly impact the success of UABPs.

With regards to adjustment effects, the analysis in Chap. 4 shows that there are significant pre- and post-promotion dips around the sales bump caused by UABPs. These adjustment effects are an indicator for stockpiling and anticipation effects.
Despite sales troughs around the UABPs, the net effect remains strongly positive. The adjustment effects over time decrease, while the positive sales bump of UABPs also decreases; this hints at a general wear-off effect of this type of promotion. This has also been experienced in other settings, where retailers were forced to increase the number of UABP days in order to keep up (elevated) sales levels. Part of this can be explained by behavioral learning and conditioning of the consumer as discussed in Sect. 2.3. Moreover, the analysis shows that the effect Uniform Across the Board Promotions have on baseline sales is negative, which means that the more UABPs a retailer runs (cumulatively) the more their regular baseline sales decrease. This effect can be explained by consumers shifting purchases to UABP periods, which then go missing in regular price periods.

All of the research results outlined above indicate that UABPs can increase sales and can encourage store switching, i.e. taking sales from a competitor. However there is also the notion that there are negative long-term effects and there might be a "natural cap" at which amount additional sales can be lured to a specific retailer, which is shown by the negative effects of cumulative UABPs. Once this cap has been reached (through continuous UABPs) there might be going forward more inter-temporal substitution, i.e. cannibalization, than actual market share gains. This will result in no/lower sales growth and significantly reduced margins as discussed in Chaps. 3 and 4. Overall, this is however an area that requires significantly more research, as the competitive perspective and the profit perspective have not been the focus of this study and would have to be taken into account when further discussing short- and long-term effects of UABPs for the retailer.

In addition to the impact UABPs have on sales, this thesis has also shown that there are certain antecedents that positively impact the likelihood of a household to participate in UABPs. The research has shown, that older, female customers, who shop for a larger household and are more knowledgeable and interested in prices are more likely to shop during UABPs, especially if they possess a car and do not live to far away from the retailer. Interestingly their household income or the amount they spend on fashion (the respective category) did not influence the likelihood to participate in UABPs after all.

Regarding the consequences of UABPs, the data shows, that UABPs have a positive impact on customer satisfaction and customer loyalty, while they do not influence the price image of the retailer. Despite the customers being happier and more loyal though, there is also a clear tendency that these loyal customers only want to visit the store when the next UABP is taking place—hence there is clear evidence for a shift in purchase behavior on a household level.

While all the areas for further research have been discussed before, the three major caveats are (1) store and product characteristics: this research is well applicable to durable goods and particularly fashion, however product characteristics likely have a big impact and researchers should hence be encouraged, to test these results in different settings. (2) The focus and recommendations have only been based on a sales perspective, while initial discussion of a profit perspective has been made in Sect. 3.5. Going forward researchers should extend on this to discuss the impact on retailer's profits during UABPs but also in periods following UABPs.

(3) As mentioned before a general market and competitive perspective will help to discuss how UABPs are affected if competitors react and/or how the conditioning of a customer in one retailer affects her shopping behavior in another retailer.

6.2 Managerial Implications

The managerial implications of UABPs have in detail been discussed in Sects. 3.5, 4.5, and 5.5, so this chapter just aims at providing some highlights:

- 1. UABPs can be advantageous for retailers in certain settings and situations, and there are times, when they should be considered in the marketing mix. They have a stronger impact on sales than any other type of promotion and could hence, e.g., be used for a de-stockage of inventory or to drive customers to a store. This being said it does not make sense to use them early in the season, to sell "new inventory" at discounted prices, that would have likely also been bought at regular prices, nor does it make sense to try to attract existing customers to the store—focus should be on new customers or at least customers that for a long-time have not visited the store.
- 2. The long-term impact of UABPs is negative. Research has shown that despite having a positive impact on customer loyalty and satisfaction, customers are conditioned to shift purchases from regular price periods into UABP periods—in the long-run this negatively affects margins and makes the retailer dependent on this type of "promotional drug"—UABPs hence need to be used with extreme care and the behavioral learning process of the customer needs to be made as difficult as possible, e.g. making them unpredictable and in a way also uncomfortable (to only attract the really price sensitive customers).
- 3. When using UABPs it makes a lot of sense to view them as a marketing investment to attract new customers, while at the same time, the store and customer experience needs to be made impeccable for customers visiting a store for the first time during a UABP, in order for those customers to return to the store in regular price periods. As already discussed in Chap. 5, it goes without saying that the customers that shop during UABPs need to be encouraged to sign up for any type of loyalty program for the retailer to keep in contact with them in order to trigger further visits to the store. Moreover though, they must be carried away by the product and store so the UABPs in that moment become an additional reason to shop at that store—not the primary and only reason. This is potentially what has caused the problems at Praktiker, where at some point in time the UABPs became Praktiker's USP, while other competitors claimed areas such as "great customer service" or "breadth of assortment"—the UABP is not a unique tool to any retailer and hence no retailer should bet its success (or a disproportional share of their revenues) solely on it.

References

- Gupta S (1988) Impact of sales promotions on when, what and how much to buy. J Market Res 25 (4):342–355
- Metha N, Rajiv S, Srinivasan K (2003) Price uncertainty and consumer search: a structural model of consideration set formation. Market Sci 22(1):58–84

Stigler GJ (1961) The economics of information. J Polit Econ 69(3):213-225

Sun B, Neslin SA, Srinivasan KA (2003) Measuring the impact of promotions on brand switching under rational consumer behavior. J Market Res 50(4):389–405

Appendix A: De-seasonalizing and De-trending of Sales

Step 1: Use a linear time-trend regression model with monthly seasonal dummy variables to predict sales. The dependent variable is sales, the independent variables are 12 monthly dummy variables set equal to one for the respective month and 0 otherwise. Other independent variable is a linear time variable, reflecting same-store-sales growth in the research period

$$S_t^i = T_t^i + D_t^i$$

 S_t^i = store sales for store i at time t T_t^i = store trend for store i at time t

- D_t^i = dummy variable D for each month at time t for store i
- Step 2: Estimate sales if only seasonal and time-trend would be present, i.e. sales without any influence from promotions or error terms.
- Step 3: Deduct sales estimated in step 2 from actual sales for each store to derive de-trended and de-seasonalized sales for each store. Those sales are now only influenced by promotions and error terms and no longer subject to any time trend or seasonal component.

Appendix B: Survey Results (Antecedents and Consequences of UABPs)

Question	Mean	Median	SD	Included
Sociodemographic information (anteced	ents)			
How many people are you shopping for at the retailer?	2.59	2.00	1.18	Yes
What is the distance to the next [name of store] outlet (in km)	7.36	5.00	6.54	Yes
For how many years have you been customer with the company (years)	7.85	7.00	4.21	No
How often are you shopping at the retailer (per year)	13.60	10.00	17.86	No
How much are you spending per shopping trip at this retailer (€)	44.86	40.00	29.29	No
How often do you shop at other fashion retailers (per year)	12.22	10.00	13.06	No
How much are you spending per shop- ping trip at other fashion retailers (€)	45.25	40.00	32.40	No
Gender	95 % female; 5 % male			Yes
Do you have a car (which you can use for your shopping trips)	93 % yes, 7 % no			Yes
Do you shop online for fashion	40 % yes, 60 % no			No
How much money did you spent on fashion last month (€)	111.37	100.00	85.70	Yes
How much money have you spent on fashion online last month (% of total)	26.96 %	0 %	37.29 %	No
How old are you	38.82	40.00	9.80	Yes
What is your monthly income $(1 \le \pounds 1,000; 2 = \pounds 1,001 - \pounds 1,500; 3 = \pounds 1,501 - \pounds 2,000; 4 = \pounds 2,001 - \pounds 2,500; 5 = \pounds 2,501 - \pounds 3,000; 7 \ge \pounds 3,000)$	3.01 (€1,501–€2,000)	3.00 (€1,501– €2,000)	1.76	Yes
Due to the UABP I am visiting the retailer more often	5.56	6.00	1.56	No

(continued)

M.-O. Jauch, *Uniform Across-the-Board Promotions*, Contributions to Management Science, DOI 10.1007/978-3-319-07115-2, © Springer International Publishing Switzerland 2014

Question	Mean	Median	SD	Included
I go to the retailer specifically during periods when the UABP takes place	5.48	6.00	1.65	No
The UABP is a main reason why I shop at the retailer	4.65	5.00	1.76	No
If no promotion is offered I am buying fewer things	4.33	4.00	1.77	No
The retailer offers more attractive pro- motions than other retailers	5.37	6.00	1.32	No
I can buy promoted articles more often at this retailer than at other retailers	5.03	5.00	1.45	No
This retailer offers promotions more frequently than other retailers	5.14	5.00	1.52	No
This retailer offer more attractive pro- motions than other retailers	5.23	5.00	1.49	No
The UABP offered is a very attractive type of promotion	6.13	7.00	1.25	No
The UABP offered is a more attractive type of promotion than other types of promotions with other retailers	5.47	6.00	1.42	No
Price knowledge and involvement (Liker (antecedents)	t scale 1 = Stron	agly disagree, $7 = s$	trongly agre	ee)
I am very interested in the price of articles and I do compare prices	5.66	6.00	1.42	Yes
I know quite precisely what a similar article would cost at one of the retailer's competitors	4.88	5.00	1.57	Yes
I am very involved regarding my deci- sion on where to buy clothes	4.67	5.00	1.63	Yes
I am often contemplating where to best buy fashion articles	4.68	5.00	1.71	Yes
UABP user intensity (antecedents and co	onsequences)			
Number of UABPs the customer has participated in (from 2008 to 2012)	8.06	7.00	5.03	Yes
Participation rate since first having par- ticipated in a UABP (in %)	39.99 %	37.03 %	20.83 %	Yes
Average value of shopping basket dur- ing UABP (in €)	34.36	31.45	17.01	No
Average number of articles bought dur- ing a UABP	4.33	4.00	1.84	No
Average value of items bought during a UABP	8.05	7.66	2.64	No
Customer who has participated in at least 2 of the last 4 UABPs $(1 = \text{for} yes; 0 = no)$	0.50	1.00	0.50	Yes

(continued)

Mean	Median	SD	Included
e 1 = Strong	ly disagree, $7 = strong$	ly agree)	
4.92	5.00	1.58	No
5.28	5.00	1.33	No
4.45	4.00	1.62	No
Strongly disa	gree, 7 = strongly agre	e) (conseq	uences)
5.67	6.00	1.19	No
5.02	5.00	1.24	No
5.03	5.00	1.28	Yes
5.62	6.00	1.29	Yes
5.35	5.00	1.23	Yes
5.71	6.00	1.19	Yes
5.85	6.00	1.32	Yes
6.05	7.00	1.30	No
r (Likert sca	le 1 = Strongly disagree	e, 7 = strong	ngly agree)
5.49	6.00	1.22	Yes
4.86	5.00	1.38	No
5.04	5.00	1.29	Yes
5.09	5.00	1.26	Yes
4.93	5.00	1.33	Yes
havior (Like	ert scale $1 = Strongly di$	isagree, 7	= strongly
5.48	6.00	1.65	Yes
4.37	5.00	1.91	Yes
5.84	6.00	1.52	Yes
3.51	4.00	1.85	No
	Mean e I = Strong 4.92 5.28 4.45 trongly disa 5.67 5.02 5.03 5.62 5.35 5.71 5.85 6.05 r (Likert sca 5.49 4.86 5.04 5.09 4.93 havior (Like 5.48 4.37 5.84 3.51	Mean Median $e \ l = Strongly \ disagree, \ 7 = strongly$ 5.00 4.92 5.00 5.28 5.00 4.45 4.00 $trongly \ disagree, \ 7 = strongly \ agree, \ 5.67$ 6.00 5.02 5.00 5.03 5.00 5.62 6.00 5.71 6.00 5.85 6.00 6.05 7.00 r (Likert scale $l = Strongly \ disagree, \ 5.49$ 6.00 5.04 5.00 5.09 5.00 5.09 5.00 4.93 5.00 4.37 5.00 5.84 6.00 3.51 4.00	MeanMedianSD $e l = Strongly disagree, 7 = strongly agree)$ 4.92 5.00 5.28 5.00 5.28 5.00 4.45 4.00 4.45 4.00 4.45 4.00 1.62 $trongly disagree, 7 = strongly agree)$ 5.67 6.00 5.02 5.00 5.03 5.00 5.62 6.00 5.62 6.00 5.62 6.00 5.71 6.00 5.85 6.00 5.85 6.00 6.05 7.00 7.00 1.30 r (Likert scale $l = Strongly disagree, 7 = $

(continued)

Question	Mean	Median	SD	Included
Customer loyalty (Likert scale $1 =$ Strongly disagree, $7 =$ strongly agree) (consequences)				
I would recommend the retailer to my friends	5.58	6.00	1.32	Yes
I will visit this retailer more often in the future	4.47	4.00	1.40	Yes
I will continue to visit this retailer in the future	6.03	6.00	1.17	Yes